

To
The General Manager,
Department of Corporate Service
BSE Limited
Phiroze Jecjeebhoy Towers,
Dalal Street,
Mumbai - 400 001

Dear Sir,

Sub: Application under Regulation 37 of the SEBI (Listing Obligations and Disclosure Requirements), Regulations, 2015, as amended (Listing Regulation) for the Composite Scheme of Arrangement among Heritage Foods Limited ("Transferor Company" or "HFL") and Heritage Foods Retail Limited ("Transferor Company" or "Demerged Company" or "HFRL") and Future Retail Limited ("Resulting Company" or "FRL") and their respective Shareholders and Creditors proposed to be filed under sections 391 to 394 read with Sections 100 to 103 of the Companies Act, 1956 and Section 52 of the Companies Act, 2013 and other applicable provisions of the Companies Act, 1956 and Companies Act, 2013 (as may be applicable)

Ref: Stock Code: 519552

Please find enclosed the following documents:

Sr. No.	Documents to be submitted along with application under Regulation 37 of the SEBI (Listing Obligations and Disclosure Requirements), Regulations, 2015	Annexure	Page Nos.
1.	Certified true copy of the resolution passed by the Board of Directors of the company.	Annexure-A	01 - 04
2.	Certified copy of the draft Scheme of Amalgamation / Arrangement, etc. proposed to be filed before the High Court.	Annexure-B	05 - 41
3.	Valuation report from Independent Chartered Accountant as applicable as per		42 – 55
4.	Report from the Audit Committee recommending the draft scheme taking into consideration, inter alia, the valuation report at sr. no. 3 above.	Annexure-D	56 - 56
5.	Fairness opinion by Merchant Banker.	Annexure-E	57 - 60
6.	Shareholding pattern of all the Companies pre and post Amalgamation / Arrangement as per the format provided under Regulation 31 of the SEBI (Listing Obligations and Disclosure Requirements), Regulations, 2015.	Annexure-F	61 - 87
7.	Audited financials of the transferee/resulting and transferor/demerged companies for the last 3 financial years (financials not being more than 6 months old) as per Annexure I.	Annexure- G	88 - 108
8.	Quarterly compliance Report on Corporate Governance as per Regulation 27 (2)(a) of the SEBI (Listing Obligations and Disclosure Requirements), Regulations, 2015.	Аплехиге-Н	109 - 112
9.	Complaint report as per Annexure-III of SEBI Circular no. CIR/CFD/CMD/16/2015 dated November 30, 2015 (To be submitted within 7 days of expiry of 21 days from the date of uploading of Draft Scheme and related documents on Exchange's website).	Will submit within 7 days of expiry of 21 days from the date of uploading Draft scheme	
10.	Compliance report with the requirements specified in Part-A of Annexure I of SEBI circular no. CIR/CFD/CMD/16/2015 dated November 30, 2015.	Annexure-I	113 - 113



HERITAGE FOODS LIMITED

(Formerly known as M/s. Heritage Foods (India) Limited) CIN: L15209TG1992PLC014332

AN ISO: 22000 CERTIFIED COMPANY







Sr. No.	Documents to be submitted along with application under Regulation 37 of the SEBI (Listing Obligations and Disclosure Requirements), Regulations, 2015	Annexure	Page Nos.
10000	If as per the company, approval from the shareholders through postal ballot and e-voting, as required under Para (I)(A)(9)(a) of Annexure I of SEBI Circular no. CIR/CFD/CMD/16/2015 dated November 30, 2015, is not applicable then as required under Para (I)(A)(9)(c) of said SEBI circular, submit the following:		
11,	 a) An undertaking certified by the auditor clearly stating the reasons for non applicability of Sub Para 9(a). 	Annexure-J	114 - 115
	 b) Certified copy of Board of Director's resolution approving the aforesaid auditor certificate. 	Annexure-A	01 - 04
12.	Name of the Designated Stock Exchange (DSE) for the purpose of coordinating with SEBI. Certified true copy of the resolution passed by the Board of Directors, in case BSE is DSE.	BSE Limited Annexure-A	01 - 04
13.	Brief details of the transferee/resulting and transferor/demerged companies as per format enclosed.	Annexure-K	116 - 120
14.	Networth certificate (excluding Revaluation Reserve) together with related workings pre and post scheme for the transferee and/or resulting company.	Annexure-L	121 - 126
15.	Capital evolution details of the transferee/resulting and transferor/demerged companies as per format enclosed.	Annexure-M	127 - 128
16.	Confirmation by the Managing Director/ Company Secretary as per format enclosed.	Annexure-N	129 - 129
17.	Statutory Auditor's Certificate confirming the compliance of the accounting treatment etc. as specified in Para (I)(A)(5)(a) of Annexure I of SEBI Circular no. CIR/CFD/CMD/16/2015 dated November 30, 2015, as per the format given in Annexure II of aforesaid SEBI circular.	Annexure-O	130 – 130
18.	Annual Reports of all the listed transferee/resulting/demerged/etc. companies involved and audited financial of all the unlisted transferor/ demerged/resulting etc. companies for the last financial year.	Annexure-P	Enclosed Separately
	Processing fee (non-refundable) payable will be as below, though RTGS - Details given in Annexure IX or though Cheque/DD favoring 'BSE Limited'	ICICI Bank Cheque	
19.	Rs.1,00,000/- plus Service Tax as applicable, where one entities/companies are Merged or one new company formed due to De-merger	No:322307 Dt:15/11/16 for Rs.2,30,000/- in	Enclosed Separately
	Rs.2,00,000/- plus Service Tax as applicable, where more than one entity/ company is Merged or more than one new company formed due to De-merger.	favour of BSE Limited	
	Name & Designation of the Contact Person: Mr. Umakanta Barik		
20.	Telephone Nos. (landline & mobile): (off) 040-23391221 (mob) 9392277999		
	Email ID: umakanta@heritagefoods.in & hf@heritagefoods.in		

We hope you may find the above mentioned documents in order and give your approval to the draft scheme and issue "No Objection Certificate" as per the provisions of the Listing Regulation and SEBI Circular.

Thanking you,

For HERITAGE FOODS LIMITED

UMAKANTA BARIK Company Secretary M No: FCS-6317

Place: Hyderabad

Encl: a/a



(Formerly known as M/s. Heritage Foods (India) Limited) CIN - L15209TG1992PI C014332

CIN: L15209TG1992PLC014332 AN ISO: 22000 CERTIFIED COMPANY





EXTRACT OF THE RESOLUTION PASSED AT THE 1718T MEETING OF THE BOARD OF DIRECTORS OF THE COMPANY HELD ON MONDAY, 7TH NOVEMBER, 2016 AT 3.30 PM AT PARK HYATT HYDERABAD, ROAD NO.2, BANJARA HILLS, HYDERABAD, TELANGANA 500034

 Valuation Certificate of Retail, Agri, Bakery & VetCa verticals of the Company submitted by M/s. Raju & Prasad, Statutory Auditor of the Company.

"RESOLVED THAT the valuation certificate submitted by M/s. Raju and Prasad Chartered Accountants, Hyderabad indicating the value of Retail, Agri, Bakery & VetCa verticals of the Company as of close of business on 31st October, 2016 as reviewed and recommended by the Audit Committee of the Company be and is hereby approved.

 Valuation Report for issue of shares and Draft Composite Scheme Of Arrangement Among Heritage Foods Limited ("Transferor Company" Or "HFL") And Heritage Foods Retail Limited ("Transferee Company" Or "Demerged Company" Or "HFRL") And Future Retail Limited ("Resulting Company" Or "FRL") And Their Respective Shareholders And Creditors.

"RESOLVED THAT pursuant to the provisions of Sections 391 to 394 read with Sections 100 to 103 of the Companies Act, 1956 and Section 52 of the Companies Act, 2013 and other applicable provisions of the Companies Act, 1956 and Companies Act, 2013 (as may be applicable), upon their notification (including any statutory modifications or reenactment(s) thereof), all relevant securities laws, regulations and circulars including Circular No: CIR/CFD/DIL/5/2013 dated February 4, 2013 as further clarified/amended by Circular No: CIR/CFD/DIL/8/2013 dated May 21, 2013 and Circular No: CIR/CFD/CMD/16/2015 dated November 30, 2015 (hereinafter collectively referred to as the Circulars), other applicable provisions if any of the Memorandum and Articles of Association of the Company and subject to receipt of approval of the shareholders of the Company and all relevant corporate, regulatory, governmental, judicial approvals and third party consents, and subject to compliance with all applicable laws and regulations, the draft Composite Scheme of Arrangement among Heritage Foods Limited ("Transferor Company" or "HFL"), Heritage Foods Retail Limited ("Transferee Company" or "Demerged Company" or "HFRL") and Future Retail Limited ("Resulting Company" or "FRL") and their respective shareholders and creditors, placed before this meeting and duly initialed by the Chairman for purpose of identification, be and is hereby considered and approved.

"RESOLVED FURTHER THAT the valuation certificate of Retail, Agri, Bakery & VetCa verticals of the Company as of close of business on 31st October, 2016 submitted by M/s Raju & Prasad, Hyderabad statutory auditor of the Company for transferring the above said verticals to its wholly owned subsidiary Company viz. Heritage Foods Retail Limited, on a slump sale basis with a consideration of Rs.135 Crores (One Hundred and Forty Crores) by way of issue of 1,40,00,000 equity shares, at face value of Rs.10/- each with a premium of Rs.86.43/- to the HFL, as reviewed and recommended by the Audit Committee, and placed before the Board for its review, be and is hereby taken on record and approved for the purposes of the Composite Scheme of Arrangement.

"RESOLVED FURTHER THAT the Valuation report submitted by M/s S R Batliboi Chartered Accountants, Hyderabad recommending the issue of 17,847,420 no's equity shares (face value of Rs.2/- each with a premium of Rs.163.29/-, the price of the shares has been determined as per SEBI Regulation) of shares of Future Retail Limited, as a consideration for the demerger to the shareholders of Heritage Foods Retail Limited (i.e. Heritage Foods Limited) and the fairness opinion submitted by Keynote Corporate Services Ltd, Mumbai as reviewed and recommended by the Audit Committee, and placed before the Board for its review, be and are hereby taken on record and approved for the purposes of the Composite Scheme of Arrangement.



HERITAGE FOODS LIMITED

(Formerly known as M/s. Heritage Foods (India) Limitet) CIN: L15209TG1992PLC014332

AN ISO: 22000 CERTIFIED COMPANY





"RESOLVED FURTHER THAT in the opinion of the Board, the said Composite Scheme of Arrangement will be of advantage and beneficial to the Company and its shareholders, creditors and other stakeholders and that the terms thereof are fair and reasonable."

"RESOLVED FURTHER THAT Smt. N Bhuvaneswari, Vice Chairperson & Managing Director, Smt. N Brahmani, Executive Director and Dr M Sambasiva Rao, President of the Company be and are hereby severally authorized on behalf of the Company to sign and execute any application, affidavit, petition, agreement, undertakings or any other documents or deeds or pleadings, as may be required to be filed for this purpose and to agree and to make such modification/alterations as may be required by any regulatory, judicial or governmental authorities in connection with the "Composite Scheme of Arrangement among Heritage Foods Limited ("Transferor Company" or "HFL"), Heritage Foods Retail Limited ("Transferee Company" or "Demerged Company" or "HFRL") and Future Retail Limited ("Resulting Company" or "FRL") and their respective shareholders and creditors" and are further severally authorized to do all such things, deeds and acts as may be necessary and expedient in connection with the consummation of: (i) transfer of the Retail, Agri, Bakery and VetCa verticals of the Company on a slump sale basis to the wholly owned subsidiary of the Company, Heritage Foods Retail Limited; and (ii) the demerger of the undertaking comprising the Retail, Agri and Bakery verticals from Heritage Foods Retail Limited to Future Retail Limited, including but not limited to the filing and implementation of the Composite Scheme of Arrangement, submission of all necessary documents to the Courts in this regard and to give effect to the Composite Scheme of Arrangement.

"RESOLVED FURTHER THAT the draft of the Implementation Agreement inter alia between the Company, Heritage Foods Retail Limited and Future Retail Limited, as reviewed and recommended by the Audit Committee, and placed before the Board for its review, be and is hereby considered and approved.

"RESOLVED FURTHER THAT Smt. N Bhuvaneswari, Vice Chairperson & Managing Director, Smt. N Brahmani, Executive Director and Dr M Sambasiva Rao, President of the Company be and are hereby severally authorized on behalf of the Company to sign and execute the Implementation Agreement inter alia between the Company, Heritage Foods Retail Limited and Future Retail Limited and to do all acts, deeds and things required to execute, deliver and perform such agreement;

"RESOLVED FURTHER THAT the Common Seal of the Company, if required, be affixed on the aforesaid documents in presence of any Director of the Company who does sign the same and the same be countersigned by Sri. Umakanta Barik, Company Secretary of the Company being the person authorized in this regard.

"RESOLVED FURTHER THAT subject to the directions as may be issued by the Hon'ble High Court of Hyderabad and/or the High Court of Bombay, the Company appoints Sri. D Seetharamaiah, (DIN:00005016) or failing him Smt. N Bhuvaneswari, (DIN: 00003741) or such other person as may be directed by the Hon'ble High Court of Hyderabad and/or the High Court of Bombay as Chairman of the Court convened meetings of the Shareholders and creditors of the Company.



HERITAGE FOODS LIMITED

(Formerly known as M/s. Heritage Foods (India) Limited)
CIN: L15209TG1992PLC014332



"RESOLVED FURTHER THAT any one of the following Directors/Executives of the Company:

- 1. Smt N Bhuvaneswari, Vice Chairperson & Managing Director
- 2. Smt. N Brahmani, Executive Director
- 3. Dr M Sambasiva Rao, President
- 4. Sri A Prabhakara Naidu, Chief Financial Officer
- 5. Sri Umakanta Barik, Company Secretary

be and are hereby severally authorized on behalf of the Company to:

- i. File the Scheme and/or any other information/ details with the concerned stock exchanges and/or any other body or regulatory authority or agency including appearing and representing the Company before SEBI / stock exchange(s) and to take all necessary steps as may be required for the implementation and consummation of the Scheme in all respects whatsoever in order to obtain no-objection, approval or sanction of the scheme or for giving effect thereto and take all steps as may be required, including without limitation for obtaining approvals and/or consents of the shareholders, creditors, banks, financial institutions, all relevant regulatory, governmental and judicial authorities and all relevant third parties, as may be required from time to time in that behalf.
- ii. Give such directions as they may consider necessary to settle any question or resolve any difficulty or matter arising under the Composite Scheme of Arrangement or in regard to the meaning or interpretation of the Scheme or the implementation thereof and/or to review the position relating to the satisfaction of various conditions as stipulated in the said Scheme and if necessary, subject to consent of the Board of the Company, to waive any of those conditions (to the extent permissible under law);
- iii. Make necessary applications and take all steps, actions, deeds and things required for the Company to perform its obligations under the Implementation Agreement, including the formation of a suitable committee to oversee the operations of the Retail, Agri and Bakery verticals, and to make such applications and take such steps, actions, deeds and things as may be required for the completion of conditions required to be satisfied for the consummation of transactions contemplated in the Implementation Agreement.
- iv. Appoint solicitors, advocates, attorneys, company secretaries, advisors, valuers, auditors, accountants, registrars or such other persons or agencies, as may be required in relation to or in connection with the Scheme, on such terms and conditions including as regards their fees or remuneration as they may deem fit and issue letters of appointment or vakalatnamas or authority letters in this regard;
- v. Do any other acts, deeds, matters and things as may be necessary, proper or expedient for the implementation and to give effect to the Scheme and for the matters connected therewith or incidental thereto.

vi.



HERITAGE FOODS LIMITED

(Formerly known as M/s. Heritage Foods (India) Limited)
CIN: L15209TG1992PLC014332





- vii. Make and agree to such alteration and changes to the Scheme as may in their opinion be desirable or expedient or advised by the counsels and/or including as may be necessary for satisfying the requirements or conditions imposed by the Court
 - or any other authority concerned, provided that no alteration which amounts to a substantial change shall be made to the Scheme except with the prior approval of the Board of Directors.
- viii. Withdraw the Scheme at any stage, in case the changes or modifications required or the conditions imposed by the Stock Exchanges the Court and/or any other authority are not acceptable and Scheme cannot be implemented otherwise and,
- ix. Take all such actions and steps in the above matter, as may be required from time to time to give effect to the above resolution;

"RESOLVED FURTHER THAT BSE Limited be and is hereby appointed as the designated stock exchange for getting the observation/approval of the SEBI for the Scheme."

*RESOLVED FURTHER THAT in terms of paragraph 5.16 (b) of the SEBI Circular No. CIR/CFD/DIL/5/2013 dated February 4, 2013, (as amended by Circular No. CIR/CFC/DIL/8/2013 dated May, 21, 2013 and by paragraph (9) (c) of Circular CIR/CFD/CMD/16/2015 dated November 30, 2015), the Board of Directors takes on record and approves the undertaking of the Company, as certified by M/s. Raju & Prasad, Hyderabad, Statutory Auditor, which undertaking has clearly stated the reasons for non-applicability of paragraph 5.16(a) of the Circular No. CIR/CFD/DIL/5/2013 dated February 4, 2013, paragraph 5.16(a) of the Circular No. CIR/CFC/DIL/8/2013 and paragraph (9) (c) of Circular No. CIR/CFD/CMD/16/2015 dated November 30, 2015".

"RESOLVED FURTHER THAT the copies of the foregoing resolutions certified to be true copies by any one of the above mentioned authorized persons be furnished to all concerned as may be necessary and they be requested to act thereon.

//CERTIFIED TRUE COPY//

UMAKANTA BARIK

Company Secretary M No: FCS 6317



FD WAPPINE





Heritage Foods Retail Limited

EXTRACT OF THE RESOLUTION PASSED AT THE 38TH MEETING OF THE BOARD OF DIRECTORS OF THE COMPANY HELD ON MONDAY, 7TH NOVEMBER, 2016 AT 04:30 PM AT PARK HYATT HYDERABAD, ROAD NO. 2, BANJARA HILLS, HYDERABAD, TELANGANA 500034

"RESOLVED THAT pursuant to the provisions of Sections 391 to 394 read with Sections 100 to 103 of the Companies Act, 1956 and Section 52 of the Companies Act, 2013 and other applicable provisions of the Companies Act, 1956 and Companies Act, 2013 (as may be applicable), upon their notification (including any statutory modifications or re-enactment(s) thereof), all relevant securities laws, regulations and other applicable provisions if any of the Memorandum and Articles of Association of the Company and subject to receipt of approval of the shareholders of the Company and all relevant corporate, regulatory, governmental, judicial approvals and third party consents, and subject to compliance with all applicable laws and regulations, the draft Composite Scheme of Arrangement among Heritage Foods Limited ("Transferor Company" or "HFL"), Heritage Foods Retail Limited ("Transferee Company" or "Demerged Company" or "HFRL") and Future Retail Limited ("Resulting Company" or "FRL") and their respective shareholders and creditors, placed before this meeting and duly initialed by the Chairman for purpose of identification, be and is hereby considered and approved.

"RESOLVED FURTHER THAT the valuation certificate of Retail, Agri, Bakery & VetCa verticals of the Holding Company as of close of business on 31st October, 2016 submitted by M/s Raju & Prasad, Hyderabad statutory auditor of the Company for transferring Retail, Agri, Bakery & VetCa verticals of the Holding Company as of close of business on 31st October, 2016 to the Company, on a slump sale basis be and is hereby taken on record and approved.

*RESOLVED FURTHER THAT the slump sale value of the above verticals is Rs.135 Crores which shall be discharged/paid to the holding company (i.e. HFL) by way of issue of 1,40,00,000 equity shares, at face value of Rs.10/- each with a premium of Rs.86.43/- for the purposes of the Composite Scheme of Arrangement.

"RESOLVED FURTHER THAT the Valuation report submitted by M/s S R Batliboi Chartered Accountants, Hyderabad recommending the issue of 17,847,420 no's equity shares (face value of Rs.2/- each with a premium of Rs.163.29/-, the price of the shares has been determined as per SEBI Regulation) of shares of Future Retail Limited, as a consideration for the demerger to the shareholders of the company (i.e. Heritage Foods Limited) be and are hereby taken on record and approved for the purposes of the Composite Scheme of Arrangement.

"RESOLVED FURTHER THAT in the opinion of the Board, the said Composite Scheme of Arrangement will be of advantage and beneficial to the Company and its shareholders, creditors and other stakeholders and that the terms thereof are fair and reasonable."

"RESOLVED FURTHER THAT Smt. N Brahmani, Managing Director, Smt. N Bhuvaneswari, Director and Dr M Sambasiva Rao, Director of the Company be and are hereby severally authorized on behalf of the Company to:

i. to sign and execute any application, affidavit, petition, agreement, undertakings or any other documents or deeds or pleadings, as may be required to be filed for this purpose and agree to make such modification/alterations as may be required by any regulatory, judicial or governmental authorities in connection with the above Composite Scheme of Arrangement.

Regd. Office: #6-3-541/C, Panjagutta, Hyderabad - 500082



Heritage Foods Retail Limited

- ii. File the Scheme and/or any other information/ details with the regulatory authority or agency including appearing and representing the Company and to take all necessary steps as may be required for the implementation and consummation of the Scheme in all respects whatsoever in order to obtain no-objection, approval or sanction of the scheme or for giving effect thereto and take all steps as may be required.
- Give such directions as they may consider necessary to settle any question or resolve any difficulty or matter arising under the Composite Scheme of Arrangement or in regard to the meaning or interpretation of the Scheme or the implementation thereof and/or to review the position relating to the satisfaction of various conditions as stipulated in the said Scheme.
- Make necessary applications and take all steps, actions, deeds and things required for the Company to perform its obligations under the Implementation Agreement, including the formation of a suitable committee to oversee the operations of the Retail, Agri and Bakery verticals, and to make such applications and take such steps, actions, deeds and things as may be required for the completion of conditions required to be satisfied for the consummation of transactions contemplated in the Implementation Agreement.
- v. Appoint solicitors, advocates, attorneys, company secretaries, advisors, valuers, auditors, accountants, registrars or such other persons or agencies, as may be required in relation to or in connection with the Scheme, on such terms and conditions including as regards their fees or remuneration as they may deem fit and issue letters of appointment or vakalatnamas or authority letters in this regard;
- Make and agree to such alteration and changes to the Scheme as may in their opinion be desirable or expedient or advised by the counsels and/or including as may be necessary for satisfying the requirements or conditions imposed by the Court
- Take all such actions and steps in the above matter, as may be required from time to time to give effect to the above resolution;

//CERTIFIED TRUE COPY//

Dr M Sambasiva Rao

Director

Regd. Office: #6-3-541/C, Panjagutta, Hyderabad - 500082





CERTIFIED TRUE COPY OF THE RESOLUTION PASSED AT THE MEETING NO. BM-06/2016-17 OF THE BOARD OF DIRECTORS OF FUTURE RETAIL LIMITED HELD ON NOVEMBER 07, 2016 AT THE BOARD ROOM, BASEMENT LEVEL 1, HOTEL PARK HYATT, ROAD NO. 2, BANJARA HILLS, HYDERABAD, TELANGANA 500344

APPROVAL OF THE SCHEME, THE PROPOSAL, ISSUE OF SHARES TO DISCHARGE CONSIDERATION FOR THE DEMERGED UNDERTAKING

"RESOLVED THAT pursuant to the provisions of Sections 391 to 394 and Sections 100 to 103 of the Companies Act, 1956 and Section 52 of the Companies Act, 2013 and other applicable provisions, if any, read with related rules thereto as applicable under the Companies Act, 1956 and/or the Companies Act, 2013 as amended (including any statutory modification or re-enactment or amendment thereof), and enabling provisions of the Memorandum of Association and Articles of Association of the Company and subject to the approval of Competition Commission of india, pursuant to provisions of the Competition Act, 2002 and the rules and regulations made thereunder (including any statutory modification or re-enactment thereof) and the sanction of the High Court of Judicature at Bombay and Hyderabad or National Company Law Tribunal, as the case may be, the Securities and Exchange Board of India (SEBI), Stock Exchange(s) or such other competent authority, as may be applicable and subject to the requisite approval of the Shareholders and Creditors of the Company, the consent of the Board be and is hereby accorded to the Composite Scheme of Arrangement amongst Heritage Foods Limited ("Transferor Company" or "HFL"), Heritage Foods Retail Limited ("Transferoe Company" or "Demerged Company" or "HFRL") and Future Retail Limited ("Resulting Company" or "FRL") and their respective Shareholders and Creditors (hereinafter referred to as "Scheme" or "the Scheme").

RESOLVED FURTHER THAT the draft of the Scheme providing for demerger of Retail Undertaking of Mis. Heritage Foods Retail Limited ("Transferee Company" or "Demerged Company" or "HFRL") and vesting in the Company with effect from the "Demerger Appointed Date" as defined in the Scheme and also providing for various other matters connected with and/or consequential end/or incidental thereto, be and is hereby approved.

RESOLVED FURTHER THAT the appointment of M/s. S.R. Batibol & Co. LLP, independent Chartered Accountants as valuer for the purpose of Scheme of Arrangement / Demerger be and is hereby approved and ratified and that the report on valuation submitted by the aforesaid valuer placed before the Board be and is hereby approved, which under the present proposal of Scheme, in the opinion of the Board of Directors, is advantageous and beneficial to the Stakeholders of the Company.

RESOLVED FURTHER THAT for the purposes of Scheme and based on the Valuation Report of an independent valuer, Ms. S.R. Battibol & Co. LLP. Chartered Accountants (Firm Registration No. 301003E/E300005) the issuance of following Equity Shares be and is hereby approved:

1,78,47,420 (One Crore Seventy Eight Lakhs Forty Seven Thousand Four Hundred and Twenty) equity shares of Rs.
 2/- (Rupees Two) each, fully paid up to the equity shareholders of HFRL as on the Record Date (as of now HFL) on a proportionate basis.

RESOLVED FURTHER THAT the appointment of M/s. Keynote Corporate Services Limited, a Category I Merchant Banker, who have been appointed for the purpose of issuing fairness opinion, be and is hereby approved and ratified and that the Fairness Opinion submitted by the aforesaid Merchant Banker placed before the Board be and is hereby approved.

RESOLVED FURTHER THAT the report from the Audit Committee dated November 07, 2016, recommending the draft. Scheme, taking into consideration, inter alia the issuance of Equity shares to the shareholders of HFRL as placed before the Board, be and is hereby approved.

RESOLVED FURTHER THAT an undertaking certified by M/s. NGS & Co. LLP, Statutory Auditors of the Company stating clearly the reasons for non-applicability under Para (I)(A)(9)(c) of Para (I)(A)(9)(a) of Annexure I of SEBI Circular no. CIR/CFD/CMD/16/2015 dated November 30, 2015 be and is hereby reviewed noted and approved.

RESOLVED FURTHER THAT BSE Limited be and is hereby chosen / appointed as designated Stock Exchange for the purpose of co-ordinating with SEBI and for the said Scheme and matters connected therewith.

FUTURE RETAIL

RESOLVED FURTHER THAT Mr. Kishore Biyani, Chairman and Managing Director, Mr. Rakesh Biyani, Joint Managing Director, Mr. C. P. Toshniwal, Chief Financial Officer, Mr. Virendra Samani, Dy. Company Secretary, Mr. Sanjay Rathi, Mr. Deepak Tanna, Mr. Sanjay Mutha, Mr. Vimal Dhruve, Mr. Anant Gude, Mr. Arun Bhandari, Mr. Subodh More, Mr. Chetan Dalai, Authorised Persons ("Authorised Signatories") of the Company be and are hereby authorised severally to take all the necessary actions and steps on behalf of the Company, infer alia, to:

- a) file the Scheme, application and/or any other information / details, documents, papers with the concerned Stock Exchange(s) or SEBI or any other body or regulatory authority or agency to obtain approval or sanction to any of the provisions of the Scheme or for giving effect thereto;
- file applications with the Competition Commission of India and other regulatory authorities seeking approval for the transaction as contemplated under the Scheme;
- c) file applications with the Hon'ble High Court of Judicature at Bombay and/or at Hyderabad and/or at National Company Law Tribunal or such other appropriate authority seeking directions as to convening / dispensing with the meeting of the shareholders / creditors of the Company and, where necessary, to take steps to convene and hold such meetings as per the directions of the High Court / Tribunal;
- d) finalise and settle the draft of the notices for convening the shareholders' creditors' meetings and the draft of the explanatory statement under Section 393 of the Companies Act, 1956 or other applicable provisions, if any, with any modifications as they may deem fit;
- e) file applications, petitions for confirmation of the Scheme with the Hon'ble High Court of Judicature at Bombay and/or Hyderabad or such other competent authority;
- f) file affidavits, petitions, pleadings, applications or any other proceedings incidental or deemed necessary or useful in connection with the above proceedings and to engage Counsels, Advocates, Solicitors, Chartered Accountants, Consultants, Advisors and other professionals and to sign and execute vakalatnama, wherever necessary, and sign and issue public advertisements and notices;
- g) take all necessary steps in the matter that may be required for approving the Scheme and obtaining consent and confirmation for the said Scheme from such other authorities and parties including the Hon'ble High Court of Judicature at Bombay and/or Hyderabad or National Company Law Tribunal, shareholders, creditors, lenders as may be considered necessary;
- apply for and obtain requisite approval and represent before Registrar of Companies, Regional Director and such other authorities and entitles including shareholders, term loan / working capital lenders, financial institution(s), other lenders as may be considered necessary to give effect to the Scheme;
- i) make such alterations / changes to the Scheme as may be expedient or necessary including withdrawal of Scheme and applications, particularly for satisfying the requirements or conditions imposed by the Central Government or the Hon'ble High Court of Judicature at Bombay and/or Hyderabad or any other authority provided that no alteration which amounts to a material change shall be made to the substance of the Scheme except with the prior approval of the Board of Directors / Committee of Directors;
- j) settle any question or difficulty that may arise with regard to the implementation of the above Scheme, and to give effect to the above resolution; and
- do all further acts, deeds, matters and things as may be necessary, proper or expedient to give effect to the Scheme and for matters connected therewith or incidental thereto.

RESOLVED FURTHER THAT for the purpose of exercising any of the above authorities / powers or where any matter or issue concerning the Scheme or for giving effect thereto is required to be decided and approved by means of a resolution, the same be approved by the Committee of Directors (as constituted by the Board from time to time) and the said Committee be and is hereby authorized to decide all matters and to do all such acts, deeds, matters and things as may be necessary and settle any questions that may arise concerning the Scheme and transaction and matters relating or incidental thereto.

RESOLVED FURTHER THAT the Common Seal of the Company, if required, be affixed as per the provisions of the Articles of Association of the Company on all necessary applications, deeds, guarantees, agreements, documents, and such other certificates, documents and papers as may be required in the presence of any one of the aforesaid Authorised Signatory(ies), who do sign the same in token thereof and further severally authorised to carry the Common Seal of the Company when the same is required to be affixed, on any document to be executed out of the registered office of the Company as per the provisions of Articles of Association of the Company.



RESOLVED FURTHER THAT any Director or Chief Financial Officer or Company Secretary be and are hereby severally authorized to sign any copy of this resolution as a certified true copy thereof and furnish the same to whomsoever concerned or required."

Certified True Copy For Future Retail Limited

Virendra Samani Dy. Company Secretary ACS # 12971



COMPOSITE SCHEME OF ARRANGEMENT

UNDER SECTIONS 391 TO 394 AND SECTIONS 100 to 103 OF THE COMPANIES ACT, 1956 AND SECTION 52 OF THE COMPANIES ACT, 2013

AMONG

HERITAGE FOODS LIMITED ("Transferor Company" or "HFL")

AND

HERITAGE FOODS RETAIL LIMITED ("Transferee Company" or "Demerged Company" or "HFRL")

AND

FUTURE RETAIL LIMITED ("Resulting Company" or "FRL")

AND

THEIR RESPECTIVE SHAREHOLDERS AND CREDITORS

PREAMBLE

This Composite Scheme of Arrangement is presented pursuant to the provisions of Sections 391 to 394 read with Sections 100 to 103 of the Companies Act, 1956 and Section 52 of the Companies Act, 2013 and other applicable provisions of the Companies Act, 1956 and Companies Act, 2013 (as may be applicable) to reorganize the business of Heritage Foods Limited (hereinafter referred to as the "Transferor Company" or "HFL") by way of Slump Sale (as hereinafter defined) of the Retail Undertaking (as hereinafter defined) and the VetCa Undertaking (as hereinafter defined) to Heritage Foods Retail Limited ("Transferee Company" or "Demerged Company" or "HFRL") and demerger of the Retail Undertaking by the Demerged Company to Future Retail Limited ("Resulting Company" or "FRL").

A. Description of Companies:

- (a) Heritage Foods Limited ("Transferor Company" or "HFL")
 - HFL is a public limited company incorporated under the Companies Act, 1956 (CIN: L15209TG1992PLC014332) and having its registered office at #6-3-541 / C, Punjagutta, Hyderabad - 500082.
 - The equity shares of HFL are listed on the BSE Limited (Stock Code: 519552) and the National Stock Exchange (Stock Code: HERITGFOOD).
 - HFL, formerly known as Heritage Foods (India) Limited, is a company which has 6 (six) key business verticals:

- a. <u>Dairy business vertical</u> HFL produces, sources and markets a complete range
 of dairy products including fresh milk, curd, buttermilk, ice creams and other
 value added products across various States in India.
- Retail business vertical HFL is engaged in the grocery and food retail business, undertaken from its dedicated retail stores.
- c. <u>Agri business vertical</u> HFL is engaged in the business of sourcing, processing and marketing fresh fruits and vegetables. This business vertical acts as the supply chain arm for the Retail Undertaking vertical while also supplying the products to other retail chains and stores.
- Bakery business vertical HFL is engaged in the business of manufacturing and supplying bakery products to other customers, besides HFL retail outlets.
- Veterinary care business vertical HFL supplies cattle feed to dairy farmers and general traders and maize to poultries and distilleries.
- Renewable energy business vertical HFL is engaged in the production of solar energy (2.34 MW) and wind energy (4.2 MW) for captive consumption.

The Retail Undertaking (as hereinafter defined) comprises the retail business vertical, the agri business vertical, and the bakery business vertical. The VetCa Undertaking (as hereinafter defined) comprises the veterinary care business vertical.

(b) Heritage Foods Retail Limited ("Transferee Company" or "Demerged Company" or "HFRL")

HFRL is a public limited company incorporated under the Companies Act, 1956 (CIN: U15400TG2008PLC062054) and having its registered office at #6-3-541 / C, Punjagutta, Hyderabad – 500082 for undertaking the following activities: trading and dealing in goods and produce, and processing, packaging and selling agri products. HFRL is a wholly owned subsidiary of the Transferor Company.

- (c) Future Retail Limited ("Resulting Company" or "FRL")
 - FRL is a public limited company incorporated under the Companies Act, 1956 (CIN: U51909MH2007PLC268269) and having its registered office at Knowledge House, Shyam Nagar, Off Jogeshewari – Vikhroli Link Road, Jogeshwari (East), Mumbai -400060.
 - The equity shares of FRL are listed on the BSE Limited (Stock Code: 540064) and the National Stock Exchange (Stock Code: FRETAIL).
 - FRL currently operates multiple retail formats in the Indian consumer market under different brand names including: Big Bazaar; FBB; Food Bazaar; Foodhall; Home Town and eZone.

B. Rationale and Purpose of the Scheme of Arrangement:

HFL is inter alia engaged in six different business verticals: the dairy business, the retail business, the agri business, the bakery business, the veterinary care business and the renewable energy business.

The Board of Directors and management of HFL believe and are of the view that risk and reward associated with each of the aforesaid business verticals is different. Further, the reorganization / arrangement will enable HFL to provide greater business attention and focus on the dairy and renewable energy business verticals which have high growth potential, which may result in increasing the profitability while simultaneously attracting strategic partners and lenders for the retail, agri and bakery business verticals of HFL and creating long term value for the various stakeholders. In addition, the veterinary care business vertical, which supplements the agri business vertical, will be restructured into a wholly owned subsidiary to unlock value. Accordingly, the Board of Directors of HFL and HFRL are of the opinion that the Retail Undertaking (comprising the retail business vertical, the agri business vertical and the bakery business vertical) and the VetCa Undertaking (comprising the veterinary care business vertical) should be transferred to a wholly owned subsidiary (HFRL) of HFL. Upon such transfer: (i) HFL would continue to carry on the dairy business, the renewable energy business and other businesses not transferred pursuant to this Scheme; and (ii) the Retail Undertaking (comprising the retail business vertical, the agri business vertical and the bakery business vertical) and the VetCa Undertaking would be transferred to HFRL. Subsequently, the Demerged Undertaking (comprising of the retail business vertical, the agri business vertical and the bakery business vertical) would be demerged from HFRL into FRL; each in terms of this Scheme. This would inter alia help in consolidation of the retail operations of FRL and HFRL in FRL. Upon such demerger, HFRL would continue to carry on the VetCa Undertaking and FRL would continue to carry on retail business transferred to it pursuant to the Scheme.

The Board of Directors of the Transferor Company and the Demerged Company are of the opinion that the arrangement under this Scheme would result in benefit to members, creditors and employees of each of the Transferor Company and the Demerged Company and will not be detrimental to the public. The Board of Directors of the Resulting Company is of the opinion that the demerger under this Scheme would result in expansion of retail business attached with the increase in the value for its members in long run. Further, the proposed arrangement would inter alia achieve the following objectives:

- facilitate each business to be effectively integrated for achieving growth for each of the verticals independently;
- II. enhance management focus and operational flexibility;
- facilitate investment by strategic players;
- create a platform to enhance financial flexibility to pursue growth;
- v. consolidation of the retail operations of FRL and HFRL;
- VI. unlocking of value; and



VII. synergies expected to bring in cost savings in the marketing, selling and distribution expenses for FRL.

In view of the aforesaid, the Board of Directors of all the Companies have considered and proposed this Composite Scheme of Arrangement under the provisions of Sections 391 to 394 read with Sections 100 to 103 of the Companies Act, 1956 and Section 52 of the Companies Act, 2013 and other applicable provisions of the Companies Act, 1956 and the Companies Act, 2013 (as may be applicable).

C. Parts of the Scheme:

The Scheme is divided into the following parts:

- PART I sets out the Definitions, Share Capital and date of taking effect of the Scheme;
- PART II sets out provisions with respect to Slump Sale of the Retail Undertaking and VetCa Undertaking to the Transferee Company;
- PART III sets out provisions for transfer and vesting of the Demerged Undertaking (as defined hereinafter) to and in the Resulting Company;
- PART IV sets out provisions with respect to the reduction of share capital of the Demerged Company through a cancellation of the shares held by its existing shareholders; and
- E. PART V sets out the General Terms and Conditions.

PART I DEFINITIONS, SHARE CAPITAL AND DATE OF TAKING EFFECT

DEFINITIONS:

In this Scheme, unless inconsistent with the subject or context, the following expressions shall have the following meaning:

- 1.1 "Act" means the Companies Act, 1956 or, as the case may be, the Companies Act, 2013 (to the extent applicable) and the rules made thereunder and any statutory modification, amendment or re-enactment thereof for the time being in force.
- 1.2 "Board of Directors" or "Board" shall mean the Board of Directors or any duly authorized committee thereof of HFL, HFRL or FRL, as the case maybe or any other person duly authorized by the Board for the purpose of this Scheme.
- 1.3 "Companies" means HFL, HFRL and FRL, collectively and "Company" means HFL, HFRL or FRL, as the context may require.
- 1.4 "Demerged Undertaking" means the entire undertaking of HFRL pertaining to its Retail Undertaking and includes:

* HAPPINESS

- 1.4.2 Without prejudice to the generality of the provisions of the sub-Clause 1.4.1 above, the Retail Undertaking of HFRL shall include without limitation the following:
- All assets (whether moveable or immoveable) including freehold land, leasehold land, 1.4.3 leasehold premises, office premises, all other assets and properties (whether tangible or intangible, real or personal, corporeal or incorporeal, present, future or contingent) including, without limitation, interests, loans, deposits, advances (including accrued interest), investments including investments in overseas subsidiaries, receivables, cash on hand, investment in mutual funds, liquid funds, balance with banks (including bank fixed deposits), equipment, plant and machinery and the related assets and agreements, capital work in progress, unbilled revenue, furniture, fixtures, office equipment, appliances, accessories, vehicles, power connections, utilities and other service connections, all customer contracts, forward cover contracts, hedging contracts, receivables, claims, refunds, earnest moneys paid, rights and benefits under any agreements or security arrangements and funds, contingent rights, rights arising under contracts and all other rights, title, interests, privileges and benefits of every kind wherever located (including in the possession of vendors, third parties or elsewhere) and used or held, by the Demerged Company in, or otherwise identified for use in, or relating to, the business activities and operations pertaining to the Retail Undertaking of the Demerged Company;
- 1.4.4 All liabilities and all debts, guarantees, assurances, commitments, obligations, loans, and undertakings of any kind, nature and description whatsoever and howsoever arising, present or future and including, without limitation, borrowings, working capital facilities, advances from customers, unearned revenues, bills payable, interest, whether fixed, contingent or absolute, secured or unsecured, asserted or unasserted, matured or unmatured, liquidated or unliquidated, accrued or not accrued, known or unknown, due or to become due, whenever or however arising, (including, without limitation, whether arising out of any contract or tort based on negligence or strict liability) pertaining to or relatable to the Retail Undertaking of the Demerged Company;

Explanation: For the purpose of this Scheme, it is hereby clarified that the liabilities pertaining to the Retail Undertaking of the Demerged Company shall include:

- liabilities, which accrue or arise out of the activities or operations of the Retail Undertaking of the Demerged Company;
- specific loans and borrowings raised, incurred and utilized solely for the activities or operations of the Retail Undertaking of the Demerged Company; and
- (iii) liabilities other than those referred to in sub-clauses (i) and (ii) above, so much of the amounts of general corporate nature or multipurpose borrowings, if any, of the Demerged Company as stand in the same proportion which the value of assets transferred in the demerger bears to the total value of assets of such Demerged

* HEALTH * HAPPINESS

- 1.4.5 All contracts, agreements, leases, memoranda of understanding, memoranda of agreements, arrangements, undertakings, whether written or otherwise, deeds, bonds, schemes, arrangements, sales orders, purchase orders or other instruments of whatsoever nature to which the Demerged Company is a party, relating to its Retail Undertaking, or otherwise identified to be for the benefit of the same, approvals, electricity permits, telephone connections, building and parking rights, pending applications for consents or extension pertaining to or relatable to the Retail Undertaking of the Demerged Company;
- 1.4.6 All intellectual properties, labels, brands, trademarks, trade names, service marks, copyrights, patents, designs, software and computer programmes, databases, domain names, including those pending registrations and applications for brands, trademarks, labels, trade names, service marks, copyrights, patents, designs, software and computer programs, databases and domain names, used by the Demerged Company exclusively in the business, activities and operations pertaining to its Retail Undertaking;
- 1.4.7 All permits, licenses, consents, approvals, authorizations, quotas, rights, powers, permissions, arrangements, assignments, sanctions, entitlements, allotments, exemptions, incentives, tax benefits, deferrals, subsidies, concessions, grants, claims, liberties, special status, benefits and privileges enjoyed or conferred upon or held or availed of by the Demerged Company in relation to or pertaining to its Retail Undertaking, registrations, advantages, no-objection certificates, certifications, easements, and any waivers of the foregoing, issued by any legislative, executive or judicial unit of any Governmental or quasi-Governmental entity or any department, commission, board, agency, bureau, official or other regulatory, local (including Municipal), administrative or judicial authority, used or held for use by the Demerged Company in respect of business, activities and operations pertaining to its Retail Undertaking;
- 1.4.8 All tax credits, including cenvat credits, refunds, reimbursements, claims, exemptions, benefits under service tax laws, value added tax (VAT), purchase tax, sales tax or any other duty or tax or cess or imposts under any Central or State law including sales tax deferrals, special economic zone benefits, excise duty benefits, tax deducted at source, right to carry forward and set-off unabsorbed losses, and depreciation, if any and exemptions, deductions, benefits and incentives under the Income-tax Act in respect of business, activities and operations pertaining to the Retail Undertaking of the Demerged Company;
- 1.4.9 All rights, benefits and other interest, whether held in trust or otherwise, contracts, agreements, powers, engagements, arrangements of all kind, privileges and all other rights including title, interests, other benefits (including tax benefits), easements, privileges, liberties and advantages of whatsoever nature and wheresoever situate belonging to or in the ownership, possession, power or custody of or in the control of or vested in or granted in favour of or enjoyed by the Demerged Company, whether in India or abroad, all pertaining to or relatable to the Retail Undertaking of the Demerged Company;
- 1.4.10 All records, files, papers, manuals, data, sales and advertising materials, lists and other details of customers and suppliers, credit information, pricing information, whether in physical or electronic form, all pertaining to or relatable to the Retail Undertaking of the Demerged

* HEALTH * Hebrage * HAPPINESS

Company;

1.4.11 All such employees including contract employees of the Demerged Company, as are primarily engaged in or in relation to the business activities and operations pertaining to the Retail Undertaking of the Demerged Company, its respective offices, branches, or by its subsidiaries, etc, that are in the employment of the Demerged Company as of the Effective Date, and any other employees/personnel hired by the Transferor Company on and after the Demerger Appointed Date (as hereinafter defined) who are primarily engaged in or in relation to the business, activities and operations pertaining to its Retail Undertaking, that are in the employment of the Demerged Company as of the Effective Date;

Any question that may arise as to whether a specific asset or liability or any other property or employee pertains or does not pertain to the Retail Undertaking of the Demerged Company or whether it arises out of the activities or operations of the Retail Undertaking of the Demerged Company shall be decided by mutual agreement between the Board of Directors of the Demerged Company and the Resulting Company.

- 1.5 "Demerger Appointed Date" shall mean the close of business on 31 March, 2017.
- 1.6 "Effective Date" or "coming into effect of this Scheme" or "upon the Scheme becoming effective" means the date on which last of the actions set out in Clause 36 are fulfilled.
- 1.7 "Encumbrances" shall mean: (a) any mortgage, charge (whether fixed or floating), pledge, lien, hypothecation, assignment, deed of trust, title retention, security interest or other encumbrance or interest of any kind securing, or conferring any priority of payment in respect of any obligation of any Person, including without limitation any right granted by a transaction which, in legal terms, is not the granting of security but which has an economic or financial effect similar to the granting of security under Applicable Law, (b) proxy, any voting agreement, interest, option, right of first offer, refusal or transfer restriction in favor of any Person, or any other preferential arrangement having a similar effect, of any kind or nature, whether arising by agreement, by statute or otherwise,(c) any adverse claim as to title, possession or use, and (d) a contract to give or refrain from giving any of the foregoing.
- 1.8 "Governmental Authority" means any applicable central, state or local government (including Municipality, Municipal Corporation), statutory, legislative, regulatory or administrative authority, agency or commission or any court, tribunal, board, bureau or instrumentality thereof or arbitration or arbitral body having jurisdiction, exercising powers conferred by Applicable Law in India.
- 1.9 "HFL" or the "Transferor Company" means Heritage Foods Limited, a public limited company incorporated under the Companies Act, 1956 and having its registered office at #6-3-541 / C, Punjagutta, Hyderabad 500082.
- 1.10 "HFRL" or the "Transferee Company" or the "Demerged Company" means Heritage Foods Retail Limited, a public limited company incorporated under the Companies Act, 1956 and having its registered office at #6-3-541 / C, Punjagutta, Hyderabad – 500082.
- 1.11 "High Courts" means the High Court of Judicature at Hyderabad and the Mumbai High Court and/or, as the case may be, the National Company Law Tribunal.

* Habrage *

- 1.12 "Income-tax Act" means the Income-tax Act, 1961 and the rules framed thereunder, including any statutory modification, re-enactment or amendment thereto, for the time being in force.
- 1.13 "Permitted Encumbrances" shall mean the list of litigation pending in respect of the Retail Undertaking and Demerged Undertaking, more specifically set out in <u>Schedule I</u>.
- 1.14 "Person" shall mean any individual, entity, joint venture, company (including a limited liability company), corporation, partnership (whether limited or unlimited), proprietorship, trust or other enterprise (whether incorporated or not), Hindu undivided family, union, association of persons, government (central, state or otherwise), or any agency, department, authority or political subdivision thereof, and shall include their respective successors and in case of an individual shall include his/her legal representatives, administrators, executors and heirs and in case of a trust shall include the trustee or the trustees and the beneficiary or beneficiaries from time to time;
- 1.15 "Record Date" shall mean the date to be fixed by the Board of Directors of the Demerged Company or a committee thereof in consultation with the Board of Directors of the Resulting Company or a committee thereof for the purpose of determining the members of the Demerged Company to whom shares of the Resulting Company will be allotted pursuant to Part III of this Scheme in terms of Clause 26.1.1.
- 1.16 "Remaining Undertaking of the Demerged Company" means all the undertakings, business, activities and operations, including all the assets and liabilities of the Demerged Company (including but not limited to the VetCa Undertaking), excluding the Demerged Undertaking.
- 1.17 "Remaining Business of the Transferor Company" means all the undertakings, business, activities and operations, including all the assets and liabilities, of the Transferor Company, excluding the Retail Undertaking and the VetCa Undertaking. It is clarified that the dairy business and the renewable energy business, along with all their assets and liabilities, shall form part of the Remaining Business of the Transferor Company.
- 1.18 "Retail Undertaking" means the entire undertaking, business, activities and operations of Transferor Company, pertaining to: (i) grocery and food retail business, undertaken from its dedicated retail stores; (ii) sourcing, processing and marketing fresh fruits and vegetables; and (iii) manufacturing and supplying bakery products to the retail outlets of HFL and other customers, and which shall include:
 - 1.18.1 All assets (whether moveable or immoveable), wherever situated, whether leasehold or freehold, including land, building, plant and machinery, installations, equipments, capital works-in-progress, vehicles, furniture, fixtures, appliances, accessories, stocks, inventory, receivables, cash on hand, balance with banks (including bank fixed deposits), advances paid to any persons, loans, advances and deposits, of the Transferor Company with respect to the Retail Undertaking, along with all rights, title, liability and interest in connection therewith.
 - 1.18.2 All trade liabilities, obligations and debts, accruing or arising out of the business, activities or operations of the Retail Undertaking of the Transferor Company; whether secured or unsecured, present or future, raised or incurred, including obligations of every kind, nature and description whatsoever and howsoever arising or accruing, guarantees, advances from customers, bills payable and interest, in relation to the Retail Undertaking of the Transferor

* HEALTH * Hapstage * Further, it is clarified that apart from the abovementioned liabilities of the Transferor Company pertaining to the Retail Undertaking, any other liabilities of the Transferor Company shall not be allocated towards the Retail Undertaking of the Transferor Company.

- 1.18.3 All contracts, agreements, leases, memoranda of understanding, memoranda of agreements, arrangements, undertakings deeds, bonds, schemes, arrangements, sales orders, purchase orders or other instruments of whatsoever nature, whether written or otherwise, relating to the Retail Undertaking of the Transferor Company, along with all rights, title, liability and interest in connection therewith.
- 1.18.4 All trademarks, trade names, service marks, copyrights, patents, designs, databases, whether registered or not, used by the Transferor Company exclusively in the business, activities and operations of the Retail Undertaking.
- 1.18.5 All permits, licenses, registrations, certificates, consents, approvals, authorizations, no-objection certificates, quotas, rights (including rights under any agreement, contracts, applications, letter of intent or any other contract), subsidies, grants, exemptions, tax benefits, tax credits, refunds, quality certifications and approvals, product registrations, industrial and other licences, granted by any authority including from central government, state government, local authority, customs, central excise, income tax, service tax, sales tax, value added tax, Reserve Bank of India, department of Weights & Measures, Food Safety & Standards Authority of India, of the Transferor Company in relation to the Retail Undertaking, registrations.
- 1.18.6 All records, files, papers, manuals, data, sales and advertising materials, lists and other details of customers and suppliers, credit information, pricing information, whether in physical or electronic form, all pertaining to or relating to the Retail Undertaking of the Transferor Company;
- 1.18.7 All employees, staff and workers of the Transferor Company, as are primarily engaged in the Retail Undertaking.

Any question that may arise as to whether a specific asset or liability or any other property or employee pertains or does not pertain to the Retail Undertaking of the Transferor Company or whether it arises out of the activities or operations of the Retail Undertaking of the Transferor Company shall be decided by mutual agreement between the Board of the Transferor Company and the Transferee Company but with the written consent of the Resulting Company.

- 1.19 "Scheme of Arrangement" or "Scheme" means this Composite Scheme of Arrangement as submitted in the present form to the High Courts, with any modification(s) approved or imposed or directed by the High Courts or made pursuant to Clause 34 of this Scheme.
- 1.20 "SEBI" means the Securities and Exchange Board of India established under the Securities and Exchange Board of India Act, 1992;
- 1.21 "SEBI Circular" means Circular number CIR/CFD/CMD/16/2015 dated November 30, 2015 issued

* Hechage *

- 1.22 "SEBI LODR Regulations" means the SEBI (Listing Obligations and Disclosure Requirements) Regulations, 2015;
- 1.23 "Slump Sale Appointed Date" shall mean the commencement of business on November 1, 2016;
- 1.24 "Slump Sale" means sale of an undertaking on a going concern basis as defined under Section 2(42C) of the Income Tax Act, for a lump sum consideration without values being assigned to individual assets and liabilities;
- 1.25 "Stock Exchanges" means BSE Limited and the National Stock Exchange of India Limited;
- 1.26 "VetCa Undertaking" means the entire undertaking of HFL pertaining to its VetCa Undertaking and includes:
 - 1.26.1 All assets (whether moveable or immoveable) including freehold land, office premises, all other assets and properties (whether tangible or intangible, real or personal, corporeal or incorporeal, present, future or contingent) including, without limitation, interests, loans, deposits, advances (including accrued interest), investments including investments in overseas subsidiaries, receivables, cash on hand, investment in mutual funds, liquid funds, balance with banks (including bank fixed deposits), equipment, plant and machinery and the related assets and agreements, capital work in progress, unbilled revenue, furniture, fixtures, office equipment, appliances, accessories, vehicles, power connections, utilities and other service connections, all customer contracts, forward cover contracts, hedging contracts, receivables, claims, refunds, earnest moneys paid, rights and benefits under any agreements or security arrangements and funds, contingent rights, rights arising under contracts and all other rights, title, interests, privileges and benefits of every kind wherever located (including in the possession of vendors, third parties or elsewhere) and used or held, by the Transferor Company in, or otherwise identified for use in, or relating to, the business activities and operations pertaining to the VetCa Undertaking of the Transferor Company;
 - 1.26.2 All liabilities and all debts, guarantees, assurances, commitments, obligations, loans, and undertakings of any kind, nature and description whatsoever and howsoever arising, present or future and including, without limitation, borrowings, working capital facilities, advances from customers, unearned revenues, bills payable, interest, whether fixed, contingent or absolute, secured or unsecured, asserted or unasserted, matured or unmatured, liquidated or unliquidated, accrued or not accrued, known or unknown, due or to become due, whenever or however arising, (including, without limitation, whether arising out of any contract or tort based on negligence or strict liability) pertaining to the VetCa Undertaking of the Transferor Company;

Explanation: For the purpose of this Scheme, it is hereby clarified that the liabilities pertaining to the VetCa Undertaking of the Transferor Company shall include:

 liabilities, which accrue or arise out of the activities or operations of the VetCa Undertaking of the Transferor Company; and

* Laritage

- specific loans and borrowings raised, incurred and utilized for the activities or operations of the VetCa Undertaking of the Transferor Company.
- 1.26.3 All contracts, agreements, leases, memoranda of understanding, memoranda of agreements, arrangements, undertakings, whether written or otherwise, deeds, bonds, schemes, arrangements, sales orders, purchase orders or other instruments of whatsoever nature to which the Transferor Company is a party, relating to its VetCa Undertaking, or otherwise identified to be for the benefit of the same, approvals, electricity permits, telephone connections, building and parking rights, pending applications for consents or extension pertaining to the VetCa Undertaking of the Transferor Company;
- 1.26.4 All intellectual properties, labels, brands, trademarks, trade names, service marks, copyrights, patents, designs, software and computer programmes, databases, domain names, including those pending registrations and applications for brands, trademarks, labels, trade names, service marks, copyrights, patents, designs, software and computer programs, databases and domain names, used by the Transferor Company or held for use by the Transferor Company exclusively in the business, activities and operations of the VetCa Undertaking;
- 1.26.5 All permits, licenses, consents, approvals, authorizations, quotas, rights, powers, permissions, arrangements, assignments, sanctions, entitlements, allotments, exemptions, incentives, tax benefits, deferrals, subsidies, concessions, grants, claims, liberties, special status, benefits and privileges enjoyed or conferred upon or held or availed of by the Transferor Company in relation to or pertaining to its VetCa Undertaking, registrations, advantages, no-objection certificates, certifications, easements, and any waivers of the foregoing, issued by any legislative, executive or judicial unit of any Governmental or quasi-Governmental entity or any department, commission, board, agency, bureau, official or other regulatory, local (including Municipal), administrative or judicial authority, used or held for use by the Transferor Company in respect of business, activities and operations pertaining to its VetCa Undertaking;
- 1.26.6 All tax credits, including cenvat credits, refunds, reimbursements, claims, exemptions, benefits under service tax laws, value added tax (VAT), purchase tax, sales tax or any other duty or tax or cess or imposts under any Central or State law including sales tax deferrals, special economic zone benefits, excise duty benefits, tax deducted at source, right to carry forward and set-off unabsorbed losses, and depreciation, if any and exemptions, deductions, benefits and incentives under the Income-tax Act in respect of business, activities and operations pertaining to the VetCa Undertaking of the Transferor Company;
- 1.26.7 All rights, benefits and other interest, whether held in trust or otherwise, contracts, agreements, powers, engagements, arrangements of all kind, privileges and all other rights including title, interests, other benefits (including tax benefits), easements, privileges, liberties and advantages of whatsoever nature and wheresoever situate belonging to or in the ownership, possession, power or custody of or in the control of or vested in or granted in favour of or enjoyed by the Transferor Company, whether in India or abroad, all pertaining to the VetCa Undertaking of the Transferor Company;
- 1.26.8 All records, files, papers, manuals, data, sales and advertising materials, lists and other details of customers and suppliers, credit information, pricing information, whether in physical or

* HEALTH * Helitage * electronic form, all pertaining to the VetCa Undertaking of the Transferor Company;

1.26.9 All such employees including contract employees of the Transferor Company, as are primarily engaged in or in relation to the business activities and operations pertaining to the VetCa Undertaking of the Transferor Company its respective offices, branches, or by its subsidiaries, etc, and any other employees/personnel hired by the Transferor Company on and after the date hereof who are primarily engaged in or in relation to the business, activities and operations pertaining to its VetCa Undertaking;

Any question that may arise as to whether a specific asset or liability or any other property or employee pertains or does not pertain to the VetCa Undertaking of the Transferor Company or whether it arises out of the activities or operations of the VetCa Undertaking of the Transferor Company shall be decided by mutual agreement between the Board of Directors of the Transferor Company and the Transferee Company.

- 1.27 All terms and words not defined in this Scheme shall, unless repugnant or contrary to the context or meaning thereof, have the same meaning ascribed to them under the Act, the Income-tax Act, the Securities Contracts Regulation Act, 1956, the Securities and Exchange Board of India Act, 1992, the Depositories Act, 1996 and other applicable laws, rules, regulations, bye-laws, as the case may be or any statutory modification or re-enactment thereof from time to time.
- 1.28 References to clauses and recitals, unless otherwise provided, are to clauses and recitals of and to this Scheme.
- 1.29 The headings herein shall not affect the construction of this Scheme.

2. SHARE CAPITAL

2.1 HFL:

The authorised share capital and the issued, subscribed and fully paid-up share capital of HFL, as on September 30, 2016, was as follows:

Particulars	INR
Authorised Share Capital	50,00,00,000
4,80,00,000 equity shares of INR 10 each	48,00,00,000
20,00,000 preference shares of INR 10 each	2,00,00,000
Total	50,00,00,000
ssued, Subscribed & Fully Paid-up Share Capital	
2,31,99,000 equity shares of INR 10 each	23,19,90,000
Total	23,19,90,000

The shares of HFL are currently listed on the Stock Exchange.



2.2 HFRL:

The authorised share capital and the issued, subscribed and fully paid-up share capital of HFRL, as on September 30, 2016, was as follows:

Particulars	INR
Authorised Share Capital	
1,50,00,000 equity shares of INR 10 each	15,00,00,000
Total	15,00,00,000
Issued, Subscribed & Fully Paid-up Share Capital	
1,65,600 equity shares of INR 10 each	16,56,000
Total	16,56,000

The entire paid up share capital of HFRL is held by HFL and its nominees.

2.3 FRL:

The authorised share capital and the issued, subscribed and fully paid-up share capital of FRL, as on September 30, 2016, was as follows:

Particulars	INR
Authorised Share Capital	
2,50,00,00,000 Equity Shares of INR 2 each	25,00,00,00,000
Total	25,00,00,00,000
Issued Share Capital	
47,14,42,928 Equity Shares of INR 2 each	94,28,85,856
Subscribed and Paid-up Share Capital	
47,13,38,557 Equity Shares of INR 2 each	94,26,77,114

The shares of FRL are currently listed on the Stock Exchange.

3. DATE OF TAKING EFFECT AND OPERATIVE DATE

The Scheme shall be operative from the Effective Date, but shall be effective from and be implemented with effect from the Slump Sale Appointed Date and the Demerger Appointed Date, as may be applicable.



PART II

TRANSFER OF THE RETAIL UNDERTAKING AND THE VETCA UNDERTAKING BY WAY OF SLUMP SALE TO HFRL

4. TRANSFER OF THE RETAIL UNDERTAKING AND THE VETCA UNDERTAKING

- 4.1 On the coming into effect of this Scheme and with effect from the Slump Sale Appointed Date, each of the Retail Undertaking and the VetCa Undertaking together with their respective assets, properties, liabilities, rights, benefits and interests therein, subject to existing charges if any, thereon, shall, without any further deed, act, matter or thing, stand transferred to and vested with Transferee Company pursuant to Sections 391 to 394 and other applicable provisions of the Companies Act, 1956 on a going concern and a 'Slump Sale' basis for lump sum consideration as set out hereinafter in this Part II of the Scheme, free from all Encumbrances and litigations, except the Permitted Encumbrances.
- 4.2 The transfer of each of the Retail Undertaking and the VetCa Undertaking under this Scheme is in compliance with the Income-tax Act, specifically Section 2(42C) and other relevant provisions. If any of the terms of this Scheme are inconsistent with the provisions of Sections 2(42C) of the Income-tax Act, the provisions of Sections 2(42C) of the Income-tax Act shall to the extent of such inconsistency, prevail and the Scheme shall, stand and be deemed to be modified to that extent to comply with the said provisions and such modifications shall not affect the other parts of the Scheme.

ASSETS AND LICENCES

- 5.1 Without limiting the generality of Clause 4.1, upon the Scheme becoming effective, and with effect from the Slump Sale Appointed Date:
- 5.1.1 All the immovable properties (including land, building and other immovable property) of each of the Retail Undertaking and VetCa Undertaking shall stand transferred to, and be vested in, the Transferee Company, pursuant to the provisions of Sections 391 to 394 of Companies Act, 1956 and all other applicable provisions, if any, of the Act, without requiring any deed or instrument of conveyance and shall upon such transfer become the property of the Transferee Company so as to vest in the Transferee Company all the rights, title and interest in such immovable properties. Such immovable property transferred shall be free from all Encumbrances except the Permitted Encumbrances. With effect from the Slump Sale Appointed Date, the Transferee Company shall be entitled to exercise all rights and privileges, and be liable to pay all taxes and charges, and fulfill all obligations, in relation to or applicable to such immovable properties.
- 5.1.2 All the assets of each of the Retail Undertaking and the VetCa Undertaking as are movable in nature or are otherwise capable of transfer by endorsement and delivery, shall stand transferred to, and be vested in, the Transferee Company, pursuant to the provisions of Sections 391 to 394 of the Companies Act, 1956 and all other applicable provisions, if any, of the Act, without requiring any deed or instrument of conveyance and shall upon such transfer become the property of the Transferee Company so as to vest in the Transferee Company all the rights, title and interest in such assets. The transfer or vesting pursuant to this sub-Clause shall be deemed to have occurred by physical delivery or endorsement and delivery, as appropriate to the property being transferred/ vested and the title to

* Heritage *

- such property shall be deemed to have been transferred and vested accordingly. Such property transferred shall be free from all Encumbrances except the Permitted Encumbrances.
- 5.1.3 All the intellectual property which relate exclusively to the Retail Undertaking and the VetCa Undertaking, shall stand transferred to, and be vested in, the Transferee Company, pursuant to the provisions of Sections 391 to 394 of Companies Act, 1956 and all other applicable provisions, if any, of the Act, without requiring any act or deed to be done by the Transferor Company and / or the Transferee Company.
- 5.1.4 All the assets (other than those specified in sub-clauses 5.1.1, 5.1.2 and 5.1.3 above) of each of the Retail Undertaking and the VetCa Undertaking, including sundry debtors, receivables, bills, outstanding loans and advances, bank balances, deposits, etc., the same shall stand transferred to and vested in the Transferee Company, pursuant to the provisions of Sections 391 to 394 of Companies Act, 1956 and all other applicable provisions, if any, of the Act, without requirement of any notice or other intimation to any person, to the end and intent that right of the Transferor Company to recover or realize the same stands transferred to the Transferee Company. The Transferee Company shall at its sole and absolute discretion, and without being obliged and if it so deems appropriate, give notice in such form as it may deem fit and proper to each person, debtor or depositee that pursuant to the Scheme sanctioned by the High Courts, the said debt, receivable, bill, loan, advance or deposit stands transferred and vested in the Transferee Company and the same be paid to or made good to or held on account of the Transferee Company
- 5.1.5 All licences, permissions, approvals, consents, certificates, registrations, no-objections, clearances, concessions, exemptions or rights granted to, issued to or executed in favour of the Transferor Company in relation to the Retail Undertaking and/or the VetCa Undertaking, shall, pursuant to the provisions of Sections 391 to 394 of Companies Act, 1956 and all other applicable provisions, if any, of the Act, stand vested in or transferred to or deemed amended in favour of the Transferee Company as if the same were originally granted to, issued to or executed in favour of the Transferee Company, and shall be appropriately transferred or assigned by the concerned statutory authorities in favour of the Transferee Company upon vesting of the Retail Undertaking and / or the VetCa Undertaking, as the case may be, pursuant to this Scheme. The benefit of all statutory and regulatory permissions, environmental approvals and consents, registration or other licences, and consents shall vest in and become available to the Transferee Company pursuant to the Scheme.
- 5.1.6 All the existing Encumbrances, if any, on the assets of the Retail Undertaking and / or the VetCa Undertaking relating to the liabilities of the Retail Undertaking and / or the VetCa Undertaking, shall, after the Slump Sale Appointed Date, continue to relate and attach to only such assets or any part thereof to which they are related or attached. Further, the Encumbrances, if any, on the assets of the Remaining Business of the Transferor Company in relation to the liabilities of the Retail Undertaking and / or the VetCa Undertaking, shall without any further act, instrument or deed be released and discharged from such Encumbrance.
- 5.1.7 All the existing Encumbrances, if any, on the assets of the Retail Undertaking and / or the VetCa Undertaking relating to the liabilities of the Remaining Business of the Transferor Company, shall without any further act, instrument or deed be released and discharged from such Encumbrance.
- 5.1.8 The Transferee Company shall be entitled to benefit of all insurance policies which have been issued in relation to the Retail Undertaking and / or VetCa Undertaking and the name of the Transferee

* HEALTH * HAPPINESS Company shall be substituted as the "insured party" in the policies as if the Transferee Company was initially a party.

6. LIABILITIES

Without limiting the generality of Clause 4.1, upon the Scheme becoming effective, and with effect from the Slump Sale Appointed Date, all secured and unsecured debts, sundry creditors, liabilities, contingent liabilities, guarantees, duties and obligations of every kind, nature and description (whether in Indian Rupees or foreign currency) whatsoever and howsoever arising, raised or incurred or utilised by the Transferor Company in relation to each of the Retail Undertaking and the VetCa Undertaking, shall become and be the debts, liabilities, guarantees, duties and obligations of the Transferee Company along with any charge, lien, encumbrance or security thereon, pursuant to the provisions of Sections 391 to 394 of Companies Act, 1956 and all other applicable provisions, if any, of the Act, without requirement of any further act, instrument, matter, thing or deed. It shall not be necessary to obtain the consent of any third party or other person who is a party to any contract or arrangement by virtue of which such debts, liabilities, contingent liabilities, duties and obligations have arisen in order to give effect to the provisions of this sub-clause. Further, all debts and loans raised and duties, liabilities and obligations incurred or which arise or accrue to the Transferor Company in relation to each of the Retail Undertaking and the VetCa Undertaking on or after the Slump Sale Appointed Date till the Effective Date, shall be deemed to have been raised, used or incurred for and on behalf of the Transferee Company and to the extent they are outstanding on the Effective Date, shall also without any further act or deed be and stand transferred to and be deemed to be transferred to the Transferee Company and shall become the debts, loans raised, duties, liabilities and obligations incurred by the Transferee Company by virtue of this Scheme. Where any of the debts, liabilities, loans raised and used, liabilities and obligations incurred, duties and obligations of the Transferor Company in relation to each of the Retail Undertaking and the VetCa Undertaking as on the Slump Sale Appointed Date deemed to be transferred to the Transferee Company have been discharged by the Transferor Company after the Slump Sale Appointed Date and prior to the Effective Date, such discharge shall be deemed to have been for and on account of the Transferee Company.

EMPLOYEES, STAFF AND WORKMEN

- 7.1 Without limiting the generality of Clause 4.1, upon the Scheme becoming effective, and with effect from the Slump Sale Appointed Date, all employees, staff and workmen of the Transferor Company engaged in the Retail Undertaking and the VetCa Undertaking and who are in such employment as on the Effective Date shall become employees of the Transferee Company from the Slump Sale Appointed Date or their respective joining date, whichever is later, without any break or interruption in their service and on the basis of continuity of service, and the terms and conditions of their employment with the Transferee Company shall be no less favourable than those on which they are engaged in the Transferor Company.
- 7.2 The Transferee Company agrees that the services of all the employees of each of the Retail Undertaking and the VetCa Undertaking prior to the transfer, as aforesaid, shall be taken into account for the purposes of all benefits to which such employees may be eligible and accordingly, the period of service of such employees shall be reckoned therefore from the date of their respective appointment in the Transferor Company.



- 7.3 In the event of retrenchment of the employees of the Retail Undertaking and/or the VetCa Undertaking, the Transferee Company will be liable to pay compensation in accordance with law on the basis that the services of the employees shall have been continuous and shall not have been interrupted by reason of such Slump Sale.
- 7.4 Till the Effective Date of this Scheme, the Transferor Company shall make contributions to the government maintained provident fund and / or other funds in relation to the staff, workmen and employees of each of the Retail-Undertaking and the VetCa Undertaking. On and from the Effective Date, the Transferee Company shall make appropriate contributions to such provident fund and/or other funds in respect of the staff, workmen and employees taken over by it pursuant to this Scheme. The contributions, and all accretions thereto, in the provident fund account, superannuation fund, gratuity fund and other benefit funds if any, of which such employees are members or beneficiaries till the Effective Date, shall, with the approval of the concerned authorities be transferred (in such proportion as is allocable to the employees of the Retail Undertaking and/or the VetCa Undertaking being transferred to the Transferee Company) to the relevant funds of the Transferee Company for the benefit of the employees of the Retail Undertaking and/or the VetCa Undertaking on terms no less favourable. In the event that the Transferee Company has its own funds in respect of any of the funds referred to above, such investments shall, subject to the necessary approvals and permissions, be transferred to the relevant funds. In the event that the Transferee Company does not have its own fund in respect of any of the aforesaid matters, the Transferor Company may, subject to necessary approvals and permissions, continue to contribute in respect of the employees engaged in the Retail Undertaking and/or the VetCa Undertaking to the existing funds, until such time that the Transferee Company creates its own fund, at which time the investments and contributions pertaining to the employees of the Retail Undertaking and/or the VetCa Undertaking shall be transferred to the funds created by the Transferee Company. In case, necessary approvals are not received and there is delay, all such amounts shall continue to be administered by the Transferor Company in trust for the Transferee Company from the Effective Date till the date of actual transfer and, on receiving the approvals all the accumulated amounts till such date, shall be transferred to the respective funds of the Transferee Company suo moto.
- 7.5 Any disciplinary action initiated by the Transferor Company against any employee of the Retail Undertaking and/or the VetCa Undertaking shall have full force, effect and continuity as if it was initiated by the Transferee Company instead of the Transferor Company.
- 7.6 Any question that may arise as to whether any employee belongs or does not belong to the Retail Undertaking shall be decided by the Board of the Companies.
- 7.7 Any question that may arise as to whether any employee belongs or does not belong to the VetCa Undertaking shall be decided by the Board of the Transferor Company and the Transferee Company.

8. CONTRACTS AND DEEDS

8.1 Without limiting the generality of Clause 4.1, upon the Scheme becoming effective, and with effect from the Slump Sale Appointed Date, subject to the other provisions of this Scheme, all contracts, deeds, bonds, agreements, arrangements and other instruments (including all leases, licenses and other assurances in favour of the Transferor Company or powers or authorities granted by or to it) of whatsoever nature to which the Transferor Company is a party or to the benefit of which the Transferor Company may be eligible, all in relation to the Retail Undertaking and/or the VetCa

* HEALTH HAPPINESS Undertaking and which are subsisting or having effect immediately before the Effective Date, shall, without any further act, instrument or deed, continue in full force and effect in favour of, by, for or against the Transferee Company and may be enforced as fully and effectually as if, instead of the Transferor Company, the Transferee Company had been a party or beneficiary or obligee or obligor thereto or thereunder. It is hereby clarified that upon the Scheme becoming effective, and with effect from the Slump Sale Appointed Date, the Transferor Company shall have no liabilities in respect of any of the aforesaid contracts / arrangements transferred to the Transferee Company.

- 8.2 Without prejudice to the other provisions of this Scheme and notwithstanding the fact that the vesting of the Retail Undertaking and/or the VetCa Undertaking of the Transferor Company in the Transferoe Company occurs by virtue of this Scheme itself, the Transferoe Company may, at any time after the coming into effect of this Scheme, in accordance with the provisions hereof, if so required, under any law or otherwise, take such actions or enter into, or issue or execute deeds, writings, confirmations, novations, declarations, or other documents with, or in favour of, any party to any contract or arrangement to which the Transferor Company is a party or any writings as may be necessary to be executed in order to give formal effect to the provisions of this Scheme. The Transferoe Company shall, under the provisions of this Scheme, be deemed to be authorised to execute any such writings on behalf of the Transferor Company and to carry out or perform all such formalities or compliances required for the purposes referred to above.
- 8.3 For avoidance of doubt and without prejudice to the generality of the foregoing, it is clarified that upon the coming into effect of this Scheme, all consents, permissions, certificates, authorities, powers of attorney given by, issued to or executed in favour of the Transferor Company in relation to the Retail Undertaking and/or the VetCa Undertaking shall stand transferred to the Transferee Company, as if the same were originally given by, issued to or executed in favour of the Transferee Company, and the Transferee Company shall be bound by the terms thereof, the obligations and duties thereunder and the rights and benefits under the same shall be available to the Transferee Company.
- 8.4 Without prejudice to the aforesaid, it is clarified that if any contracts, deeds, bonds, agreements, schemes, arrangements or other instruments of whatsoever nature in relation to the Retail Undertaking and/or the VetCa Undertaking which the Transferor Company owns or to which the Transferor Company is a party, cannot be transferred to the Transferee Company for any reason whatsoever, the Transferor Company shall hold such assets, contracts, deeds, bonds, agreements, schemes, arrangements or other instruments of whatsoever nature in trust for the benefit of the Transferee Company, in so far as it is permissible so to do, till such time as the transfer is given effect to.

LEGAL PROCEEDINGS

9.1 Without limiting the generality of Clause 4.1, upon the Scheme becoming effective, and with effect from the Slump Sale Appointed Date, all Permitted Encumbrances of whatever nature pending and / or arising after the Slump Sale Appointed Date, in any court or before any authority, judicial, quasi-judicial or administrative or any adjudicating authority or any arbitral tribunal, by or against the Transferor Company in relation to the Retail Undertaking and/or the VetCa Undertaking shall be continued and/or enforced until the Effective Date as desired by the Transferor Company, and on and from the Effective Date, the same shall not abate or be discontinued or in any way be prejudicially affected by reason of this Slump Sale or by anything contained in this Scheme, but the said suits, appeals or other legal proceeding shall be continued and/or enforced by or against the Transferee

* HEALTH * HEALTH * HAPPINESS Company as effectually and in the same manner and to the same extent as if the same had been originally instituted and/or had arisen and/or were pending by or against the Transferee Company.

- 9.2 On and from the Effective Date, the Transferee Company shall have the right to initiate, defend, compromise or otherwise deal with any legal proceedings in relation to the Permitted Encumbrances relating to the Retail Undertaking and / or the VetCa Undertaking, in the same manner and to the same extent as it would or might have been initiated by the Transferor Company, as the case may be, had the Scheme not been made.
- 9.3 On and from the Slump Sale Appointed Date, if any proceedings in relation to the Permitted Encumbrances are taken by or against the Transferor Company in relation to the Retail Undertaking and/or the VetCa Undertaking, the Transferor Company shall till the Effective Date continue and/or defend the same at the cost of the Transferor Company, and the Transferoe Company shall reimburse and indemnify the Transferor Company against all liabilities and obligations incurred by the Transferor Company in respect thereof.
- 9.4 The Transferee Company undertakes to have all legal or other proceedings in relation to the Permitted Encumbrances initiated by or against the Transferor Company referred to in Clause 9.1 above transferred to its name on and after the Effective Date, and to have the same continued, prosecuted and enforced by or against the Transferee Company as the case may be, to the exclusion of the Transferor Company.
- 9.5 Notwithstanding the above, in case the proceedings referred to in Clause 9.1 above cannot be transferred for any reason, or the transfer takes time, till such transfer the Transferor Company shall defend the same in accordance with the advice of the Transferee Company and at the cost of the Transferee Company, and the Transferee Company shall reimburse, indemnify and hold harmless the Transferor Company against all liabilities and obligations incurred by the Transferor Company in respect thereof.

TAXES

- 10.1 Upon the Scheme becoming effective, and with effect from the Slump Sale Appointed Date, all taxes, duties, cess of any nature (including income-tax, sales tax, excise duty, customs duty, service tax, VAT etc.) paid or payable, including any tax deduction or collection at source, service tax input credit receivables, by the Transferor Company in relation to the Retail Undertaking and the VetCa Undertaking and relating to the period after the Slump Sale Appointed Date until the Effective Date, shall be deemed to have been on account of or on behalf of or paid or payable by the Transferee Company and shall, in all proceedings, be dealt with accordingly.
- 10.2 On the Scheme becoming effective, the Transferor Company and the Transferee Company may revise their respective returns pertaining to income tax, service tax, sales tax, VAT and other tax returns, and claim refunds and/or credits, including credits for tax deducted at source, as applicable pursuant to the provisions of this Scheme.

11. CONDUCT OF BUSINESS TILL THE EFFECTIVE DATE

11.1 With effect from the Slump Sale Appointed Date and till the Effective Date:

* HEALTH * HAPPINESS

- 11.1.1 The Transferor Company shall carry on, and shall be deemed to have carried on, all the business, activities and operations relating to the Retail Undertaking and/or the VetCa Undertaking, and shall hold and stand possessed of and shall be deemed to have held and stood possessed of the assets, properties and liabilities of each of the Retail Undertaking and/or the VetCa Undertaking, on account of and / or on behalf of and / or for the benefit of and / or in trust for, the Transferee Company.
- 11.1.2 All the profits or incomes accruing or arising and all expenditure or losses arising or incurred (including all taxes, if any, paid or accruing in respect of any profits and income) by the Transferor Company in relation to each of the Retail Undertaking and the VetCa Undertaking shall, for all purposes, be treated and be deemed to be and accrue as the profits or incomes, or as the case may be, expenditure or losses (including taxes) of, the Transferee Company.
- 11.1.3 Any of the rights, powers, authorities and privileges attached or related or pertaining to each of the Retail Undertaking and the VetCa Undertaking and exercised by or available to the Transferor Company, shall be deemed to have been exercised for and on behalf of and as an agent for the Transferee Company. Further, any of the obligations, duties and commitments attached, relating or pertaining to each of the Retail Undertaking and the VetCa Undertaking that have been undertaken or discharged by the Transferor Company shall be deemed to have been undertaken or discharged for and on behalf of and as an agent for the Transferee Company.

12. SAVING OF CONCLUDED TRANSACTIONS

12.1 Subject to the terms of this Scheme, the transfer and vesting of the Retail Undertaking and the VetCa Undertaking under this Scheme shall not affect any transactions or proceedings already concluded on or after the Slump Sale Appointed Date till the Effective Date, to the end and intent that the Transferee Company accepts and adopts all acts, deeds and things made, done and executed by the Transferor Company in relation to the Retail Undertaking and/or the VetCa Undertaking as acts, deeds and things made, done and executed by or on behalf of the Transferee Company.

13. CONSIDERATION

- 13.1 In consideration of the transfer of and vesting of the Retail Undertaking and the VetCa Undertaking with the Transferee Company in accordance with this Scheme by way of slump sale as defined under the provisions of section 2(42C) of the Income-tax Act, the Transferee Company shall pay a consideration of INR 135,00,00,000/- (Rupees One Hundred and Thirty Five Crores), subject to adjustment of net working capital between the Slump Sale Appointed Date and the Effective Date, which shall be discharged in the manner specified in Clause 13.2.
- 13.2 The consideration would be discharged by the Transferee Company, without any further application, deed, action or thing, by way of issuance and allotment of 1,40,00,000 (One Crore Forty Lakhs) equity shares of the Transferee Company, each of a face value of INR 10 (Rupees Ten only) and a premium of INR 86.43 (Rupees Eighty Six and Paisa Forty Three), credited as fully paid-up to the Transferor Company.
- 13.3 Upon the issuance of the Equity Shares as per Clause 13.1, the issued, subscribed and paid-up share capital of the Transferee Company shall stand increased to INR 14,16,56,000 (Rupees Fourteen Crores Sixteen Lakhs Fifty Six Thousand) comprising of 1,41,65,600 (One Crore Forty One Lakhs Sixty Five Thousand and Six Hundred only) equity shares having a face value of INR 10 (Rupees Ten

* HEALTH * Hebrage * only).

13.4 The approval of this Scheme by the shareholders of the Transferee Company and the Transferor Company, under Sections 391 to 394 of the Companies Act, 1956 and other applicable provisions of the Act, shall also be deemed to be the approval by the shareholders under the provisions of Section 62 of the Companies Act, 2013 and all other applicable provisions of the Act and applicable law for the purpose of subscription and issuance and allotment of the Equity Shares of the Transferee Company to the Transferor Company in accordance with the Scheme. It is clarified that no additional special resolution under Section 62(1)(c) of the Companies Act, 2013 or any other provision of the Act or applicable law shall be required to be passed for issuance and allotment of the equity shares of the Transferee Company to the Transferor Company under this Scheme.

14. REMAINING BUSINESS OF THE TRANSFEROR COMPANY

- 14.1 The Remaining Business of the Transferor Company including all the properties and assets, investments including investments in subsidiaries, debts, liabilities and obligations of the Transferor Company relating to the Remaining Business of the Transferor Company and which do not form part of the Retail Undertaking or the VetCa Undertaking shall continue to belong to and remain vested in the Transferor Company.
- 14.2 The Transferor Company shall be entitled to carry on its business and activities pertaining to the Remaining Business of the Transferor Company in such manner as it may deem fit and proper and nothing herein contained shall affect the business and activities of the Transferor Company in relation to the Remaining Business. Further, the Transferor Company shall be deemed to have been carrying on all business and activities relating to the Remaining Business of the Transferor Company for and on its own behalf.
- 14.3 All assets and properties acquired by the Transferor Company at any time including on and after the start of business on the Slump Sale Appointed Date shall, to the extent that the same do not relate to the Retail Undertaking or the VetCa Undertaking, form part of the Remaining Business.
- 14.4 All liabilities, debts and obligations incurred by or arising against the Transferor Company at any time including on and after the start of business on the Slump Sale Appointed Date shall, to the extent that the same do not relate to the Retail Undertaking or the VetCa Undertaking, form part of the Remaining Business of the Transferor Company.
- 14.5 The Transferor Company shall be entitled to enter into such contracts as the Transferor may deem fit and proper in respect of the Remaining Business of the Transferor Company.
- 14.6 All the profits or incomes accruing or arising and all expenditure or losses arising or incurred (including all taxes, if any, paid or accruing in respect of any profits and income) by the Transferor Company in relation to Remaining Business of the Transferor Company shall, for all purposes, be treated and be deemed to be and accrue as the profits or incomes, or as the case may be, expenditure or losses (including taxes) of, the Transferor Company.
- 14.7 All the legal and other proceedings by or against the Transferor Company under any statute, whether pending on the Slump Sale Appointed Date or which may be instituted after the Slump Sale Appointed Date, relating to the Remaining Business of the Transferor Company shall be continued

* Holstage *

and enforced by or against the Transferor Company.

15. ACCOUNTING TREATMENT IN THE BOOKS OF THE TRANSFEREE COMPANY

- 15.1 Upon the Scheme becoming effective, the Transferee Company shall record the assets and liabilities comprised in the Retail Undertaking and VetCa Undertaking of the Transferor Company transferred to the Transferee Company pursuant to this Scheme, by undertaking a purchase price allocation for the Slump Sale consideration to the respective assets and liabilities based upon the values determined by an independent valuer or by the Board of the Transferee Company.
- 15.2 The Transferee Company shall credit its share capital account in its books of account with the aggregate face value of the equity shares issued to the Transferor Company pursuant to Clause 13.2 of the Scheme. Further, the Transferee Company shall credit to its securities premium account, the aggregate premium on securities issued by it pursuant to Clause 13.2 of the Scheme. The securities premium account recorded by the Transferee Company shall be applied as per the provisions of Section 52 of Companies Act, 2013.
- 15.3 The difference, if any, in the value of consideration and net value of assets and liabilities of the Retail Undertaking and the VetCa Undertaking, as determined under Clause 15.1 above, shall be accounted in accordance with principles as laid down in the applicable accounting standards, the applicable provisions of the Act and generally accepted accounting principles in India.

16. ACCOUNTING TREATMENT IN THE BOOKS OF THE TRANSFEROR COMPANY

- 16.1 Upon the Scheme becoming effective, the Transferor Company shall transfer the Retail Undertaking and the VetCa Undertaking on a going concern basis along with all their assets, liabilities, rights and obligations as defined in Clause 1.19 and Clause 1.25 respectively of this Scheme, to the Transferee Company.
- 16.2 The Transferor Company shall reduce the book value of assets and liabilities comprised in the Retail Undertaking and the VetCa Undertaking from its books of account.
- 16.3 The aggregate value of the equity shares allotted under Clause 13.2 above, along with the premium shall be debited to the investments account.
- 16.4 The excess or shortfall of aggregate value of the equity shares allotted under Clause 13.2 above along with the premium over the book value of Retail Undertaking and VetCa Undertaking and directly attributable transaction cost shall be adjusted to the capital reserve account.

PART III

TRANSFER AND VESTING OF DEMERGED UNDERTAKING TO AND IN THE RESULTING COMPANY

TRANSFER OF DEMERGED UNDERTAKING

17.1 For the purposes of this Part III, "after giving effect to Part II of the Scheme" will be determined

* HEALTH * Hestage * mutually by the Board of Directors of the Demerged Company and the Resulting Company.

- 17.2 With effect from the Demerger Appointed Date and after giving effect to Part II of the Scheme, the Demerged Undertaking, together with its assets, properties, liabilities, rights, benefits and interests therein, shall, pursuant to the provisions of Sections 391 to 394 of the Companies Act, 1956 and without any further act, instrument, deed, matter or thing, be and stand transferred to and vested in the Resulting Company, free of charges, on a going concern basis in consideration for the issuance of the Resulting Company's shares as set out hereinafter in this Part III of the Scheme, so as to vest in the Resulting Company all rights, title and interest pertaining to the Demerged Undertaking, free from all Encumbrances, except the Permitted Encumbrances.
- 17.3 The demerger of the Demerged Undertaking under this Scheme shall be in compliance with the conditions of "demerger" as specified under Section 2(19AA) of the Income-tax Act and other relevant provisions, i.e. transfer of all assets and liabilities relating to the Demerged Undertaking at values appearing in the books of accounts of Demerged Company immediately before the demerger, issue of shares to shareholders of Demerged Company on a proportionate basis, etc...

ASSETS AND LICENCES

- 18.1 Without prejudice to the generality of Clause 17.1 above, upon the Scheme becoming effective, and with effect from the Demerger Appointed Date and after giving effect to Part II of the Scheme:
- 18.1.1 All the immovable properties (including land, building and other immovable property) of the Demerged Undertaking shall stand transferred to, and be vested in, the Resulting Company, pursuant to the provisions of Sections 391 to 394 of the Companies Act, 1956 and all other applicable provisions, if any, of the Act, without requiring any deed or instrument of conveyance and shall upon such demerger, become the property of the Resulting Company so as to vest in the Resulting Company all the rights, title and interest in such immovable properties, on and from the Demerger Appointed Date and after giving effect to Part II of this Scheme. Such immovable property transferred shall be free from all Encumbrances except Permitted Encumbrances. With effect from the Demerger Appointed Date and after giving effect to Part II of this Scheme, the Resulting Company shall be entitled to exercise all rights and privileges, and be liable to pay all taxes and charges, and fulfil all obligations, in relation to or applicable to such immovable properties.
- 18.1.2 All the assets of the Demerged Undertaking as are movable in nature or are otherwise capable of transfer by endorsement and delivery, shall stand transferred to, and be vested in, the Resulting Company, pursuant to the provisions of Sections 391 to 394 of the Companies Act, 1956 and all other applicable provisions, if any, of the Act, without requiring any deed or instrument of conveyance and shall upon such transfer become the property of the Resulting Company so as to vest in the Resulting Company all the rights, title and interest in such assets, on and from the Demerger Appointed Date and after giving effect to Part II of the Scheme. The transfer or vesting pursuant to this sub-Clause shall be deemed to have occurred by physical delivery or endorsement and delivery, as appropriate to the property being transferred/ vested and the title to such property shall be deemed to have been transferred and vested accordingly. Such property transferred shall be free from all Encumbrances except Permitted Encumbrances.
- 18.1.3 All the intellectual property relating exclusively to the Demerged Undertaking, shall stand transferred to, and be vested in, the Resulting Company, pursuant to the provisions of Sections 391 to 394 of

* Hebitage *

Companies Act, 1956 and all other applicable provisions, if any, of the Act, without requiring any act or deed to be done by the Demerged Company and / or the Resulting Company, on and from the Demerger Appointed Date and after giving effect to Part II of the Scheme.

- 18.1.4 All the assets (other than those specified in sub-Clause 18.1.2 and 18.1.3 above) of the Demerged Undertaking, including sundry debtors, receivables, bills, outstanding loans and advances, bank balances, deposits, etc., the same shall stand transferred to and vested in the Resulting Company, pursuant to the provisions of Sections 391 to 394 of Companies Act, 1956 and all other applicable provisions, if any, of the Act, without requirement of any notice or other intimation to any person, to the end and intent that the right of the Demerged Company to recover or realize the same stands transferred to the Resulting Company, on and from the Demerger Appointed Date and after giving effect to Part II of the Scheme. The Resulting Company shall at its sole and absolute discretion, and without being obliged and if it so deems appropriate, give notice in such form as it may deem fit and proper to each person, debtor or depositee that pursuant to the Scheme sanctioned by the High Courts, the said debt, receivable, bill, loan, advance or deposit stands transferred and vested in the Resulting Company and the same be paid to or made good to or held on account of the Resulting Company.
- 18.1.5 All licences, permissions, approvals, consents, certificates, registrations, no-objections, clearances, concessions, exemptions or rights granted to, issued to or executed in favour of the Demerged Company in relation to the Demerged Undertaking, shall, pursuant to the provisions of Sections 391 to 394 of the Companies Act, 1956 and all other applicable provisions, if any, of the Act, stand vested in or transferred to or deemed amended in favour of the Resulting Company as if the same were originally granted to, issued to or executed in favour of the Resulting Company, and shall be appropriately transferred or assigned by the concerned statutory authorities in favour of the Resulting Company upon vesting of the Demerged Undertaking on and from the Demerger Appointed Date and after giving effect to Part II of the Scheme. The benefit of all statutory and regulatory permissions, environmental approvals and consents, registration or other licences, and consents shall vest in and become available to the Demerged Company pursuant to the Scheme, on and from the Demerger Appointed Date.
- 18.1.6 All the existing Encumbrances other than the Permitted Encumbrances, on the assets of the Demerged Undertaking shall be released. Such Encumbrances shall not relate to or attach to any assets of the Demerged Undertaking or any other assets of the Resulting Company.
- 18.1.7 The Encumbrances, other than the Permitted Encumbrances, shall after the Effective Date relate and attach to the assets or any part thereof of the Remaining Undertaking of the Demerged Company.
- 18.1.8 The Resulting Company shall be entitled to benefit of all insurance policies which have been issued in relation to the Demerged Undertaking and the name of the Resulting Company shall be substituted as the "insured party" in the policies as if the Resulting Company was initially a party.

LIABILITIES

19.1 Without limiting the generality of Clause 17.1, upon the Scheme becoming effective, and with effect from the Demerger Appointed Date and after giving effect to Part II of this Scheme, all secured and unsecured debts, sundry creditors, liabilities, contingent liabilities, guarantees, duties and obligations of every kind, nature and description (whether in Indian Rupees or foreign currency) whatsoever and howsoever arising, raised or incurred or utilised by the Demerged Company in relation to each of the

* HEALTH HAPPINESS

Demerged Undertaking, shall become and be the debts, liabilities, guarantees, duties and obligations of the Resulting Company along with any charge, lien, encumbrance or security thereon, pursuant to the provisions of Sections 391 to 394 of Companies Act, 1956 and all other applicable provisions, if any, of the Act, without requirement of any further act, instrument, matter, thing or deed. It shall not be necessary to obtain the consent of any third party or other person who is a party to any contract or arrangement by virtue of which such debts, liabilities, contingent liabilities, duties and obligations have arisen in order to give effect to the provisions of this sub-clause. Further, all debts and loans raised and duties, liabilities and obligations incurred or which arise or accrue to the Demerged Company in relation to the Demerged Undertaking on or after the Demerger Appointed Date till the Effective Date, shall be deemed to have been raised, used or incurred for and on behalf of the Resulting Company and to the extent they are outstanding on the Effective Date, shall also without any further act or deed be and stand transferred to and be deemed to be transferred to the Resulting Company and shall become the debts, loans raised, duties, liabilities and obligations incurred by the Resulting Company by virtue of this Scheme. Where any of the debts, liabilities, loans raised and used, liabilities and obligations incurred, duties and obligations of the Demerged Company in relation to the Demerged Undertaking as on the Demerger Appointed Date deemed to be transferred to the Resulting Company have been discharged by the Demerged Company after the Demerger Appointed Date and prior to the Effective Date, such discharge shall be deemed to have been for and on account of the Resulting Company.

20. EMPLOYEES, STAFF & WORKMEN

- 20.1 Without limiting the generality of Clause 17.1, upon the Scheme becoming effective, and with effect from the Demerger Appointed Date and after giving effect to Part II of the Scheme, all employees, staff and workmen of the Demerged Company engaged in the Demerged Undertaking and who are in such employment as on the Effective Date, shall become employees, staff and workmen of the Resulting Company from the Demerger Appointed Date or their respective joining date, whichever is later, without any break or interruption in their service and on the basis of continuity of service, and the terms and conditions of their employment with the Resulting Company shall be no less favourable than those on which were immediately prior to the Effective Date in the Demerged Company.
- 20.2 The Resulting Company agrees that the services of all the employees of the Demerged Undertaking prior to the transfer, as aforesaid, shall be taken into account for the purposes of all benefits to which such employees may be eligible and accordingly, the period of service of such employees shall be reckoned therefore from the date of their respective appointment in the Demerged Company.
- 20.3 In the event of retrenchment of the employees of the Demerged Undertaking on and after the Demerger Appointed Date, the Resulting Company will be liable to pay retrenchment compensation in accordance with law on the basis that the services of the employees shall have been continuous and shall not have been interrupted by reason of such demerger.
- 20.4 Till the Effective Date of this Scheme, the Demerged Company shall make contributions to the government maintained provident fund and / or other funds in relation to the staff, workmen and employees of the Demerged Undertaking. On and from the Effective Date, the Resulting Company shall make appropriate contributions to such provident fund and / or other funds in respect of the staff, workmen and employees taken over by it pursuant to this Scheme. The contributions, and all accretions thereto, in the provident fund account, superannuation fund, gratuity fund and other benefit funds if any, of which the employees of the Demerged Undertaking are members or beneficiaries till

* HAPPINESS

the Effective Date, shall, with the approval of the concerned authorities and after giving effect to Part II and Part III of the Scheme, be transferred (in such proportion as is allocable to the employees of the Demerged Undertaking being transferred to the Resulting Company) to the relevant funds of the Resulting Company or government for the benefit of the employees of the Demerged Undertaking on terms no less favourable than immediately existing prior to the Effective Date. In the event that the Resulting Company has its own funds in respect of any of the funds referred to above, such investments shall, subject to the necessary approvals and permissions, be transferred to the relevant funds of the Resulting Company. [In the event that the Resulting Company does not have its own fund in respect of any of the aforesaid matters, the Resulting Company may, subject to necessary approvals and permissions, continue to contribute in respect of the employees engaged in the Demerged Undertaking to the relevant funds of the Demerged Company, until such time that the Resulting Company creates its own fund, at which time the investments and contributions pertaining to the employees of the Demerged Undertaking shall be transferred to the funds created by the Resulting Company.] In case, necessary approvals are not received by the Effective Date and there is delay, all such amounts shall continue to be administered by the Demerged Company in trust for the Resulting Company from the Effective Date till the date of actual transfer and, on receiving the approvals all the accumulated amounts till such date, shall be transferred to the respective funds of the Resulting Company suo moto.

- 20.5 Any disciplinary action initiated by the Demerged Company against any employee of the Demerged Undertaking shall have full force, effect and continuity as if it was initiated by the Resulting Company instead of the Demerged Company.
- 20.6 Any question that may arise as to whether any employee belongs or does not belong to the Demerged Undertaking shall be mutually decided by Board Demerged Company and the Resulting Company.

21. CONTRACTS, DEEDS, ETC.

- 21.1 Without limiting the generality of Clause 17.1, upon the Scheme becoming effective, and with effect from the Demerger Appointed Date and after giving effect to Part II of the Scheme, subject to the other provisions of this Scheme, all contracts, deeds, bonds, agreements, arrangements and other instruments (including all leases, licenses and other assurances in favour of the Demerged Company or powers or authorities granted by or to it) of whatsoever nature to which the Demerged Company is a party or to the benefit of which the Demerged Company may be eligible, all in relation to the Demerged Undertaking and which are subsisting or having effect immediately before the Effective Date, shall, without any further act, instrument or deed, continue in full force and effect in favour of, by, for or against the Resulting Company and may be enforced as fully and effectually as if, instead of the Demerged Company, the Resulting Company had been a party or beneficiary or obligee or obligor thereto or thereunder. It is hereby clarified that upon the Scheme becoming effective, and with effect from the Demerger Appointed Date and after giving effect to Part II and Part III of the Scheme, the Demerged Company shall have no rights and liabilities in respect of any of the aforesaid contracts / arrangements transferred to the Resulting Company for the period after the Demerger Appointed Date.
- 21.2 Without prejudice to the other provisions of this Scheme and notwithstanding the fact that the vesting of the Demerged Undertaking of the Demerged Company in the Resulting Company occurs by virtue of this Scheme itself, the Resulting Company may, at any time after the coming into effect of this Scheme in accordance with the provisions hereof, if so required, under any law or otherwise, take such actions or enter into, or issue or execute deeds, writings, confirmations, novations, declarations,



or other documents with, or in favour of, any party to any contract or arrangement to which the Demerged Company is a party or any writings as may be necessary to be executed in order to give formal effect to the provisions of this Scheme. The Resulting Company shall, under the provisions of this Scheme, be deemed to be authorised without requirement of any consent, approval of authority of the Demerged Company, whether in writing or verbal, to execute any such writings in place and substitution of the Demerged Company and to carry out or perform all such formalities or compliances required for the purposes referred to above.

- 21.3 Upon the coming into effect of the Scheme, all powers of attorney given, issued or executed by the Demerged Company, in relation to the Demerged Undertaking, in favour of any person shall cease to have effect without any further act, deed or instrument.
- 21.4 For avoidance of doubt and without prejudice to the generality of the foregoing, it is clarified that upon the coming into effect of this Scheme and with effect from the Demerger Appointed Date, all consents, permissions, certificates, authorities, powers of attorney given by, issued to or executed in favour of the Demerged Company in relation to the Demerged Undertaking shall stand transferred to the Resulting Company, as if the same were originally given by, issued to or executed in favour of the Resulting Company, and the Resulting Company shall be bound by the terms thereof, the obligations and duties thereunder and the rights and benefits under the same shall be available to the Resulting Company.
- 21.5 On and from the Effective Date, the Resulting Company shall, in its own right, be entitled to realise all monies and complete and enforce all pending contracts and transactions in respect of the Demerged Undertaking in the name of the Demerged Company but for the benefits and entitlement of the Resulting Company, in so far as may be necessary, until the transfer of rights and obligations of the Demerged Company to the Resulting Company under this Scheme is formally accepted by the parties concerned.
- 21.6 Without prejudice to the aforesaid, it is clarified that if any contracts, deeds, bonds, agreements, schemes, arrangements or other instruments of whatsoever nature in relation to the Demerged Undertaking which the Demerged Company owns or to which the Demerged Company is a party, cannot be transferred to the Resulting Company for any reason whatsoever, the Demerged Company shall hold such assets, contracts, deeds, bonds, agreements, schemes, arrangements or other instruments of whatsoever nature in trust for the benefit of the Resulting Company, in so far as it is permissible so to do, till such time as the transfer is given effect to.

22. LEGAL PROCEEDINGS

22.1 Without limiting the generality of Clause 17.1, upon the Scheme becoming effective, and with effect from the Demerger Appointed Date, all suits, claims, actions, appeals and legal proceedings of whatever nature pending and / or arising after the Demerger Appointed Date in relation to the Permitted Encumbrances and all suits, claims, actions, appeals and legal proceedings of whatever nature in relation to the Demerged Undertaking arising after the Demerger Appointed Date and pertaining to the period commencing on or after the Demerger Appointed Date, in any court or before any authority, judicial, quasi-judicial or administrative or any adjudicating authority or any arbitral tribunal, by or against the Demerged Company in relation to the Demerged Undertaking shall be continued and/ or enforced until the Effective Date as desired by the Resulting Company and on and from the Effective Date and after giving effect to Part II of the Scheme, the same shall not abate or be

* Herisage *

discontinued or in any way be prejudicially affected by reason of this demerger or by anything contained in this Scheme, but the said suits, appeals or other legal proceeding shall be continued and/or enforced by or against the Resulting Company as effectually and in the same manner and to the same extent as if the same had been originally instituted and/or had arisen and/or were pending by or against the Resulting Company.

- 22.2 On and from the Effective Date, and with effect from the Demerger Appointed Date and after giving effect to Part II of the Scheme, the Resulting Company shall have the right to initiate, defend, compromise or otherwise deal with any legal proceedings relating to the Demerged Undertaking, in the same manner and to the same extent as it would or might have been initiated by the Demerged Company, as the case may be, had the Scheme not been made.
- 22.3 On and from the Demerger Appointed Date and after giving effect to Part II of the Scheme, if any proceedings are taken by or against the Demerged Company in relation to the Demerged Undertaking pertaining to the period commencing on or after the Demerger Appointed Date, the Demerged Company shall till the Effective Date continue and/or defend the same at the cost of the Resulting Company, and the Resulting Company shall reimburse and indemnify the Demerged Company against all liabilities and obligations incurred by the Demerged Company in respect thereof.
- 22.4 The Resulting Company undertakes to have all legal or other proceedings initiated by or against the Demerged Company in relation to the Permitted Encumbrances and in relation to the Demerged Undertaking pertaining to the period commencing on or after the Demerger Appointed Date referred to in Clause 22.1 above transferred to its name on and after the Effective Date, and after giving effect to Part II of the Scheme, and to have the same continued, prosecuted and enforced by or against the Resulting Company as the case may be, to the exclusion of the Demerged Company.
- 22.5 Notwithstanding the above, in case the proceedings referred to in Clause 22.1 above cannot be transferred for any reason, or the transfer takes time, till such transfer the Demerged Company shall defend the same in accordance with the advice of the Resulting Company and at the cost of the Resulting Company, and the Resulting Company shall reimburse, indemnify and hold harmless the Demerged Company against all liabilities and obligations incurred by the Demerged Company in respect thereof.

TAXES

- 23.1 Without limiting the generality of Clause 17.1, upon the Scheme becoming effective, and with effect from the Demerger Appointed Date and after giving effect to Part II of the Scheme, all taxes, duties, cess of any nature (including income-tax, sales tax, excise duty, customs duty, service tax, VAT etc.) paid or payable, including any tax deduction or collection at source, service tax input credit receivables, by the Demerged Company in relation to the Demerged Undertaking and relating to the period after the Demerger Appointed Date until the Effective Date, shall be deemed to have been on account of or on behalf of or paid or payable by the Resulting Company and shall, in all proceedings, be dealt with accordingly.
- 23.2 On the Scheme becoming effective and after giving effect to Part II of the Scheme, the Demerged Company and the Resulting Company may revise their respective returns pertaining to income tax, service tax, sales tax, VAT and other tax returns, and claim refunds and/or credits including credits relating to tax deducted at source, as applicable pursuant to the provisions of this Scheme.



24. CONDUCT OF BUSINESS TILL EFFECTIVE DATE

- 24.1 With effect from the Demerger Appointed Date and till the Effective Date:
- 24.1.1 The Demerged Company shall carry on and shall be deemed to have carried on, all the business, activities and operations relating to the Demerged Undertaking and shall hold and stand possessed of and shall be deemed to have held and stood possessed of the assets, properties and liabilities of the Demerged Undertaking on account of, and/or on behalf of and/or for the benefit of, and/or in trust for, the Resulting Company.
- 24.1.2 All the profits or incomes accruing or arising to the Demerged Company and all expenditure or losses arising or incurred (including all taxes, if any, paid or accruing in respect of any profits and income) by the Demerged Company in relation to the Demerged Undertaking shall, for all purposes, be treated and be deemed to be and accrue as the profits or incomes or as the case may be, expenditure or losses (including taxes) of the Resulting Company.
- 24.1.3 Any of the rights, powers, authorities and privileges attached or related or pertaining to the Demerged Undertaking and exercised by or available to the Demerged Company, shall be deemed to have been exercised by the Demerged Company for and on behalf of and as an agent for the Resulting Company. Further, any of the obligations, duties and commitments attached, relating or pertaining to the Demerged Undertaking that have been undertaken or discharged by the Demerged Company shall be deemed to have been undertaken or discharged for and on behalf of and as an agent for the Resulting Company.

25. SAVING OF CONCLUDED TRANSACTIONS

Subject to the terms of this Scheme, the transfer and vesting of the Demerged Undertaking of the Demerged Company under this Scheme shall not affect any transactions or proceedings already concluded by the Demerged Company on or after the Demerger Appointed Date till the Effective Date, to the end and intent that the Resulting Company accepts and adopts all acts, deeds and things made, done and executed by the Demerged Company in relation to the Demerged Undertaking as acts, deeds and things made, done and executed by or on behalf of the Resulting Company.

26. ISSUE OF SHARES BY RESULTING COMPANY

26.1 Issue of Shares:

26.1.1 In consideration of the transfer and vesting of the Demerged Undertaking to and in the Resulting Company in terms of this Scheme, the Resulting Company shall, without any further application, act, instrument or deed and without any payment but subject to applicable law, after the Effective Date after giving effect to Part III of this Scheme, issue and allot to the members of the Demerged Company whose names appear on the Register of Members of the Demerged Company on the Record Date or to his / her / their respective heirs, executors, administrators or, as the case may be, successors, equity shares of the Resulting Company as under;

1,78,47,420 (One Crore seventy eight lakhs forty seven thousand four hundred and twenty) equity shares of the face value of INR 2/- (Rupees Two), each fully paid-up, of the Resulting Company to be

* HEALTH * HAPPINESS issued on a proportionate basis to members or his / her / their respective heirs, executors, administrators or, as the case may be, successors holding fully paid-up equity shares in the Demerged Company on the Record Date. The new equity shares to be issued by the Resulting Company under this Clause are in this Scheme referred to as the "New Equity Shares".

- 26.1.2 The New Equity Shares to be issued and allotted by the Resulting Company to equity shareholders of the Demerged Company under Clause 26.1.1 above shall be subject to adjustments to take into account any corporate actions including but not limited to issuances of bonus shares, stock splits, and stock consolidation but excluding any dividend announced or to be announced on the shares of the Resulting Company prior to the Effective Date.
- 26.1.3 No fractional shares shall be issued by the Resulting Company. Fractional entitlements, if any, arising shall be rounded off to the nearest integer. A fraction of less than half shall be rounded down to the nearest lower integer and a fraction of half or more shall be rounded up to the nearest higher integer.
- 26.1.4 Upon the Scheme becoming effective, the issued, subscribed and paid-up share capital of the Resulting Company shall stand suitably increased consequent upon the issuance of the New Equity Shares in accordance with Clause 26.1.1. The approval of this Scheme by the shareholders of the Resulting Company and the Demerged Company, under Sections 391 to 394 of the Companies Act, 1956 and other applicable provisions of the Act, shall also be deemed to be the approval by the shareholders under the provisions of Section 62 of the Companies Act, 2013 and all other applicable provisions of the Act and applicable law for the purpose of subscription and issuance and allotment of the New Equity Shares in accordance with the Scheme. It is clarified that no additional special resolution under Section 62(1)(c) of the Companies Act, 2013 or any other provision of the Act or applicable law shall be required to be passed for issuance and allotment of the New Equity Shares under this Scheme.

26.2 Issue in Dematerialized Form:

- 26.2.1 All New Equity Shares to be issued and allotted under Clause 26.1.1 by the Resulting Company shall be issued in dematerialized form.
- 26.2.2 If the requisite details of the account of any shareholder with a depository participant are not recorded with the Demerged Company, such shareholder concerned will be required to provide the said details to enable the Resulting Company to allot the New Equity Shares in dematerialized form to the concerned shareholder.

26.3 New Equity Shares to rank pari passu:

- 26.3.1 The New Equity Shares issued and allotted in terms of this Scheme shall rank pari passu in all respects with the existing equity shares of the Resulting Company including in respect of dividends, if any, that may be declared by the Resulting Company on or after the Effective Date.
- 26.3.2 It is clarified that the aforesaid Clause 26.3.1 in respect of declaration of dividends is an enabling provision only and shall not be deemed to confer any right on any member of the Resulting Company and the Demerged Company to demand or claim any dividends which, subject to the provisions of the Act, shall be entirely at the discretion of the Board of Directors of the Resulting Company and the Demerged Company and subject to the approval of the shareholders of the Resulting Company and

* HEALTH * Hartage * the Demerged Company.

26.4 Listing:

- 26.4.1 The New Equity Shares issued by the Resulting Company will be listed and/or admitted to trading on the Stock Exchanges where the shares of the Resulting Company are listed and/or admitted to trading and all necessary applications will be made in this aspect by the Resulting Company.
- 26.4.2 The New Equity Shares allotted by the Resulting Company pursuant to the Scheme, shall remain frozen in dematerialized form for listing and trading on respective Stock Exchanges pending permissions for the same from the respective Stock Exchange.

26.5 Resulting Company to obtain necessary approvals:

The Resulting Company shall, if and to the extent required, apply for and obtain the required statutory approvals of the concerned Governmental Authority for the issue and allotment of the New Equity Shares.

27. REMAINING UNDERTAKING OF THE DEMERGED COMPANY

- 27.1 The Remaining Undertaking of the Demerged Company including all the properties and assets, investments in all subsidiaries, debts, liabilities and obligations of the Demerged Company, relating to the Remaining Undertaking of the Demerged Company and which do not form part of the Demerged Undertaking shall continue to belong to and remain vested in the Demerged Company.
- 27.2 The Demerged Company shall be entitled to carry on its business and activities pertaining to the Remaining Undertaking of the Demerged Company in such manner as it may deem fit and proper and nothing herein contained shall affect the business and activities of the Demerged Company in relation to the Remaining Undertaking of the Demerged Company. Further, the Demerged Company shall be deemed to have been carrying on all business and activities relating to the Remaining Undertaking of the Demerged Company for and on its own behalf.
- 27.3 All assets and properties acquired by the Demerged Company at any time including on and after the start of business on the Demerger Appointed Date, shall, to the extent that the same do not relate to the Demerged Undertaking, form part of the Remaining Undertaking of the Demerged Company.
- 27.4 All liabilities, debts and obligations incurred by or arising against the Demerged Company at any time including on and after the start of business on the Demerger Appointed Date, shall, to the extent that the same do not relate to the Demerged Undertaking, form part of the Remaining Undertaking of the Demerged Company.
- 27.5 The Demerged Company shall be entitled to enter into such contracts as the Demerged Company may deem fit and proper in respect of the Remaining Undertaking of the Demerged Company.
- 27.6 All the profits or incomes accruing or arising and all expenditure or losses arising or incurred (including all taxes, if any, paid or accruing in respect of any profits and income) by the Demerged Company in relation to the Remaining Undertaking of the Demerged Company shall, for all purposes, be treated and be deemed to be and accrue as the profits or incomes, or as the case may be,

* Appiness

expenditure or losses (including taxes) of, the Demerged Company.

48

27.7 All the legal and other proceedings by or against the Demerged Company under any statute, whether pending on the Demerger Appointed Date or which may be instituted after the Demerger Appointed Date, whether or not relating to the Demerged Undertaking of the Demerged Company, shall be continued and enforced by or against the Demerged Company.

28. ACCOUNTING TREATMENT IN THE BOOKS OF THE RESULTING COMPANY

28.1 The Resulting Company shall account for the demerger in its books of account as per the applicable accounting principles prescribed under Indian accounting standards (IND AS) prescribed under the Act.

29. ACCOUNTING TREATMENT IN THE BOOKS OF THE DEMERGED COMPANY

- 29.1 Pursuant to the Demerger, the Demerged Company shall transfer and reduce its assets and liabilities by the values of assets and liabilities (including any directly attributable transaction costs) relating to the Demerged Undertaking appearing in the books of accounts of the Demerged Company, immediately before the Demerger, with the resulting adjustment to be recorded in the capital reserve account.
- 30. ACCOUNTING TREATMENT IN THE BOOKS OF THE SHAREHOLDERS OF DEMERGED COMPANY
- 30.1 The adjustment required in the book value of investment in the Demerged Company for reduction by the value of the net assets transferred by Demerged Company to Resulting Company, shall be recorded as cost of the New Equity Shares issued by the Resulting Company.

PART IV

REDUCTION OF CAPITAL THROUGH CANCELLATION OF SHARES OF DEMERGED COMPANY HELD BY ITS EXISTING SHAREHOLDERS AND UTILISATION OF SECURITIES PREMIUM ACCOUNT

31. REDUCTION OF SHARE CAPITAL OF DEMERGED COMPANY

- 31.1 On and from the Effective Date, and with effect from the Demerger Appointed Date and after giving effect to Part II and Part III of the Scheme, the issued, subscribed and paid-up equity share capital of the Demerged Company shall, without any further application, act, instrument or deed and without any payment, be reduced.
- 31.2 The above reduction of equity share capital shall be carried out by reducing the number of shares held by existing shareholders of the demerged company on a proportionate basis. Such cancellation of the share capital of the Demerged Company shall result in a mirror image of the pre-demerger shareholding pattern of the Demerged Company.
- 31.3 On and from the Effective Date and with effect from the Demerger Appointed Date, the balance in

* APPINESS

the Securities Premium Account of the Demerged Company shall be adjusted against the debit balance of Capital Reserve Account of the Demerged Company. The debit balance, if any, in the Capital Reserve Account of the Demerged Company, post the adjustment of the Securities Premium Account, shall be adjusted against the face value of the equity share capital as per Clauses 31.1 and 31.2.

- 31.4 Such reduction of Equity Share Capital and Securities Premium Account of the Demerged Company as provided in Clause 31.1 above, shall be effected as an integral part of the Scheme on the Effective Date and the order of the High Courts sanctioning the Scheme shall be deemed to be an order under Section 102 of the Companies Act, 1956, confirming the reduction in share capital of the Demerged Company, and no separate sanction under Sections 100 to 103 of the Companies Act, 1956 and Section 52 of the Companies Act, 2013 will be necessary.
- 31.5 The reduction would not involve a diminution of liability in respect of the unpaid share capital or payment of paid-up share capital, and the provisions of Section 101 of the Companies Act, 1956 will not be applicable.
- 31.6 Notwithstanding the reduction of the issued, subscribed and paid-up equity share capital of the Demerged Company, it shall not be required to add the words "And Reduced" as suffix to its name.

PART V

GENERAL TERMS AND CONDITIONS

32 APPROVALS

- 32.1 The Transferee Company shall be entitled, pending the sanction of the Scheme, to apply to any Governmental Authority and all agencies, departments and authorities concerned as are necessary under any law for such consents, approvals and sanctions which the Transferee Company may require to own and operate the Retail Undertaking and the VetCa Undertaking to be transferred to them under this Scheme.
- 32.2 The Resulting Company shall be entitled, pending the sanction of the Scheme, to apply to any Governmental Authority and all agencies, departments and authorities concerned as are necessary under any law for such consents, approvals and sanctions which the Resulting Company may require to own and operate the Demerged Undertaking to be transferred to it under this Scheme.

33 ADMINISTRATIVE CONVENIENCE

- 33.1 Notwithstanding anything contained in other clauses of this Scheme, the Transferor Company, the Demerged Company and the Resulting Company, shall enter into such documents, agreements, make applications to various authorities, regulatory bodies to facilitate the uninterrupted transitions of the business from the Transferor Company to the Transferee Company, and from the Demerged Company to the Resulting Company.
- 33.2 Notwithstanding anything contained in other clauses of this Scheme but in accordance with the Act and other applicable laws, the Transferor Company, the Demerged Company and the Resulting Company, may enter into such documents, agreements, arrangements and make applications to various authorities, regulatory bodies to facilitate the sharing of, inter alia any common services,

HEALTH HAPPINESS

employees, intellectual properties and other assets (whether moveable or immoveable).

34 MODIFICATION OF SCHEME

- Each of the Transferor Company, the Demerged Company and the Resulting Company by their respective Boards of Directors or any committee thereof or any Director authorised in that behalf (hereinafter referred to as the "Delegate") may together assent to, or make, from time to time, any modifications or amendments or additions to this Scheme which the High Courts or any Government Authority may deem fit to approve of or impose and which the Companies may in their discretion accept, or such modifications or amendments or additions as the Companies or as the case may be, their respective Delegate may deem fit, or required for the purpose of resolving any doubts or difficulties that may arise in carrying out the purpose of this Scheme and as approved by the High Courts, and the Companies by their respective Boards of Directors or Delegates are authorised to do and execute all acts, deeds, matters and things necessary for bringing this Scheme into effect, or review the position relating to the satisfaction of the conditions of this Scheme and if necessary, waive any of such conditions (to the extent permissible under law) for bringing this Scheme into effect. In the event that any conditions may be imposed by the High Courts or any authorities, which the Companies find unacceptable for any reason, then the Companies shall be at liberty to withdraw the Scheme. The aforesaid powers of the Companies may be exercised by the Delegate of the respective Companies. It is clarified that any modification or amendment to the Scheme by the Companies, after the sanction by the High Courts, shall only be made with the prior consent of the High Courts.
- 34.2 For the purpose of giving effect to this Scheme or to any modifications or amendments thereof or additions thereto, the Delegates (acting jointly) of the Companies may give and are authorised to determine and give all such directions as are necessary including directions for settling or removing any question of doubt or difficulty that may arise and such determination or directions, as the case may be, shall be binding on all parties, in the same manner as if the same were specifically incorporated in this Scheme.

35 FILING OF APPLICATIONS

35.1 Each of the Companies shall with all reasonable dispatch, make and file all applications/petitions under Sections 391 and 394 and other applicable provisions of the Act before the respective High Court for sanction of this Scheme and each of the Companies shall obtain all requisite approvals as may be required under law to give effect to the Scheme.

36 CONDITIONALITY OF SCHEME

- 36.1 This Scheme is conditional upon and subject to:
- 36.1.1 The Scheme being approved by the requisite majority of the members and/or creditors of the Companies and/or by such other persons as may be required under the Act and as directed by the High Courts;
- 36.1.2 The requisite sanctions and approvals of any Governmental Authority including Stock Exchanges, the Securities and Exchange Board of India, and the Competition Commission of India, as may be required by law, in respect of the Scheme being obtained;

HAPPINESS

- 36.1.3 The sanction of this Scheme by the High Courts;
- 36.1.4 Copies of the orders of the High Courts sanctioning the Scheme being filed with the Registrar of Companies, Maharashtra and the Registrar of Companies, Andhra Pradesh and Telangana;
- 36.1.5 The Scheme being approved by the Stock Exchanges in terms of Regulations 37 and 94 of the SEBI LODR Regulations and the SEBI Circular.

37 EFFECT OF NON-RECEIPT OF APPROVALS/ SANCTIONS

- 37.1 In the event of the Scheme not being sanctioned by the High Courts and/or the order or orders not being passed by December 31, 2017, or by such later date as may be agreed by the respective Boards of Directors of the Companies, the Scheme shall become fully null and void and in that event no rights and liabilities shall accrue to or be inter-se by the parties in terms of the Scheme, save and except in respect of any act or deed done prior thereto as is contemplated hereunder or as to any rights and/ or liabilities which might have arisen or accrued pursuant thereto and which shall be governed and be preserved or worked out as is specifically provided in the Scheme or as may otherwise arise in law. In such event, each party shall bear and pay its respective costs, charges and expenses for and/ or in connection with the Scheme.
- 38 CHANGE IN OBJECTS CLAUSE OF FRL
- 38.1 With effect from the Demerger Appointed Date, and upon the Scheme becoming effective, the main object clause of the Memorandum of Association of the Resulting Company shall be altered and amended, without any further act or deed, to include the objects as required for the purpose of carrying on the business activities of Demerged Undertaking, pursuant to the provisions of Sections 13 and 14 of the Companies Act, 2013 and other applicable provisions of the Act. The following clause shall be added to the Memorandum of Association of the Resulting Company in addition to its main object clause:
 - "i. To manufacture, process, prepare, preserve, refine, bottle, buy, sell and deal whether as wholesaler or retailers or as exporters or importers or as Principals or agents or as keepers or dealers in all kinds of milk products, including Cheese, Butter, Ghee, Ice creams, Baby foods, Instant foods and any by-products or co-products thereof and to carry on the business and setting up of dairy farms, milk processing plants, food processing plants, cold storage plants, research laboratories, packing units, bottling plants and to manufacture and deal in all kinds and varieties of foods for human or animal consumption.
 - II. To carry on the business of manufacturers, millers, grinders, rollers, processors, tankers, packers and preserves, and dealers of all foods from agriculture products, dairy products, horticulture and poultry products, fruits, vegetables, flowers, meats, processed meat scanned and tinned and processed foods, fast foods, processed fish and sea foods, frozen foods, protential foods, health and instant foods of all kinds, including baby and dietic foods, cereals, beverages, restoratives and aerated mineral waters and food stuffs and consumable provisions and to extract by-products, derivatives food preparations of every kind and description."
- 38.2 For the purposes of amendment in the Memorandum of Association of the Resulting Company as provided in this clause, the consent / approval given by the shareholders of the Resulting Company to

* Hechage *

this Scheme pursuant to Section 391 of the Companies Act, 1956 and any other applicable provisions of the Act shall be deemed to be sufficient and no further resolution of members of FRL as required under the provisions of Section 13 and 14 of the Companies Act, 2013 and any other applicable provisions of the Act shall be required to be passed for making such change / amendment in the Memorandum of Association of the Resulting Company.

39 SEVERABILITY

39.1 Each Section of the Scheme shall be given effect to as per the chronology in which it has been provided for in the Scheme. Each part in each Section is independent of each Section and is severable. However, failure of any one part of one Section for lack of necessary approval from the shareholders / creditors / statutory regulatory authorities or for any other reason that the Board of Directors may deem fit than this shall not result in the whole Scheme failing. It shall be open to the Board of Directors concerned to consent to sever such part(s) of the Scheme and implement the rest of the Scheme with such modification.

40 COSTS, CHARGES AND EXPENSES

40.1 All costs, charges, and all expenses of the Transferor Company, the Demerged Company and the Resulting Company arising out of, or incurred in carrying out and implementing this Scheme and matters incidental thereto, shall be borne and paid by the Transferor Company and the Resulting Company as mutually agreed between them.

SCHEDULE I

LIST OF PERMITTED ENCUMBRANCES

Serial No.	Case Reference	Name of counterparty	Immovable property involved	Land in acres/ Amount in Rupees involved
1.	O.S. 11 of 2016	K. Pochaya	S.No. 127 Advi Majeed Village, Mulugu Mandal, Medak District	1.17 acres
2.	O.S. No. 225/2009	B. Chandra	S.No. 139/2 Chittor District, Kuppam Sub Reg. District Santhipuram Revenue Mandal Mattam Grampanchayat, No. 28 Gundisettipalli Village	0.1 acres
3.	I.A. No. 1130 / 2009	B. Chandra	S.No. 139/2 Chittor District, Kuppam Sub Reg. District Santhipuram Revenue Mandal Mattam Grampanchayat, No. 28 Gundisettipalli Village	0.1 acres
4.	I.A. No. 1161 / 2009	B. Chandra	S.No. 139/2 Chittor District, Kuppam Sub Reg. District Santhipuram Revenue Mandal Mattam Grampanchayat, No. 28 Gundisettipalli Village	0.1 acres
5.	I.A. No. 1081 /	B. Chandra	S.No. 139/2	0.1 acres

* HEALTH * Heritage *

	2009		Chittor District, Kuppam Sub Reg, District Santhipuram Revenue Mandal Mattam Grampanchayat, No. 28 Gundisettipalli Village	
6.	Caveat	B. Chandra	S.No. 139/2 Chittor District, Kuppam Sub Reg, District Santhipuram Revenue Mandal Mattam Grampanchayat, No. 28 Gundisettipalli Village	0.1 acres
7.	Order in ROC.A/26/2010 Dt, /09/2010 issued by Tahasildar, Shantipuram Mandal	Revenue Divisional Office, Madanapalli	S.No. 139/2 Chittor District, Kuppam Sub Reg. District Santhipuram Revenue Mandal Mattam Grampanchayat, No. 28 Gundisettipalli Village	INR 3,11,092
8.		Revenue Divisional Office, Madanapalli	S.Nos: 134/3, 138/3, 136/3, 138/1B. Chittor District, Kuppam Sub Reg. District Santhipuram Revenue Mandal Mattam Grampanchayat, No. 28 Gundisettipalli Village	6.86 acres

9. Financial Lease:

HFL has acquired Servers and related accessories from CISCO SYSTEMS (INDIA) PVT LTD on a financial lease starting June 10, 2015 for a period of 60 months ending March 10, 2020. Installments are paid on a quarterly basis for which post-dated cheques have been issued to CISCO SYSTEMS (India) PVT LTD. The Principal outstanding as on October 31, 2016 is INR 1,31,48,034 (One crore Thirty one lakhs Forty eight thousand and Thirty four Rupees).



Valuation Certificate ('Certificate') of Retail Undertaking and VetCa Undertaking of Heritage Foods Limited ('HFL')

Raju & Prasad Chartered Accountants

401, Diamond House, Adj. Amrutha Hills, Punjagutta, Hyderabad

Phone No: 23410404/405/406, e-mail: hyderabad@rajuandprasad.com

NEW DELHI |MUMBAI |BANGALORE |JALGAON |NAVI MUMBAI

RAJU & PRASAD CHARTERED ACCOUNTANTS

Head Office:

401, "Diamond House", Adj. Amrutha Hills,

Punjagutta, Hyderabad - 500 082 Ph 040 - 23410404/05/06

Fax : 040 - 23410403

E-mail: hyderabad@rajuandprasad.com rajuandprasad@gmail.com

Date: 07.11.2016

To, The Board of Directors, Heritage Foods Limited, Hyderabad.

Dear Sirs,

Sub: Valuation Certificate of Retail Undertaking and VetCa Undertaking of Heritage Foods Limited as on 31.10.2016.

In connection with the proposed business reorganisation of Heritage Foods Limited by way of slump sale of the Retail Undertaking and the VetCa Undertaking its wholly owned subsidiaries to Heritage Foods Retail limited, we, M/s Raju & Prasad Chartered Accountants have been requested to issue valuation certificate of Retail and VetCa Undertakings.

Based on the discussions with the management and procedures carried out by us on the information received from the management we certify that the value of Retail and VetCa Undertakings of Heritage Foods limited as per Net Asset Method (using book values)as at 31.10.2016 (Closing Hours) is Rs. 134.49 Crores.

Please feel free to contact us for any further clarification/information.

We thank the management and employees of the company for the cooperation extended.

Thanking You, Yours sincerely,

For Raju & Prasad Chartered Accountants

1- Bala Kelu le

Y. Bala Krishna Reddy

Partner

M. No: 223701.

Ph: +91-9241570047

RAJU & PRASAD

CHARTERED ACCOUNTANTS

401, Diamond House, Adj. Amrutha Hills, Punjagutta, Hyderabad-082 Phone No: 23410403, 04 hyderabad@rajuandprasad.com

1. Introduction

Heritage Foods Limited (Formerly Known as Heritage Foods (India) Limited) (hereinafter referred to as "Transferor Company" or "HFL") has six business verticals (i.e. Dairy, Retail, Agri, Bakery, Renewable Energy and VetCa). The management of HFL has proposed to reorganise the business of the company by way of slump sale of Retail Undertaking (comprising of Retail, Agri and Bakery Business verticals) and VetCa Undertaking to Heritage Foods Retail Limited (hereinafter Referred to as "Transferee Company" or "HFRL"). In connection with this proposed reorganisation of business, the management of the company has requested, M/s Raju & Prasad Chartered Accountants for issuing Valuation Certificate of Retail Undertaking and VetCa Undertaking as at 31.10.2016 (Closing hours).

2. Heritage Foods Limited ("Transferor Company" or "HFL")

HFL is a public limited company incorporated under the Companies Act, 1956 (CIN: L15209TG1992PLC014332) and having its registered office at #6-3-541/ C, Punjagutta, Hyderabad - 500082. The equity shares of HFL are listed on the BSE Limited (Stock Code: 519552) and the National Stock Exchange (Stock Code: HERITGFOOD). The paid-up equity share capital of HFL is 2,31,99,000 equity shares of Rs. 10 each. HFL has 6 (six) business verticals:

- Dairy Business Vertical HFL produces, outsources and markets a complete range of dairy products including fresh milk, curd, buttermilk, ice creams and other value added products across various states in India.
- Retail Business Vertical HFL is engaged in the grocery and food retail business, undertaken from its dedicated retail stores.
- c) Agri Business Vertical HFL is engaged in the business of sourcing, processing and marketing fresh fruits and vegetables. This business vertical acts as the supply chain arm for the Retail Undertaking vertical while also supplying the products to other retail chains and stores.
- Bakery Business Vertical HFL is engaged in the business of manufacturing and supplying bakery products to other customers, besides HFL retail outlets.
- e) Veterinary Care Business Vertical (VetCa) HFL supplies cattle feed to dairy farmers and general traders and maize to poultries and distilleries.
- Renewable Energy Business Vertical HFL is engaged in the production of solar energy (2.34 MW) and wind energy (4.2 MW) for captive consumption.

The Retail Undertaking comprises the retail business vertical, the agri business vertical, and the bakery business vertical. The VetCa Undertaking comprises the veterinary care business vertical.

3. Heritage Foods Retail Limited ("Transferee Company" or "HFRL")

HFRL is a public limited company incorporated under the Companies Act, 1956 (CIN:U15400TG2008PLC062054) and having its registered office at #6-3-541 / C, Punjagutta, Hyderabad – 500082 for undertaking the activities of trading and dealing in goods and produce, and processing, packaging and selling agri products. Paid up share capital of HFRL is 1,65,600 equity shares of Rs. 10/- each. HFRL is a wholly owned subsidiary of HFL. HFRL has not commenced its commercial operations.

4. Proposed reorganisation of Business:

We understand that the management of HFL intends to reorganise company's business by way of slump sale, as defined under the provisions of section 2(42C) of the Income-tax Act, of the Retail Undertaking and the VetCa Undertaking to its wholly owned subsidiary, Heritage Foods Retail limited for a consideration to be received in the form of equity shares of HFRL. The proposed reorganisation of business is going to be carried out through a scheme of arrangement

RAJU & PRASAD

CHARTERED ACCOUNTANTS

401, Diamond House, Adj. Amrutha Hills, Punjagutta, Hyderabad-082 Phone No: 23410403, 04 hyderabad@rajuandprasad.com

under Section 391 to 394 read with Section 100 to 103 of Companies Act, 1956 and Section 52 of the Companies Act, 2013 and other provisions of Companies Act, 2013, and Companies Act, 1956, as applicable.

According to the explanations provided to us and as per Part II of Composite Scheme of Arrangement, the reorganisation of business provide greater business attention and focus on the dairy and renewable energy business verticals which have high growth potential, which may result in increasing the profitability while simultaneously attracting strategic partners and lenders for the retail, agri and bakery business verticals of HFL and creating long term value for the various stakeholders. In addition, the veterinary care business vertical will be restructured into wholly owned subsidiary to unlock value. Accordingly, the management of HFL and HFRL are of the opinion that the Retail Undertaking (comprising the retail business vertical and the bakery business vertical) and the VetCa Undertaking (comprising the veterinary care business vertical) should be transferred on a going concern to a wholly owned subsidiary (HFRL) of HFL.

5. Scope of the assignment

Scope of the present assignment is to certify the value of the Retail and VetCa Undertakings as per Net Asset Method (Using Book Values) as at 31.10.2016 (Closing Hours).

6. Procedures

In this connection, we would like to state that we had relied on the information and explanations given to us by the management of the company. The procedures used in our analysis included such substantive steps as we considered necessary under the circumstances, including, but not necessarily limited to the following:

- 6.1 Considered the audited financial statements of HFL for the year ended 31st March 2016.
- 6.2 Considered Unaudited Financial Results of the company for the quarter ended 30th September, 2016.
- 6.3 Considered and relied on carved out financial statements of Retail and VetCa undertakings as at 31stOctober, 2016 as prepared and given by the management.
- 6.4 Interviews, correspondence with the management, explanations and representations given by the management, on which we have relied.
- 6.5 Such other analysis, reviews and inquiries, as we considered necessary.

7. Value of the Retail and VetCa Undertakings

Value Conclusion: On the basis of the procedures carried out by us on the information received and as per Net Asset Method (Using Book Values) we certify that the value of Retail and Vetca Undertakings as at 31.10.2016 (Closing Hours) is Rs.134.49 Crores.

8. Limitations and Disclaimer

- 8.1 We have relied upon the information, data and explanations detailed in paragraphs above for certifying value of the Retail Undertakings and VetCa Undertakings in connection with the proposed reorganisation.
- 8.2 We have not considered the value of brands, trademarks, logos, Intellectual Property Rights (IPR's), copy rights or any assets of similar nature as their values are not captured in the books of accounts.

RAJU & PRASAD

CHARTERED ACCOUNTANTS

401, Diamond House, Adj. Amrutha Hills, Punjagutta, Hyderabad-082 Phone No: 23410403, 04 hyderabad@rajuandprasad.com

- 8.3 Provisions relating to employee benefits namely Gratuity and Leave Encashment have been made on an adhoc basis.
- 8.4 Contingent liabilities in the nature of tax disputes and export obligation related to Retail Undertaking were not reduced from the value of the undertakings and value has been arrived based only on the liabilities which are already accounted.
- 8.5 For arriving at the value, we have used financial and other information provided by the Management, which we believe to be reliable; our conclusions are dependent on such information being complete and accurate in all material respects. Our scope of work does not enable us to accept responsibility for the accuracy and completeness of the financial and other information provided to us by the Management. We have, therefore, not carried out any due diligence, review, physical verification of assets, independent audit or other test or validation of such financial and other information to establish the accuracy or sufficiency of the financial statements referred to above or of the information, explanations and representations provided to us. We have thus relied upon the unaudited financials of HFL provided to us as on 31.10.2016. Accordingly, we do not express any opinion or any other form of assurance thereon and accept no responsibility for the same.
- 8.6 The Retail and VetCa Undertakings of HFL is proposed to be transferred to HFRL with effect from the Appointed Date (i.e. 1st Nov, 2016) and we have considered the financial statements of HFL and carved out financial statements of the Retail and VetCa Undertaking as at 31stOctober, 2016. The Management has explained that the Business would be carried on in due course of business from the Appointed Date and subsequently, till the Scheme is approved. The Management has represented that financial statements of HFL and the carved out financial statement of the Retail and VetCa Undertakings as at 31st October, 2016, provided to us, include all disclosures necessary for a fair presentation of its financial position and results of operations in accordance with generally accepted accounting principles in India consistently applied, and disclosures otherwise required by the laws and regulations to which they are subject.
- 8.7 Our scope of work is limited to certifying the value of the Retail and VetCa Undertakings as per Net Asset Method (Using Book Values). Our report is not, nor should it be construed as, our opining or certifying the compliance of the proposed scheme with the provisions of any law including Companies Act, FEMA and taxation related laws etc., or as regards any legal implications or issues arising from such proposed demerger.
- 8.8 The fee for this engagement is not contingent upon the results of the Valuation Certificate.
- 8.9 We have not considered valuation methodologies other than 'Net Assets Method (Using book values)' in arriving at the value of Retail and VetCa Undertakings.
- 8.10 We owe responsibility to only the Board of Directors of the Companies which has appointed us, and nobody else. We do not accept any liability to any third party in relation to the issue of this Report. This Report is not a substitute for the third party's own due diligence/appraisal/enquiries/independent advice that the third party should undertake for his purpose. It is understood that this analysis does not represent a fairness opinion. The report is confidential and has been prepared exclusively for the Client namely, Heritage Foods Limited. It should not be used or relied upon by, reproduced or circulated to any other person or for any purpose other than as mentioned above, in whole or in part, without our prior written consent. Such consent will only be given after full consideration of the circumstances at the time.

14th Floor, The Ruby 29 Senapati Bapat Marg Dadar (West) Mumbai-400 028, India

Tel: +91 22 6192 0000 Fax: +91 22 6192 1000

Dated: 7 November 2016

To

The Board of Directors

Heritage Foods Limited #6-3-541/C, Panjagutta, Hyderabad - 500 082, Telangana

The Board of Directors

Future Retail Limited Knowledge House, Shyam Nagar, Off Jogeshwari-Vikhroli Link Road, Jogeshwari (East), Mumbai – 400 060

Sub: Recommendation of Share Entitlement for the proposed demerger of identified divisions of Heritage Foods Limited into Future Retail Limited after the proposed spin-off of these divisions into a wholly owned subsidiary of Heritage Foods Limited

Dear Sir / Madam,

We refer to the engagement letter whereby Heritage Foods Limited (hereinafter referred to as "HFL") and Future Retail Limited (hereinafter referred to as "FRL") have requested S. R. Batliboi & Co. LLP (hereinafter referred to as "SRBC") for recommendation of number of equity shares to be issued by FRL as consideration for the proposed demerger of identified divisions of HFL into FRL after the proposed spin-off of these divisions into a wholly owned subsidiary of HFL ("Share Entitlement").

HFL and FRL are hereinafter jointly referred to as the "Companies". The Share Entitlement for this report refers to number of equity shares of face value of INR 2/each of FRL, which would be issued to HFL in lieu of their proposed demerger of identified divisions into FRL.

SCOPE AND PURPOSE OF THIS REPORT

Future Retail Limited operates retail stores across India under the Big Bazaar, easyday, Foodhall, Hometown and ezone brands. FRL is listed on the Bombay Stock Exchange and the National Stock Exchange. For the year ended 31 March 2016, FRL reported operating revenues of INR 68,450 mn and a profit after tax of INR 145 mn.

Heritage Foods Limited is engaged in the dairy, retail, agri-products, veterinary care, bakery and renewable energy businesses in India. It is listed on the Bombay Stock Exchange and the National Stock Exchange. For the year ended 31 March 2016, HFL reported operating revenues of INR 23,806 mn and a profit after tax of INR 554 mn.

S.R. Batikov & Co., LLP, a Limited Liability Fortnerskip with LLP Identity No. AAB-4294 Royd, Office : 22, Carnot Street, Block 'C', 2nd Floor, Kalkata-700 016

Chartered Accountants

Recommendation of Share Entitlement for the proposed demerger of identified divisions of HFL into FRL after the proposed spin-off of these divisions into a wholly owned subsidiary of HFL

HFL is proposing to spin-off its retail, bakery, agri and veterinary operations into a wholly owned subsidiary. Subsequently, HFL is proposing to demerge the retail, bakery and agri businesses (together referred to as "Divisions") into FRL for a consideration in the form of shares of FRL. We understand that this process would take place under the provisions of the Companies Act, 1956.

For the aforesaid purpose, the Board of Directors of HFL and FRL have appointed SRBC to submit a Share Entitlement Report for recommending the Share Entitlement, for the issue of FRL's equity shares to HFL, to be placed before the Audit Committee/ Board of Directors of the Companies.

We have been provided with historical unaudited financials of the Divisions of HFL, which is carved out from the audited / unaudited financials of HFL, upto 30 June 2016. We have taken into consideration the current market parameters in our analysis and have made adjustments for additional facts made known to us till the date of our Report.

This Report is our deliverable for the above engagement.

This Report is subject to the scope, assumptions, exclusions, limitations and disclaimers detailed hereinafter. As such, the Report is to be read in totality, and not in parts, in conjunction with the relevant documents referred to therein.

SOURCES OF INFORMATION

MUMBAI

In connection with this exercise, we have used the following information as received from the Companies:

 Unaudited statement of profit and loss and statement of assets and liabilities of Divisions for FY13, FY14, FY15 and FY16 which are carved out from the financials of HFL

 Unaudited statement of profit and loss and statement of assets and liabilities of Divisions of HFL for three months ended 30 June 2016 which are carved out from the unaudited financials of HFL

 Forecast financial statements and underlying assumptions for the Divisions from 1 July 2016 to 31 March 2021;

 Other information and documents for the purpose of recommendation of the Share Entitlement.

Apart from the above, publicly available information and proprietary data bases subscribed to by us were utilized for analyzing the industry.

During the discussions with the Management of both Companies, we have also obtained explanations and information considered reasonably necessary for our

Chartered Accountants

MUMBAI

Recommendation of Share Entitlement for the proposed demerger of identified divisions of HFL into FRL after the proposed spin-off of these divisions into a wholly owned subsidiary of HFL

exercise from the Companies we have valued respectively. The Companies have been provided with the opportunity to review the draft Report (excluding the recommendation for Share Entitlement) as part of our standard practice to ensure that factual inaccuracy / omissions are avoided in our final Report.

SCOPE LIMITATIONS, ASSUMPTIONS, QUALIFICATIONS, EXCLUSIONS AND DISCLAIMERS

Provision of valuation opinions and consideration of the issues described herein are areas of our regular practice. The services do not represent accounting, assurance, accounting / tax due diligence, consulting or tax related services that may otherwise be provided by us or our affiliates.

This Report, its contents and the results herein are specific to (i) the purpose of valuation agreed as per the terms of our engagement; (ii) the date of this Report and (iii) are based on the unaudited balance sheet of the Divisions as at 30 June 2016.

The Management of respective Companies have represented to us that the business activities of HFL and FRL including their subsidiaries and associates, as applicable, have been carried out in the normal and ordinary course between 30 June 2016 and the Report date and that no material adverse change has occurred in their respective operations and financial position between 30 June 2016 and the Report date.

A valuation of this nature is necessarily based on the prevailing stock market, financial, economic and other conditions in general and industry trends in particular as in effect on, and the information made available to us as of, the date hereof. Events occurring after the date hereof may affect this Report and the assumptions used in preparing it, and we do not assume any obligation to update, revise or reaffirm this Report.

In the ultimate analysis, valuation will have to be tempered by the exercise of judicious discretion by the Valuer and judgment taking into accounts all the relevant factors. There will always be several factors, e.g. quality and integrity of the management, present and prospective competition, yield on comparable securities and market sentiment, etc. which are not evident from the face of the balance sheets but which will strongly influence the worth of a share. This concept is also recognised in judicial decisions.

The recommendation(s) rendered in this Report only represent our recommendation(s) based upon information furnished by the Companies (or their representatives) and other sources and the said recommendation(s) shall be considered to be in the nature of non-binding advice, (our recommendation will however not be used for advising anybody to take buy or sell decision, for which

Chartered Accountants

HUMBAI

Recommendation of Share Entitlement for the proposed demerger of identified divisions of HFL into FRL after the proposed spin-off of these divisions into a wholly owned subsidiary of HFL

specific opinion needs to be taken from expert advisors). We have no obligation to update this Report.

The determination of Share Entitlement is not a precise science and the conclusions arrived at in many cases will, of necessity, be subjective and dependent on the exercise of individual judgement. There is, therefore, no indisputable single result. While we have provided our recommendation of the Share Entitlement based on the information available to us and within the scope and constraints of our engagement, others may have a different opinion as to the Share Entitlement. The final responsibility for the determination of the Share Entitlement at which the proposed demerger shall take place will be with the Board of Directors of the respective Companies, who should take into account other factors such as their own assessment of the proposed demerger and input of other advisors.

In the course of the valuation, we were provided with both written and verbal information, including market, financial and operating data of the respective Companies. In accordance with the terms of our engagement, we have assumed and relied upon, without independent verification, (i) the accuracy of the information that was publicly available and formed a substantial basis for this Report and (ii) the accuracy of information made available to us by the Companies. In accordance with our Engagement Letter and in accordance with the customary approach adopted in valuation exercises, we have not audited, reviewed or otherwise investigated the historical financial information provided to us. Accordingly, we do not express an opinion or offer any form of assurance regarding the truth and fairness of the financial position as indicated in the financial statements. Also, with respect to explanations and information sought from the Companies, we have been given to understand by the Management of the Companies that they have not omitted any relevant and material factors about the Companies and that they have checked the relevance or materiality of any specific information to the present exercise with us in case of any doubt. Our recommendations are based on the assumptions and information given by/ on behalf of the Companies. The Management of the Companies has indicated to us that they have understood that any omissions, inaccuracies or misstatements may materially affect our valuation analysis/results. Accordingly, we assume no responsibility for any errors in the information furnished by the Companies and their impact on the Report. Nothing has come to our attention to indicate that the information provided was materially mis-stated/ incorrect or would not afford reasonable grounds upon which to base the Report.

The Report assumes that the Companies comply fully with relevant laws and regulations applicable in all their areas of operation unless otherwise stated, and that the Companies will be managed in a competent and responsible manner. Further, except as specifically stated to the contrary, this Report has given no consideration to matters of a legal nature, including issues of legal title and compliance with local laws, and litigation and other contingent liabilities that are not recorded in the

Chartered Accountants

Recommendation of Share Entitlement for the proposed demerger of identified divisions of HFL into FRL after the proposed spin-off of these divisions into a wholly owned subsidiary of HFL

audited/unaudited balance sheet of the Companies. Our recommendation of value assumes that the assets and liabilities of the Companies, reflected in their respective latest balance sheets remain intact as of the Report date.

We are not advisors with respect to legal, tax and regulatory matters for the proposed demerger. This Report does not look into the business/ commercial reasons behind the proposed demerger nor the likely benefits arising out of the same. Similarly, it does not address the relative merits of the proposed demerger as compared with any other alternative business transaction, or other alternatives, or whether or not such alternatives could be achieved or are available.

The valuation and result are governed by concept of materiality. The financial forecasts used in the preparation of the report reflects Management's judgement, based on present circumstances, as to the most likely set of conditions and the course of action it is most likely to take. It is usually the case that some events and circumstances do not occur as expected or are not anticipated. Therefore, actual results during the forecast period will almost always differ from the forecasts and as such differences may be material. To the extent that our recommendations are based on forecasts, we express no opinion on the achievability of those forecasts.

The fee for the engagement is not contingent upon the recommendation of this Report. We owe responsibility to only the Boards of Directors of the respective Companies that have appointed us under the terms of our engagement letter and nobody else. We do not accept any liability to any third party in relation to the issue of this Report. It is understood that this analysis does not represent a fairness opinion.

This Report is subject to the laws of India.

Neither the Report nor its contents may be referred to or quoted in any registration statement, prospectus, offering memorandum, annual report, loan agreement or other agreement or document given to third parties, other than in connection with the proposed Scheme of Arrangement, without our prior written consent. In addition, this Report does not in any manner address the prices at which equity shares of the Companies will trade following announcement of the proposed demerger and we express no opinion or recommendation as to how the shareholders of either Companies should vote at any shareholders' meeting(s) to be held in connection with the proposed demerger.

APPROACH - BASIS OF DEMERGER

MUMBAI

The proposed Scheme of Arrangement contemplates the demerger of the Divisions of HFL into FRL. Arriving at the Share Entitlement for the proposed demerger of the Divisions of HFL into FRL would require determining the relative value of the

Chartered Accountants

Recommendation of Share Entitlement for the proposed demerger of identified divisions of HFL into FRL after the proposed spin-off of these divisions into a wholly owned subsidiary of HFL

identified divisions of HFL and the equity shares of FRL. These values are to be determined independently but on a relative basis, and without considering the current proposed demerger.

There are several commonly used and accepted methods for determining the Share Exchange for the proposed demerger, which have been considered in the present case, to the extent relevant and applicable, including:

- 1. Market Price method
- 2. Comparable Companies Quoted Multiples method
- 3. Discounted Cash Flows method
- 4. Net Asset Value method

It should be understood that the valuation of any company or its assets is inherently imprecise and is subject to certain uncertainties and contingencies, all of which are difficult to predict and are beyond our control. In performing our analysis, we made numerous assumptions with respect to industry performance and general business and economic conditions, many of which are beyond the control of the Companies. In addition, this valuation will fluctuate with changes in prevailing market conditions, the conditions and prospects, financial and otherwise, of the Companies, and other factors which generally influence the valuation of Companies and their assets.

The application of any particular method of valuation depends on the purpose for which the valuation is done. Although different values may exist for different purposes, it cannot be too strongly emphasized that a valuer can only arrive at one value for one purpose. Our choice of methodology of valuation has been arrived at using usual and conventional methodologies adopted for transactions of a similar nature and our reasonable judgment, in an independent and bona fide manner based on our previous experience of assignments of a similar nature.

Market Price ("MP") Method

The market price of an equity share as quoted on a stock exchange is normally considered as the value of the equity shares of that company where such quotations are arising from the shares being regularly and freely traded in, subject to the element of speculative support that may be inbuilt in the value of the shares. But there could be situations where the value of the share as quoted on the stock market would not be regarded as a proper index of the fair value of the share especially where the market values are fluctuating in a volatile capital market. Further, in the case of a merger, where there is a question of evaluating the shares of one company against those of another, the volume of transactions and the number of shares available for trading on the stock exchange over a reasonable period would have to be of a comparable standard.



Chartered Accountants

Recommendation of Share Entitlement for the proposed demerger of identified divisions of HFL into FRL after the proposed spin-off of these divisions into a wholly owned subsidiary of HFL

Since the equity shares of FRL are listed on the Bombay Stock Exchange and the National Stock Exchange, we have considered it appropriate to use the Market Price Method for valuing the shares of FRL. The valuation under this method is as per the Securities and Exchange Board of India (Issue of Capital and Disclosure requirements) Regulations. This method could not be used for the Divisions.

Comparable Companies' Quoted Multiple ("CCM") Method

Under this method, value of the equity shares of a company is arrived at by using multiples derived from valuations of comparable companies, as manifest through stock market valuations of listed companies. This valuation is based on the principle that market valuations, taking place between informed buyers and informed sellers, incorporate all factors relevant to valuation. Relevant multiples need to be chosen carefully and adjusted for differences between the circumstances.

We have not used this methodology for the valuation of FRL as corporate actions involving FRL in the past one year have affected the historical revenue and profitability and these are not representative of the current operations of FRL as noted in recent quarterly performance reported. We have not used this methodology for the valuation of the Divisions directly, but have applied the CCM method as a part of the approach under DCF method, as the current performance is not normative as per the Management of HFL.

Discounted Cash Flows ("DCF") Method

Under the DCF method, the projected free cash flows to the equity shareholders are discounted at the cost of equity. The sum of the discounted value of such free cash flows is the value of the firm. Using the DCF analysis involves determining the following:

Estimating future free cash flows:

Free cash flows are the cash flows expected to be generated by the company that are available to the providers of the company's equity capital.

Appropriate discount rate to be applied to cash flows i.e. the weighted average cost of capital:

This discount rate, which is applied to the free cash flows, should reflect the opportunity cost to the providers of capital. The opportunity cost to the equity capital provider equals the rate of return the capital provider expects to earn on other investments of equivalent risk.

We have used the DCF method for our valuation of Divisions by using the projections provided to us by the Management.



Chartered Accountants

Recommendation of Share Entitlement for the proposed demerger of identified divisions of HFL into FRL after the proposed spin-off of these divisions into a wholly owned subsidiary of HFL

Net Asset Value ("NAV") Method

The asset based valuation technique is based on the value of the underlying net assets of the business, either on a book value basis or realizable value basis or replacement cost basis. This valuation approach is mainly used in case where the firm is to be liquidated i.e. it does not meet the "going concern" criteria or in case where the assets base dominate earnings capability. A Scheme of Arrangement would normally be proceeded with, on the assumption that the demerged business will continue on a going concern basis and an actual realization of the operating assets is not contemplated. The operating assets have therefore been considered at their book values. In such a going concern scenario, the relative earning power is of importance, with the values arrived at on the net asset basis being of limited relevance.

Since the value of the Divisions and of FRL is largely driven by intangibles which are not captured in historical financials, NAV methodology has not been considered.

BASIS OF DEMERGER

As a precursor to the demerger, the Divisions of HFL are being spun-off into a wholly owned subsidiary of HFL, Heritage Foods Retail Limited ("HFRL"). We have been informed by the HFL Management that the transfer of the Divisions was done at book values. We have not verified the book values of the assets being transferred and have taken the HFL Management's representation for the same.

The basis of demerger of the Divisions into FRL would have to be determined after taking into consideration all the factors and methods mentioned hereinabove. As discussed above, we have applied DCF method in the case of the Divisions and Market Price method in the case of FRL to value the Divisions and FRL. Suitable rounding off may have been done in the values arrived at for the purpose of arriving at a whole number recommendation.

We have thus arrived at an enterprise value of Rs. 3,100.5 mn for the Divisions and a corresponding value of Rs. 2,955.7 mn after adjusting for debt, cash and surplus assets.

In light of the above, and on a consideration of all the relevant factors and circumstances as discussed and outlined hereinabove, we consider that the Share Entitlement of equity shares for the demerger of Divisions of HFL into FRL post spinoff into HFRL should be 17,881,890 (Seventeen million, Eight hundred and Eighty One thousand, Eight hundred and Ninety) equity shares of FRL (of INR 2/each fully paid up) considering a value of Rs. 165.29 per share of FRL, to be issued to HFL, in its capacity as the shareholder of HFRL.



Chartered Accountants

Recommendation of Share Entitlement for the proposed demerger of identified divisions of HFL into FRL after the proposed spin-off of these divisions into a wholly owned subsidiary of HFL

For reference purposes, at a value of Rs. 2,950.0 mn, the number of shares to be issued would be 17,847,420 (Seventeen million, Eight hundred and Forty Seven thousand, Four hundred and Twenty) equity shares of FRL, while at a value of Rs. 2,960.0 mn, the number of shares to be issued would be 17,907,919 (Seventeen million, Nine hundred and Seven thousand, Nine hundred and Nineteen) equity shares of FRL.

Respectfully submitted,

S.R. Batliboi & Co. LLP Chartered Accountants

ICAI Firm Registration Number: 301003E/E300005

10.01

MUMBAI

RED ACCO

Ravi Bansal

Partner

Membership No:49365

Place: Mumbai

Date: 7 November 2016



Report of the Audit Committee of Heritage Foods Limited (Company) recommending the draft Composite Scheme of Arrangement among Heritage Foods Limited ("Transferor Company" or "HFL") and Heritage Foods Retail Limited ("Transferee Company" or "Demerged Company" or "HFRL") and Future Retail Limited ("Resulting Company" or "FRL") and their Respective Shareholders and Creditors

The Committee was informed about the proposal to transfer the Retail, Agri & Bakery division of Heritage Foods Retail Limited to Future Retail Limited by leaving VetCa division in its wholly owned subsidiary company by way of demerger. All the above process shall be completed in a composite acheme of arrangement where as the Future Retail Limited shall issue the shares in a consideration of the business to the share holders of Heritage Foods Retail Limited (i.e. Heritage Foods Limited).

The Scheme inter-alia provides for:

- Slump Sale of Retail Business (comprising of grocery & food retail business, agri sourcing, processing & marketing business and bakery business) and Veterinary Care Business of the Company into HFRL;
- 2. Demerger of Retail Business from HPRL into PRL; and
- 3. Reduction of paid up equity share capital of HPRL.

The Committee was also informed that as per the requirement of SEBI (Listing Obligations and Disclosure Requirements) Regulation, 2015 listed Companies are required to place the following reports before its Audit Committee:

- Net Asset Valuation Certificate of Retail, Agri, Bakery & VetCa division of the Company submitted by M/s. Raju & Prasad, Statutory Auditor of the Company.
- 2. Draft Valuation Report of Swap Ratio of shares submitted by M/s. S.R. Batliboi & Co. LLP, Hyderabad.
- Draft Composite Scheme of Arrangement among Heritage Foods Limited ("Transferor Company" or "HFL") and Heritage Foods Retail Limited ("Transferee Company" or "Demerged Company" or "HFRL") and Puture Retail Limited ("Resulting Company" or "FRL") and their Respective Shareholders and Creditors.

The arrangement under this Scheme would benefit all stakeholders of the Transferor Company and will not be detrimental to public. Further, the proposed arrangement would inter alia achieve the following objectives:

- facilitate each business to be effectively integrated for achieving growth for each of the verticals independently;
- ii. enhance management focus and operational flexibility;
- iii. facilitate investment by strategic players;
- iv. create a platform to enhance financial flexibility to pursue growth;
- v. consolidation of the Retail operations of FRL and HFRL;
- vi. unlocking of value; and
- vii. synergies expected to bring in cost savings in the marketing, selling and distribution expenses for FRL

The Members of the Committee reviewed, noted the above reports and recommend to the Board for their approval.

For Heritage Foods Limited

D Seetharamaiah

Chairman of the Audit Committee

Dt: 07/11/2016 Place: Hyderabad





HERITAGE FOODS LIMITED

(Formerly known as Mis. Heritage Foods (India) Limited) CIN: L15209TG1992PLC014332 AN ISO: 22000 CERTIFIED COMPANY CYARGE IN

KEYNOTE

November 07, 2016

The Board of Directors Future Retail Limited Knowledge House, Shyam Nagar Off Jogeshwari - Vikhroli Link Road Jogeshwari (E), Mumbai – 400060

And

The Board of Directors Heritage Foods Limited 6-3-541/C, Panjagutta Hyderabad - 500080 Telangana, India

Dear Sir/Madam,

Reg: Fairness Opinion towards the valuation for the proposed demerger of the Retail, Agri & Bakery Divisions of Heritage Foods Limited ("HFL"), through its subsidiary Heritage Foods Retail Limited ("HFRL"), into Future Retail Limited ("FRL").

Keynote Corporate Services Limited ("Keynote" or "we" or "us") is Category I Merchant Banker registered with Securities Exchange Board of India ("SEBI"). We have been requested to issue a report on fairness of the valuation of the proposed demerger with respect to draft Composite Scheme of Arrangement between Future Retail Limited ("FRL") and Heritage Foods Limited ("HFL), Heritage Foods Retail Limited ("HFRL") and their respective shareholders and creditors. We have perused the documents/ information provided by you in respect of the said Arrangement and the Valuation Report as issued by S. R. Batliboi & Co. LLP (hereafter referred to as "Batliboi") dated November 07, 2016 and state as follows:

Company Profile:

Future Retail Limited is the flagship company of Future Group. FRL currently operates multiple retail formats in hypermarket, supermarket and home segments of the Indian consumer market including; Big Bazaar (hypermarket chain); FBB (Fashion at Big Bazaar); Food Bazaar (supermarket chain); Foodhall (Premium lifestyle food destination); easyday (Consumer retail department stores); Home Town (Home Improvement Store) and eZone (High end consumer electronics specialty store).

Heritage Foods Limited founded in the year 1992 is one of the fastest growing private sector enterprises in India, with six business verticals viz., Dairy, Retail, Agri, Bakery, Renewable Energy and VetCa. Presently Heritage dairy has market presence in Telangana, Andhra Pradesh, Karnataka, Kerala, Tamil Nadu, Maharastra, Odisha, Haryana and Delhi NCR. 113,500 retail outlets sell Heritage Products in all these states and it has 124 Heritage Fresh retail stores across Hyderabad, Chennai and Bangalore. Integrated Agri pack houses are in Chittoor in Andhra Pradesh and Medak in Telangana and these serve the Heritage Fresh retail stores. The company has a bakery production facility in Hyderabad. Under the Renewable Energy vertical, the company has one solar (2.34 MW) and two wind (4.2 MW) power projects for captive consumption of its dairy factories.

ge_

Page 1 of 4

KEYNOTE

Transaction background: Future Retail Limited proposes to acquire three divisions - Retail, Agri and Bakery from the wholly owned subsidiary - HFRL of Heritage Foods Limited. FRL shall be discharging the consideration by issuing the FRL shares to the shareholders of HFRL ("Transaction").

Rationale of the Report:

We have been informed that, pursuant to a Scheme of Arrangement under Sections 391 to 394 read with Sections 100 to 103 of the Companies Act, 1956 and Section 52 of the Companies Act, 2013 and other relevant provisions of the Companies Act, 2013 (hereinafter referred to as "the Scheme") and subject to necessary approvals, the "Demerged Undertaking" of HFRL, comprising of Retail, Agri and Bakery Divisions and Vetca Division "of HFL being transferred to HFRL as part of Slump Sale and Retail, Agri and Bakery Divisions would be demerged into FRL.

In this regard, we have been requested to suggest a Fairness Opinion on the valuation for the proposed demerger of "Retail, Agri and Bakery Divisions" of HFL, through its subsidiary HFRL, into FRL.

Sources of Information:

For arriving at the fairness opinion set forth below, we have relied upon the following sources of information:

- a) Valuation Report by S. R. Batlibol & Co. LLP dated November 07, 2016;
- Historical Financial statements of the Retail, Agri and Bakery divisions of HFL for the year ended March 31, 2015 and year ended March 31, 2016;
- Projected Financials of the Retail, Agri and Bakery divisions of HFL for FY 2016-17 to FY 2020-21;
- d) Draft Scheme of Arrangement;
- Other relevant information/documents regarding HFL including information available through public domain

In addition to the above, we have also obtained such other information and explanations, which were considered relevant for the purpose of our analysis.

Our Recommendation:

As stated in the Valuation Report, S. R. Batliboi & Co. LLP has recommended the following:

Demerger of "Retail, Agri and Bakery Divisions" of HFL into FRL: "Considering Rs.165.29/share as the highest price, the number of shares to be issued to HFL by FRL based on a combined equity value of Rs. 2,955.7 million of the Retail, Agri and Bakery Divisions should be 17,881,890 equity shares. For reference purposes, at a value of Rs. 2,950.0 mn, the number of shares to be issued would be 17,847,420 equity shares of FRL, while at a value of Rs. 2,960.0 mn, the number of shares to be issued would be 17,907,919 equity shares of FRL".

The aforesald arrangement shall be pursuant to the Draft Composite Scheme of Arrangement and shall be subject to receipt of approval from the Jurisdictional High Court of Bombay and Hyderabad and other statutory approvals as may be required. The detailed terms and conditions of the demerger are more fully set forth in the Draft Scheme of Arrangement. Keynote has issued the fairness opinion with the understanding that Draft Scheme of Arrangement shall not be materially altered and the parties hereto

gen-

Page 2 of 4

KEYNOTE

agree that the Fairness Opinion would not stand good in case the final Scheme of Arrangement alters the transaction.

Based on the information, data made available to us, including the Valuation Report, to the best of our knowledge and belief, the valuation as suggested by S. R. Batliboi & Co. LLP proposed under the Scheme of Arrangement is fair in our opinion.

Exclusions and Limitations:

We have assumed and relied upon, without independent verification, the accuracy and completeness of all information that was publicly available or provided or otherwise made available to us by HFL for the purpose of this opinion. With respect to the estimated financials provided to us by the managements of HFL, we have assumed that such financials were prepared in good faith and reflect the best currently available estimates and judgments by the managements of HFL. We express no opinion and accordingly accept no responsibility with respect to or for such estimated financials or the assumptions on which they were based. Our work does not constitute an audit or certification or due diligence of the working results, financial statements, financial estimates or estimates of value to be realized for the assets of HFL. We have solely relied upon the information provided to us by HFL. We have not reviewed any books or records of HFL (other than those provided or made available to us). We have not assumed any obligation to conduct, nor have we conducted any physical inspection or title verification of the properties or facilities of HFL and neither express any opinion with respect thereto nor accept any responsibility therefore. We have not made any independent valuation or appraisal of the assets or liabilities of HFL. We have not reviewed any internal management information statements or any non-public reports, and, instead, with your consent we have relied upon information which was publicly available or provided or otherwise made available to us for the purpose of this opinion. We are not experts in the evaluation of litigation or other actual or threaten claims and hence have not commented on the effect of such litigation or claims on the valuation. We are not legal, tax, regulatory or actuarial advisors. We are financial advisors only and have relied upon, without independent verification, the assessment of HFL with respect to these matters. In addition, we have assumed that the Draft Scheme of Arrangement will be approved by the regulatory authorities and that the proposed transaction will be consummated substantially in accordance with the terms set forth in the Draft Scheme of Arrangement.

We understand that the managements of FRL and HFL during our discussion with them would have drawn our attention to all such information and matters which may have an impact on our analysis and opinion. We have assumed that in the course of obtaining necessary regulatory or other consents or approvals for the Draft Scheme of Arrangement, no restrictions will be imposed that will have a material adverse effect on the benefits of the transaction that FRL and HFL may have contemplated. Our opinion is necessarily based on financial, economic, market and other conditions as they currently exist and on the information made available to us as of the date hereof. It should be understood that although subsequent developments may affect this opinion, we do not have any obligation to update, revise or reaffirm this opinion. In arriving at our opinion, we are not authorized to solicit, and did not solicit, interests for any party with respect to the acquisition, business combination or other extra-ordinary transaction involving FRL & HFL or any of its assets, nor did we negotiate with any other party in this regard.

We have acted as a financial advisor to FRL and HFL for providing a fairness opinion on the proposed transaction and will receive professional fees for our services.



KEYNOTE

In the ordinary course of business, Keynote is engaged in securities trading, securities brokerage and investment activities, as well as providing investment banking and investment advisory services. In the ordinary course of its trading, brokerage and financing activities, any member of Keynote may at any time hold long or short positions, and may trade or otherwise effect transactions, for its own account or the accounts of customers, in debt or equity securities or senior loans of any company that may be involved in the transaction.

It is understood that this letter is solely for the benefit of and confidential use by the Board of Directors of FRL and HFL for the purpose of this transaction and may not be relied upon by any other person and may not be used or disclosed for any other purpose without our prior written consent. The opinion is not meant for meeting any other regulatory or disclosure requirements, save and except as specified above, under any Indian or foreign law, Statute, Act, guideline or similar instruction. Management of FRL and HFL should not make this report available to any party, including any regulatory or compliance authority/agency except as mentioned above. The letter is only intended for the aforementioned specific purpose and if it is used for any other purpose; we will not be liable for any consequences thereof.

We express no opinion whatever and make no recommendation at all as to FRL's or HFL's underlying decision to effect to the proposed transaction or as to how the holders of equity shares or preference shares or secured or unsecured creditors of FRL and HFL should vote at their respective meetings held in connection with the transaction. We do not express and should not be deemed to have expressed any views on any other terms of transaction. We also express no opinion and accordingly accept no responsibility for or as to the prices at which the equity shares of FRL and HFL will trade following the announcement of the transaction or as to the financial performance of FRL and HFL following the consummation of the transaction.

In no circumstances however, will Keynote Corporate Services Limited or its associates, directors or employees accept any responsibility or liability to any third party and in the unforeseen event of any such responsibility or liability being imposed on Keynote Corporate Services Limited or its associates, directors or employees by any third party, FRL, HFL and their respective affiliates shall indemnify them.

For KEYNOTE CORPORATE SERVICES LTD

Nipun Lodha

Executive Vice-President & Head- Corporate Finance

SEBI Registration No. INM000003606

(Merchant Banker)



Shareholding Pattern Under Regulation 31 of SEBI (Lisiting Obligation and Disclosure Requirements) Regulation, 2015

- 1. Name of Listed Entity: HERITAGE FOODS LIMITED
- 2. Scrip Code/Name of Scrip/Class of Security: 519552, HERITGFOOD, EQUITY SHARES
- Share Holding Pattern Filed under: Reg. 31(1)(a)/Reg.31(1)(b)/Reg.31(1)© PRE & POST Scheme of Arrangement based on 30th September, 2016
 - a. If under 31(1)(b) then indicate the report for quarter ending 30/09/2016
 - b. if under 31(1)(c) then indicate date of allotment/extinguishment
- Declaration: The Listed entity is required to submit the following declaration to the extent of submission of information:

	Particulars	YES*	NO*
ā	Whether the Listed Entity has issued any partly paid up shares		*
b	Whether the Listed Entity has issued any Convertible Securities or Warrants?		
C	Whether the Listed Entity has any shares against which depository receipts are issued?		
ď	Whether the Listed Entity has any shares in locked-in?		*
e	Whether any shares held by promoters are pledge or otherwise encumbered?		

*if the Listed Entity selectes the option 'NO' for the questions above, the columns for the partly paid up shares, Outstanding Convertible Securities/Warrants, depository receipts, locked-in shares, No of shares pledged or otherwise encumbered by promoters, as applicable, shall not be displayed at the time of dissemination on the Stock Exchange website. Also wherever there is 'No' declared by Listed entity in above table the values will be considered as 'Zero' by default on submission of the format of holding of specified securities.

5 The tabular format for disclosure of holding of specified securities is as follows:

Note: For the Draft Scheme of Arrangement Heritage Foods Limited has not issued/received any shares from Future Retail Limited. Hence there is no change in the Share Holding of the Company for pre & post arrangement.







Number of equity shares held in dematerialized form 22188722 9256846 12931876 (XIX) As a % of total Shares held Number of Shares pledged or otherwise encumbered 0.00 ¥ 0.00 0.00 NA 0.00 NA 0.00 NA As a % of total N Shares held 0.00 Number of Locked in Shares Shareholding as a %

& assuming full conversion
of convertible Securities N

(as a percentage of
diluted share capital) 39.90 0.00 100.00 (XI) No of Shares Underlying an Outstanding convertible securities (including Warrants) 8 Total as a % of (A+B+C) 39.90 100.00 0.00 Number of Voting Rights held in each class of securities 9256846 0 23199000 No of Voting Rights Class Y 9256846 100.00 Class X 39.90 Shareholding as a % of total no of shares (As a % of (A+8+C2)) 0.00 (VIII) Total No of Shares Held (VII) = (IV)+(V)+(VI) 23199000 9256846 S ŝ No of Partly paid-up equity shares held ε No of fully paid up equity shares held 9256846 23199000 Ξ No of Shareholders 16211 16225 (III) Public
Non Promoter-Non Public
Shares underlying DRs
Shares held by Employes Trusts Category of Shareholder Promoter & Promoter Group (ii) Category



Table I - Summary Statement holding of specified securities

Stareholding Pattern Under Regulation 33 of SERI (Listing Chilgation and Disclosure Regularments) Regulation, 2015

Cythag any	Category & Harver of the Shareholder		Ann of Eulip paid up equit phores held	the of Purely paid up equity shares hald	No of Shares Underlying Depository Basisipts	Total No of Sween Held (IN-V-H)	Shareholding as a 'S of total so of shares (adoutined as per SCRR, 1957 (VIII) As a 'S of (A+8-C3	Microllan of Voting Rights held in each class of occurring			No of Shares Underlying Databashing pomertible socurities (minding Warnesh)	Shareholding an a % assuming full convention of convention of convention Securities (as a persentage of discost share capital) (VV)=(X) As a % of (A+8+CX)	Number of Locked in Shares		Number of Shares pludged or otherwise accombared		Number of equity shares held in dematerialized form	
_								Model	Voting F	Eleten	Total ace % of			No.	As a 16-of footal	No.	As a % of total	0
			_	_				1000			(Anti-c)			-	Shares hald		Shares bett	_
	Dane Holding Pettern Flied under: Rep. 3111)(e)/Rep.211(2)(r/Rep.211(2)) Pre Share of Arrangement Blassed on 38th September, 2016							Class.X	Clarys 1	Tytel		1 = 1 = 1						
-	W 100.5WC0000000000000000000000000000000000	316	990	M	0/0	ties	DARE			.191		90	101	_	DES	_	Doel	DEN
20	relat							20000	-		30.00	-	10.44	_		-	0.00	800104
ń.	melondusty/Hooks undivided Femily	1.5				6003040			-	0 10011146	31.43		36.63		9.00			
	RAMAKKISHNA MENDAMKIRI	1	390			200				g 200	3.00		0.00	-	4.00	_	0.00	
	V RADA RAJA NAVDA	1	\$0000			50000				0 50000	0.23		0.32	_	0.90		0.00	
	NAMA BHUNNASSWARI		1300626		- 0	5330826				0 1830816	22.98		22.58		9.90		0.00	
	THROUGHUR VASURGARA	1	6490		- 0	6400			-	9 5400	0.03		0.01	-	8,90		0.00	
	HARA LORGSH	1	7369490		- 1	2565400				0 1365-400	35.20		30,20	- 13	0,000		0.00	
	A SAN SANGAR PRASAD	1	800		-	800				9 800	1,00		0.00		990		6:00	
7_	PATHRA N F	1	30000		-	93000				30000	8.0		0.18				0.00	
	DURGA RABIAONSHIIA N P	1	23000		- 1	20000				30000	2.09		0.09		0.00		0.09	
	AVANDAMUNI BAUAKKISHINA	7	7400			7420				3 7410	3.01		4.00		580		0.00	
	NAMERICIPA N.P.	1	100000	-	- 0	100000				3 100000	3.41		0.42		9.80		0.00	
	NEIUMAN F.		39000	-	-	20000				0 20800	8.59		0.99		y 230		0.00	
	N BRIHDBANI	1 2	3,00,000	- 0	-	121000			OI .	0 331000	8.44		3.40	-	100		0.00	
100	Central Government/State Devenment(s)	-	0.000	-	-		15.00		-	2 0	1.00		1.00	-	0.00		0.00	
12	Financial Institutions/Banks	-	-		-				-	0 1121800			3.28		0.00		0.00	
0_	Any Differ	-	1213/00			1303000				1 177/800	1.28		5.28	-	0.00	-	0.00	
_	MERAND FRANCE & INVESTMENT PAT CTS	- 3	1212800		1 - 2	323556				1 1256896	24.50		39.90	-	0.00	-	0.00	-
-	Audi-Telal (A(E1)	14	1256946	-	-	3099	10.00	SU-MARK	-	12,000	20,00		1					
20	Foreign	-	_	_	_		_	-	-	_								
	Individuals (Non-Passident Individuals	1 7					8.00	1 2	1	e	0.00		0.00		0.00	1 0	6.00	48 35
8.	(Tonign Individuals	1 1	1			1	5.00		1	p) 17	1.00		0.00		0.00	0	0.00	
21_	Continued	1 7		0	1 0		0.00		2	p 0	0.00		0.00		0,00		9.00	
72	Intifuliara			1			-						200		- 17			
141	Farvige Portfolio Intestal	- 1	- 3			-	9.80		2	D	0.00	-	6.00		0.00	1	0.00	
100	To the second se	1					5.00		y ·	ol a	0.00	- 0	0.00		0.00	- 1	0.00	
141	Any Other	1					-											
	Sub-Total (ASS)	- 1					0.30	- 3	5	0 0	5,00	- 4	0.00		0,00	- 0	1.00	
	Total Sweethalding of Francisco and Promoter Group (A)-(A)(1)-(A)(3)	1 1	*21080			80000	39.30	925184		0 1298041	E11.00		99,00	0.00	9.00		3.00	925004



Shareholding Pattern Under Regulation 25 of 5251 (Celling Chilgarius and Studiance Requirements) Regulation, 2015.

Carried Surv	Category's Name of the Shareholder	No of Herobol des		No of Partie policies regular characteristic	to of Scene sectoristing Superitory Brodight	Total No of Shares HARE (Shretros)	a distance on					No of Shares (Industrying Contracting posterities projecting Discharing Warmend)	Alternative being per a 16 menorality fail operate being fail operate being Securities (in a persentage of allernat disease agents)		of Landard In Orderes	Municipal of Wales pludged at advancing enterthined		Member of apply shares had to female initial faces
									to of Volley	Rightee	Total as a North (Andrei)			*	As a Kird lotal Shares bein	He.	As a % of hotal Shares held	
	Share Holding Fathern Filtel under Rep. 11(1)(4)/Rep. 21(1)(6)/Rep. 12(1)(6 - File Shares of Autoingminent Shares on Sidn September, 20(6)							Class II	Clais Y	Tetal	0.001/2.0						1	
	The second secon	190	350	76	.046	000	pon			100		20	DEG		(000		para	pkrvt.
-	PATEASON.												10.00				1	10000
2-	Mutual Fands	11	1318355	- 0	- 0	1356135	9.71	LEMIN	v	E 1500.66	5.77	- 1	8.77	- 1	8.08	NA.	166	13301
-	BUREARAN MUTUAL FUND A/C	_	-			-												-
	SHEHAM SMILE FIRE	1 0	1000			85,520	8.51	10000		0 81536	100	- 0	330	- 4	606	164	164	8131
_	TATA HELIAM SAVINES ESSITE FUND	1 3	265800		-	3696		265400		0 361900			1.15	-	6.00		NA.	3459
-	COVIDATION CONTRACTOR	_					D30		-		0.00	- 2	5.00	- 4	0.00		NA.	-
R1	Venture Capital Funds	-	-			-		_	-	-	3.00	- 4	500		0.00		NA.	-
12	Alternate Investment Fusilis	-	-			-	0.00	-	-	9 9		- 4		- 3			PAA.	+
MI.	Foreign Vertiars Coptial breasters	-	1				0.30	-	-	-	1.00		500		6.06			43130
11	Foreign Partholic moreclast	- 15	1337500			33(700)	5.73	1,007503	-	E 197911	5.72	- 1	8.72	- 4	6.08	- NA	NA.	13070
	GORE AGA PADIFE SMALL GAP	100	2330	100		7 1000	200	EXX	1.	G 2.000	1763		1000	100	177.533	1100.5	433.00	3.33
	SAMURITUS SAMUED	14 13	358000	- 6		35400	5.34	918000		17 158200	1.58		134	- 3	606	515	166	2180
-	(MIRSURGENT FUND	1.1	46725			44000	1.80	3440333		E 440735	1.90	- 3	1.90	- 3	0.00		96	4401
m .	Financial Indifficients	- 2	2140			3940	0.00	210	1	11 1160	3.01		500	- 3	6.00	144	166	1 13
hád	moutains Cempenter						0.00		h .		0.000	- 10			0.00	BA	- 66	
ini .	Provident Fundu/Fernico Funds	- 1					0.20			0 0	0.000	- 1	6.00	- 0	6.00	164	164	
0	Avy Citier		V		2					4.0	0.710		0.000		1000		1000	
_	Sub Tarke (MCL)	- 40	2007314			2967526	15.50	1967714	(0 1967919	33,400	- 0	63.90		6.00	- NA	NA.	24481
_	Carettal Government/State Government(s)	_						-										
nie i	President of India	100					0.00			100	16200	- 0	1000	- 4	0.00	166	166	100
44		-		1			6.80			6 4	0.00	1.0	8.00	- 1	6.00		194	
	Sub Trest (NGD)	_	1			_	-	_	1	4-	- 300							_
-	No-satura.	-					_		_	_							_	+
	Lingbolous shareholders habiling coming	309	40000	174		400895	17.26	+00000		0 400000	17.16	19	17.28		0.00	NA.	NA.	10000
<u> </u>	offere copital up to the 2 lette	20.90			-	50000	11,00	50000	4	- 100000	1.45	_	15.00			194	1	1000
	8 (redinional altern-holders holding nominal)		Parameter.		1 1 1 1 1 1 1 1 1 1 1 1 1	325400	14.05	3054650		0 105465	14.09		14/09		000	NA.	RS.	33546
_	share repital in mores of its. 2 Little		3754031	- 9	-				-				1.28	-	6.80		PA PA	1960
_	ALTH AGRENIE	_	29600			29933		25600		0 794000				- 2			NA.	3460
_	KANNU DISAMBUNDE KABA	-	14000		-	10000				E 384000	1.08		1.06		0.00			
_	Y BURNE SWINGER		116030		_	116129	5.25	19558	-	0 1161250	1.00	- 3	1.01	- 1	0.00	- 74.7	755	LMD
_		-	_			_	_		-				-					_
NI.	WENCE Fing Elected with Fill	_	- X	9	- 9				0	0 30			200	- 0	0.80	25	755	-
ы_	Employee Trush			- 9			0.20				3.00	- 0	0.00	- 2	0.00	PAR	AN .	-
for !	Overtosas Depositorias (Huilding DRs)				100	5.01	1			100	100		1000		in the		1	1 2
H	(Mahaning Egure)			- 0			0.00			70 8	1.00		2,00	. 7	0.30	AA	744	-
111	any Other					100												-
	TRUSTS	- 1	200	9	- 0	200	0.30	. 28	0	0 300	1.00		70,000	2	0.00		.04	2
	OWERS AS CORPORATE SOLES	1	9000	0		100		8000	0	6 8000	9.00	- 00	0.85		0.30	84	66	1
	NON RECEIPT POUND	. 546	397647			94794	1.71	267640	1	E 26562	1.75	- 14	1.71	- 6	0.00	No.	- As	3054
	CUTANAG MEMBURE	316	86490			8549	6 037	86436	6	0 00496	3.67		19.81	- 1	0.00	AA.	Aris .	864
	NOT NON-ACCUTATION	362				640,71			1	.0 34177	3.25			. 0	0.00		66	560
	ECONS COMPONERS	410				Saic2vd1			7	0 3440487	14.99	- 0	36.80	D	0.00		55	34416
	REDIA SECURITES PRINCIPE LIMITED		19600			29400				0 254000	1.05	- 1	1.01	- 0	0.20		Ass	7940
	HIROMAN HOLDINGS PRINATE LIMITED		2173841		-	367386				10 2517280	11.09		11.09	- 1	0.00		44	25729
		18194				1127463	-			1127409	48.81	-	46.60	-	0.00			161821
	Nuls Tetal (10(1)	18329	1317/4618			ALC: N	1000		_	100,000				-	2.00		_	-
	Tutal Addic Stendolding (III)																	



Mannholding Pattern Linder Regulation 21 of 5001 (Linking Obligation and Discissors Requiremental Regulation, 2003

Calleg	Category & Norms of the Storeholder		Catagory & Norms of the Waterfulled	res	No of Sharehold art	No of fully park up squiry shares held	ino of Perily paid-up equity shared hald	No of Stares Underlying Depositiony Seculpts	Turtal No. of Size on Held (Version)	Sterological as a N of Local as of shares (Arthritis)	Marries	of Voting 1	Egina halid maritim	in much date of	the of Shares Limitedying Outstanding Convertible securities (Sectioning Western)	Sharefulding as a % assuming full assuming full assuming full assuming full properties launting full properties launting of attachment above repties)			Number of Sisses photosic or otherwise recutificated	Number of anality shares hald in demarks fallend face.
Н									Sea of Verting Rights			Total at a 74 of (Articl			No.	Shares held	No. As a North Local Strange backs			
	there making Pattern (field coder) ling \$15126-5/big \$100000/big \$10136 - the Shame of Artengament (Based on \$105 beptimizer, \$2015)								Clean K	(Gase #	Sotal									
		-00	(41	197	M	190	DOE	000			449	4	(16)	.00		Dent	241	2870		
(T)	Curtodian/04 Walter		1	0	P	_ 3		0.00		0	0	05 0.00		6.00	_	0 8.00	4A (9A			
(E)	Triglance Sanaff 1 tute (pider 193) (Franc Sanari Triglance Sanarii) Espainium, 2014)		-					0.00		ti.	0	0 000		8.00		10 000	NA HA	3		
	Total Non-Promoter-Hon Public Shareholding (II) a COSLI-COSII							0.00			0			8.00		1 10				



Shareholding Pattern Under Regulation 31 of SEBI (Lisiting Obligation and Disclosure Requirements) Regulation, 2015

Name of Shareholder	Name of PAC	No of shares	Holding%
RAMAKRISHNA NANDAMURI	70.000	200	0.00
V NAGA RAJA NAIDU		50000	0.22
NANDAMURI VASUNDARA		6400	0.03
A SIVA SANKARA PRASAD		800	0.00
KATHYA N P	14	30000	0.13
DURGA RAMAKRISHNA N P		20000	0.09
NANDAMURI BALAKRISHNA			0.00
N BALAKRISHNA			0.00
RAMAKRISHNA N P			0.00
NEELIMA N P			0.00
N BRAHMANI			0.00
Total:		107400	0.46





Heritage Foods Retail Limited

SHARE HOLDING PRE-TRANSACTION

Name of the Company: Heritage Foods Retail Limited

Authorized Capital: Rs.15,00,00,000 (divided into 1,50,00,000 equity shares of face

value Rs.10/- each)

Paidup Capital: Rs.16,56,000 (divided into 1,65,600 equity shares of face value

Rs.10/- each)

S. No	Name of the Share Holder	No. of Shares	Amount (Rs.)	% of total Holding	Remark
1	Sri. D. Seetharamaiah	100	1000	0.00	
2	Smt. N. Bhuvaneswari	100	1000	0.00	
3	Dr. M. Sambasiva Rao	100	1000	0.00	Beneficial
4	Sri. N. Lokesh	100	1000	0,00	Holder
5	Smt. N. Brahmani	100	1000	0.00	
б	Sri. Rajesh Kilaru	100	1000	0.00	
7	M/s. Heritage Foods Limited	165,000	16,50,000	100.00	
	Total:	1,65,600	16,56,000	100.00	

SHARE HOLDING POST-TRANSACTION

Name of the Company: Heritage Foods Retail Limited

Authorized Capital: Rs. 15,00,00,000 (divided into 1,50,00,000 equity shares of face

value Rs. 10/- each)

Paidup Capital: Rs. 14,16,56,000 (divided into 1,41,65,600 equity shares of face

value Rs. 10/- each)

S. No	Name of the Share Holder	No. of Shares	Amount (Rs.)	% of total Holding	Remark
1	Sri. D. Seetharamaiah	100	1000	0.00	
2	Smt. N. Bhuvaneswari	100	1000	0.00	
3	Dr. M. Sambasiva Rao	100	1000	0.00	Beneficial
4	Sri. N. Lokesh	100	1000	0.00	Holder
5	Smt. N. Brahmani	100	1000	0.00	
6	Sri. Rajesh Kilaru	100	1000	0.00	
7	M/s. Heritage Foods Limited	1,41,65,000	14,16,50,000	100.00	
	Total:	1,41,65,600	14,16,56,000	100.00	

// Cartified True Copy //

Regd. Office: #6-3-541/C, Panjagutta, Hyderabad - 500082

FUTURE RETAIL!

Shareholding Pattern under Regulation 31 of SEBI (Listing Obligations and Disclosure Requirements) Regulations, 2015

2	Scrip (ode/Name of Scrip/Class of Security: 540064 / FRETAIL / Equity Shares
3	Share	Holding Pattern Filed under: Reg. 31(1)(a)/Reg. 31(1)(b)/Reg.31(1)(c) : - Pre Scheme of Arrangement (based on September 30, 2016)
	a.	If under 31(1)(b) then indicate the report for Quarter ending
		If under 31(1)(c) then indicate date of allotment/extinguishment:

Ξ	Particulars	Yes*	No*
	1 Whether the Listed Entity has issued any partly paid up shares?		No
_	2 Whether the Listed Entity has issued any Convertible Securities or Warrants?		No
	3 Whether the Listed Entity has any shares against which depository receipts are issued?		No
Ī	4 Whether the Listed Entity has any shares in locked-in?	Yes	
_	5 Whether any shares held by promoters are pledge or otherwise encumbered?	Yes	

* If the Listed Entity selects the option 'No' for the questions above, the columns for the partly paid up shares, Outstanding Convertible Securities/Warrants, depository receipts, locked-in shares, No of shares pledged or otherwise encumbered by promoters, as applicable, shall not be displayed at the time of dissemination on the Stock Exchange website. Also wherever there is 'No' declared by Listed Entity in above table the values will be considered as 'Zero' by default on submission of the format of holding of specified securities.

Note:

In the above paid up capital, 1,542 Optionally Convertible Debentures (OCDs) of Rs. 10 lakh each, issued pursuant to the Composite Scheme of Arrangement between Future Enterprises Limited (earlier known as Future Retail Limited) with the Company, becoming effective on May 1, 2016, have not been considered as the conversion price of these securities would be determinable at any time during the tenure of the OCDs or latest by October 31, 2017, when the option for conversion is exercised. Further, as per terms of isssue, these securities may be redeemed in case conversion option is not exercised.



Table 1 - Summary Statement holding of specified securities

Category	Category of shareholder	of	No. of fully paid up equity shares held	No. of Partiy paid-up equity	No. of shares underlyin	Yetal nes. shares held	Shareholding as a % of total no. of shares (calculated as	securities	Voting Ai	ghts held in ea	nch class of	No. of Shares Underlying Outstanding	Shareholding , as a % assuming full conversion of	Number of Loc shares	cked in	Number of Sha or otherwise e		Number of equity shares held in dematerialise
				shares held	Depositor y Receipts		per SCRR, 1957)	No of Votin	E Rights		TOTAL MAN AT THE	convertible securities (including	convertible securities (as a percentage	No. (a)	As a % of total Shares	No. (a)	As a % of total Shores	d form
								Class eg: X	Class eg: y	Total		Warrants)	of diluted share capital)		hefd[b)		held(b)	
107	(40)	(111)	(M)	(v)	(M)	(M)+(M)=	(VIII)As a % of (A+B+CZ)			(00)		(XI	(XI)= (VII)+(X) As a % of (A=B+CZ)	(xi	0	(xu	0	Onn
(A)	Promoter & Promoter Group	26	230115436		0	230115436	48.8217	230115436		230115436	48.8217		48.8217	94267712	40.9654	175223483	76.145	230115436
(8)	Public	44032	241223121		0	241223121	51,1783	The second of the later who	_	241223121			51.1783	4347826	the forest parties and the fact was		MA	240077496
(C)	Non Promoter - Non Public													- 7777				-
(C) (C1)	Shares Underlying DRs	0	0		0	0	0.0000	-	2	0 0	0.0000	- 0	0.0000		0.0000	0	NA.	0
(C2)	Shares Held By Employee Trust	0	0	. (0	0	0.0000			0	0.0000		0.0000		0.0000		NA	0
2000	Total	44058	471338557	- 1	0	471338557	100.0000	471338557	1	471338557	100.0000		100.0000	137745973			37.1757	470192932



	Category & Name of the shareholders	PAN	Nos. of sharehold ers	No. of fully paid up equity shares held	Partly paid up equity shares held	shares underlying Depositor	Total nos. shares held	Shareholdi ng % calculated us per	Number of V securities	oting 8	tights held in e	rach class of	No. of Shares Underlyin E	full	shares	doed in	Number of Sha pledged or oth encumbered	erwise	Number of equity shares held in domaterialise
						y Receipts		SCRR, 1957 As a % of (A+6+C2)	No of Voting	Alghts	8	Total as a % of (A+B+C)	Outstandi rig convertibil	of	185000	As a % of total Shares	No. (a)	total Shares	d form
								Y		Class eg: y	Total		securities (including Warrants)	as a		(held(b)		held(b)	
	0)	011	(10)	(W)	(v)	(M)	(MI) = (MI) =	(VIII) As a % of (A+B+C2)			100		(8)	(XI)= (VII)+(X) As a % of (A+B+C2)	; (x)	0	(ten)	(MV)
	1 Indian												_		_				
39	Individuals / Hindu Undivided			05.20			100											115	7 90
(a)	Family	1::		88115	. (0	88115	0.0387	88115		881.15	0.0197		0.0180	7	0.0000		9.0000	8811
	Anii Diyani			2121	. (0	2121	0.0004	2223		2121	0.0004		0,000	0	0.000		0.0000	212
	Ashni Kishore Biyani		3	71347		0	71147	0.0151	71147	- 6	71147	0.0151		0.0151	L I	0.0000		0.0000	7114
	Gopikishan Biyani			1 2121		0	2121	0.0004	2121		2171	0.0004	(0.000	6	0.0000		0,0000	
	Kohdre Byani			2)21		0	2121	0.0004	2121	- 6	2121	0.0004		0.0004	4	0.000		0.0000	
	Laxminorayan Biyoni			2121		0			2121		2121	0.0004		0.0004		0.0000	1	0.0000	The second secon
	Rakesh Biyani			2321	- (2121	0,0004	2121		7171	0.0004		0.000	a e	0.000	9 (0.0000	Approximately 17 Table
	Sunii Bişani			2321	. (-	2121		2121		2121	0.0004	1	0.0004	4	0.0000	1	0.0000	
	Vijay Bigani		- 3	2.101	- 1		2121		2121		2121	0,0004		0.0004		0,000		0.0000	
	Vivek Biyani			2121	- 1	0	2121	0.0004	2121	- 0	2121	0.0004	4 (0.0004	4	0.0000	0	0.0000	212
50	Central Government / State		1 5				100	1				10000				1000		1000	
(h)	Government(s)	-		0		0	0	0.0000	a	_	0	0.0000		0.0000		0.000		0.0000	
(0)	financial Institutions / Banks		12					0.0000	0	- 1	0 0	0.0000		0.0000		0.000		0.0000	
(4)	Any Other (Specify) Bodies Corporate	_	17				The state of the s	and the second second second	210027321	_	230027321	48.8030		48.803			and the same of th	and the same of th	Committee and the second of the
	Future Corporate Resources			23002321			230027321	48.8030	230027321	Η,	230027321	48.8030	-	48.803	9426771	2 40.981	17522348	76.1751	23002732
	Lambad			177545090			177545090	37,6683	177545090		17/545000	37,6683		37,008	3 10060768	45,3286	14777756	83.2338	17754509
	PL Industries Limited				1	0	44136090	9.3640	44136090		The International Assessment Contractions	9.3640		9.364		PROFESSION AND ADDRESS OF THE PARTY OF THE P	the same of the sa		
	Gargi Business Ventures Frivate Limited			A ROOM S			4550000	0.9853	4550000		309909	0.9632	1	0.965	30000		0817.0	100000	S SAMESTA
	Ryka Commercial Ventures Private Limited						1688663	THE REAL	1684663		1684663	0.3574		0.357				0.0000	10000
	Mana Retail Private Limited			and the second s	1	0	1579103	0.1350	1579103		_	0.3350		0.335		0.000			The second second
	Future Capital Investment Private			-			201000	4.50	2075200	-	1019368	winds.		0.500	1	1	-	1	-
	Limite			531375			531,775	0.1127	531375		531375	0.1127	7	0.112	7	0.000	0	0.0000	53137
	Akar Estate And Finance Private Limited																	0.0000	
	Sub Total (A)(1)		26			-	1000	0,0002	1000	-	1000	0.0000		0.000		0,000			
	There some body)		20	430113436		0	230115416	98.3717	230115436	1	230115436	48.8217	11	48,621	7 9426771	2 40.965	4 1752234E	76,3451	Z Z3U1154



_	2 Foreign													4.1	0	
(a)	Individuals (Non-Resident Individuals / Foreign Individuals)	0	0	0	0 0	0.0000	0	0	0 0.0000	0	0.0000	6	0.0000	۰	0.0000	0
(b)	Government	0	0	0	0 0	0.0000	0	0	0 0.0000	0	0.0000		0.0000	0	0.0000	0
(c)	Institutions	0	0	0	0 0	0.0000	0	0	0 0.0000		0.0000	e e	0.0000	el.	0.0000	0
(d)	Foreign Portfolio Investor	0	0	0	0 0	0.0000	0	0	0 0.0000	n	0.0000	- 4	0.0000	- 6	0.0000	- 0
(e)	Any Other (Specify)	0	0	0	0 0	0.0000		D)	0 0.0000	0	0.0000	- 1	0.0000	- 3	0.0000	- 0
	Sub Total (A)(2)	0	0	0	0 0	0.0000		6	0 0.0000	0	0.0000		0.0000	- 2	0.0000	0
	Total Shareholding Of Promoter And Promoter Group (A)= (A)(1)+(A)(2)	26	230115436	a	0 230115436	200,000	230115436	0 2301		0	48.8217	94267712	40.9654	175221681	76.1459	2301.15436

Details of Shares which remain unclaimed may be given hear along with details such as number of shareholders, outstanding shares held in demat/unclaimed suspense account, voting rights which are frozen etc.

Note:

- (1) PAN would not be displayed on website of Stock Eschange(c)
 (2) The term 'Encumbrance' has the same meaning as assigned under regulation 2R(1) of SEBI (Substantial Acquisition of Shares and Takeovers) Regulations, 2011.



	Category & Name of the shareholders	PAN	Nos. of sharehold era	No. of fully peld up equity shares held	equity shares	No. of shares underlyin E	Total nas. shares held	Shareholding % calculated as per SCRR, 1957 As a %	Number of Vot	ilng Algh	ts held in each cla	ess of securities	Shores	, as a % assuming full	Number of to shares	dod in	Number of pledged or excumbers	otherwise	Number of equity shares held in dematerialise
					held	Depositor y Receipts		of (A+B+C2)	No of Voting R	ights		Total as a % of [A+8+C]	convertible securities (including	of convertible securities (as a percentage		As a % of total Shares held(b)	r4o. (a)	As a % of total Shares held(b)	d form
									Class og X	Class eg: y	Total		Warrants)	of diluted share capital		5555			
	60	04	010	avi	M	CAIL	(vii) = (vii) =	(MII) As a % of (A+B+C2)		72	(IX)		(XI	(XI)= (VII)+(X) As a % of (A+9+C2)	(xe	ıı		pany	(xiv)
	1 Institutions									1									
(1)	Mutual Fund		11	11054114		0 0	11054134				0 1105413			2.3453		0.000	The state of the s	NA	11054134
(0)	Venture Capital Funds		0	0		0 0	0	0,0000		0	0 1	0.0000		0.0000		0.000	NA	MA	- 0
(6)	Albernate Investment Funds		- 0		1	0 0		0.0000		0	0 (0.0000	N 3	0.0000		0.000	NA .	MA	
241	Facelina bissanian Carlest ku antaun																		
(d) (e)	Foreign Vernare Capital Investors Foreign Portfolio Investor	-	92	0		0 0	0			0	D (The second second second						MA	
(6)	Percept Personal Investor		90	86165110	4	0 0	36185110	18.2809	86165116	0	0 86165110	18,2809		18.2809		0.000	NA	NA.	86165110
	Arisaig Partners (Asia) Pto Ltd. A/C Arisaig India Fund Limited			31687118		0 0	31637118	6,7122	31637118	8	0 3163711	8 6.7122		6.7122		0.000	NA.	MA	31637118
-	Lancon Construction of the Construction of		1	- Same			C-2000		1000000	1	W. Transaction			avenue.		les Cuerce	No.	1000	05000000
	Government Penulon Fund Global			9372942		0 0	Annual Colored Publishers (A)											MA	9572942
	Verlinaest Sa		_	10720717	18 Q Q 31637318 5.7172 33637318 Q 31637138 6.7122 Q 6.7122 Q 6.8000 NA 42 Q Q Q 31637318 5.7172 33637318 Q 31637138 6.7122 Q 6.7122 Q 6.8000 NA 42 Q Q Q Q Q Q Q Q Q Q Q Q Q Q Q Q Q Q		MA	10780712											
-	Wgi Emerging Markets Smaller Companies Fund, Uc			10755879		0 0	The second secon	The second designation of the latest designa										NA	10755870
(n	Financial institutions / Banks	-	5	5068413		a 0	5068413	1.0758	5068413	3	0 506841	1.0751	()	1.0753	1	0.000	NA.	NA	5068413
100	Ute Insurance Corporation Of			1 33.000	1			1977		1 3		16334		7533		12333	330	3 6.6	TO THE STATE OF
(-)	rdie		-	4778281								411.00						NA	4778251
un_	Insurance Companies		1	2639583		0 0	2639583	0.5600	263958	1	0 263358	0.5600	1	0.5600		0.000	O NA	NA	2639563
DA.	Provident Funds/ Panalon Funds							0.0000											
10	Any Other (Spedily)		1 8	9		0 0		0.0000		D	0 1	0.0000		0.0000		0.000		NA.	- "
	Sub Total (e)(1)		112			0 0			104927240		0 10492724	The second secon		22.2615		0.000		NA.	104927240
	Central Government/ State Government(s)/ President of										U DADETE					0.000	, max	mr.	10432724
	Z India Sub Total (8)(2)		9	0		0 0						0.0000		0.0000				1	1
	3 Non-institutions		-					0.0000	-	0	0 (0.0000	-	0.0000	-	0,000	MA	NA:	- 0
(a)	Individuals	_	+				_			+	-	0.0000		0.0000			PAA.	NA.	
	Individual shareholders holding nominal share capital up to Rs. 2 lakbs.		41703	12045429			11045429							12 = 1				1	
	R. Individual shareholders holding nominal share capital in excess of Rs. 2 lakins		1700	7313087					Weston.		0 11045429			2.3414		0.000		NA.	9905740
(6)	NBFCs registered with Hel		0	7213087		0 0	7213087		7213087		721300			1.5303		0.000		MA	7213083
003	Employee Trusts		0	0		0 0	0	0.0000		-	0 (0.0000		0.0000		0.000		MA	-
200	Overseas Depositories(holding		1			0	0	0.0000			0 (0.0000		0.0000		0.000	UNA	NA	
(d)	DRej (hatancing figure)	I Company	0			0		0.0000			0	0.0000		0.0000		0.000	aMI	MA	i la maria de la
(e)	Any Other (Specify)	6.7	2184	118037365		0 0	118037365				0 113057365			25.0430					0 11805142
2000	Trusts	S. T.	3	95561		0 0	95561	0.0203			9556			0.0200	A THREE PROPERTY AND ADDRESS OF THE PARTY.	0.000	A Rose Comment	NA	9556
	Hindu UndWided Family	0	814			0 0	2361451	0.5010			236145			0.5010		0.000		NA NA	236145

Table III - Statement showing shareholding pattern of the Public shareholder

Category & Name of the ahandrolders	PAN	Nos. of sharehold ers	Part in the contract of the	aquity drares	No. of shares underlyin &	Tatal nos. shares held	Shareholding % calculated as per SCRR, 1957 As a %	Number of Vet	ing Righ	ts held in each cla	ess of securities	Shares Underlying	Shareholding , as a % assuming full conversion	Number of Loc shares	hed in	Mumber of Si pledged or o snoumbered	therwise	Mumber of equity sharm held in dematerialise
				held	Ospasitor y Receipts		of (A+0+C2)	No of Vetting R	ights		Total as a % of (A+B+C)	convertible recurities (including		1	As a % of total Shares held[b]	Na. (»)	As a % of total Shares held(b)	d form
								Class eg: X	Class eg: y	Total		Warrents)	of diluted share capital)		пенци		metajaj	
(0)	00	(11)	avj	(4)	(vi)	(ni)+(ni) (nii) -	(VIII) As a % of (A+8+C2)			¢ox]		(x)	(XI)- (VII)-(X) As a % of .4A+B+C2)	(Mari	0	į ist	10	[kav]
Non Resident Indians (Non Report)		384	133900			133908	0.0284	133500		133906	0.0284		0.0284		0,0000		NA.	13190
Non Resident Indians (Regat)		297	259061		0 0		0.0548	258061		258081	Or and the second		0.0548		_		PAA	25588
Clearing Member		266	485683	Z 84	0		0.1030	48568		485685			0.1010		0.0900	WWW.	NA.	48568
Bodies Corporate		620		- 4	0 0	-	24,3355	114702681		114702681			24.3355	43478261	100000		NA	11469894
Cedar Support Services Limited			43478261			45478261	9,2344	4147926		43478261	9,1244		9,2244	40478201	100,0000	KA.	NA.	48478255
Srand Equity Treaties Umited			24999999		0	24999999	5,3040	24999999	9	24999999			5 3040	The second secon	0.0000	1110	NA.	2499939
Bennett, Coleman And Company Umited			20391700			20391700	4,9269			20901/00			4 3263		0.0000		1855	Comment.
Courseled Mercantile Private similed			9560574	7 5%		9560574	2,0384	9560574		2560574			2 0284		980000	00	NA.	20391700
Sub Total (B)(3)		45920			0	136295881	28.9168	136255881	_	136295601			28.9168		31,8999		NA.	9560574
Total Public Shareholding (3)= (8(£13+(8)(2)+(8)(3)		44032	Section 1		0	241225121	51.1783	241223121		241223121	51.1783		51.1781	100000		000	NA NA	240077496

Details of the shareholders acting as persons is Concert including their Shareholding (No. and Sc):

Details of Shares which remain unclaimed may be given hear along with details such as number of shareholders, outstanding shares held in demat/unclaimed suspense account, writing eights which are frozen etc.

No. of shareholders	(3)	No. of Shares
	185	301260

Note:

(1) FAN would not be displayed on website of Stock Exchange(s).

(2) The above format needs to be disclosed along with the name of following personal estitutions/Non institutions holding more than 1% of total number of shares

(3) W.r.t. the information pertaining to Depository Receipts, the same may be disclosed in the respective columns to the extent information available and the balance to be disclosed as held by custodian.



FUTURE RETAIL

Shareholding Pattern under Regulation 31 of SEBI (Listing Obligations and Disclosure Requirements) Regulations, 2015

2	Scrip C	ode/Name of Scrip/Class of Security: 540064 / FRETAIL / Equity Shares
3	Share	Holding Pattern Filed under: Reg. 31(1)(a)/Reg. 31(1)(b)/Reg.31(1)(c): - Post Scheme of Arrangement (based on September 30, 2016)
	a.	If under 31(1)(b) then indicate the report for Quarter ending
. 3	b.	If under 31(1)(c) then indicate date of allotment/extinguishment:

Particulars	Yes*	No*
1 Whether the Listed Entity has issued any partly paid up shares?		No
2 Whether the Listed Entity has issued any Convertible Securities or Warrants?		No
3 Whether the Listed Entity has any shares against which depository receipts are Issued?	The second second	No
4 Whether the Listed Entity has any shares in locked-in?	Yes	1.00
5 Whether any shares held by promoters are pledge or otherwise encumbered?	Yes	

* If the Listed Entity selects the option 'No' for the questions above, the columns for the partly paid up shares, Outstanding Convertible Securities/Warrants, depository receipts, locked-in shares, No of shares pledged or otherwise encumbered by promoters, as applicable, shall not be displayed at the time of dissemination on the Stock Exchange website. Also wherever there is 'No' declared by Listed Entity in above table the values will be considered as 'Zero' by default on submission of the format of holding of specified securities.

Note:

In the above paid up capital, 1,542 Optionally Convertible Debentures (OCDs) of Rs. 10 lakh each, issued pursuant to the Composite Scheme of Arrangement between Future Enterprises Limited (earlier known as Future Retail Limited) with the Company, becoming effective on May 1, 2016, have not been considered as the conversion price of these securities would be determinable at any time during the tenure of the OCDs or latest by October 31, 2017, when the option for conversion is exercised. Further, as per terms of isssue, these securities may be redeemed in case conversion option is not exercised.



Table I - Summary Statement holding of specified securities

Catagory	Category of shareholder	of	No. of fully paid up equity shares held	No. of Partly paid-up equity		Total nes. shares held	Shareholding as a % of total no. of shares (calculated as		Vating R	ghts held in ea	ch class of	No. of Shares Underlying Outstanding	Shareholding , as a % assuming full conversion of	Number of Lo shares	icked in			But a company of the
		200-		shares held	Depositor y Receipts		per SCRR, 1957)	No of Votin	g Rights		Total as a % of (A+B+C)	convertible securities (including	convertible securities (as a percentage	No. (a)	As a % of total Shares	No. (a)	As a % of total Shares	d form
			9 6 6				0-11	Class eg: X	Class eg:y	Total		Werrants)	of diluted share capital)		held(b)		held(b)	
(1)	(10)	(н)	(1V)	(v)	(M)	(N)+[N]+(N0) (NI) =	(VIII)As a % of (A+B+C2)			(DC)		(x)	(XI)= (VII)+(X) As a % of (A+B+CZ)	(x	a)	Dai	9	(XIV)
(A)	Promoter & Promoter Group	26	230115436		0 0	230115436	47.0405	230115438	5	230115436	47.0405	0	47.0405	9426771	2 40.9654	175223483	76.1459	230115436
(8)	Public	44033	259070541		0	259070541			_	259070541	THE RESERVE AND ADDRESS OF THE PERSON NAMED IN COLUMN TWO IN COLUMN TO THE PERSON NAMED IN COLUMN TWO IN COLUMN TW		52,9595	4347826			NA	257924916
(c)	Non Promoter - Non Public					_000000000	(C. 1999)	0.001222					75.00		-			
(C1)	Shares Underlying DRs	0	0		9	0	0.0000)	0	0.0000	0	0.0000		0.0000		NA.	0
(C2)	Shares Held By Employee Trust	0	Û	- (0	0	0.0000	- (0 0	0.0000	0	0.0000		0.0000		NA	0
120	Total	44059	489185977		0	489185977	100,0000	489185977	1	489185977	100,0000	0	100,0000	13774597	28.1582	175223483	35.8194	488040352



-	Category & Name of the shareholders	PAN	Nos. of sharehold era	No. of felly paid up equity shares held	Partly paid up equity shares held	shares underlying Dapositor		Shareholdi ng % calculated as per	Number of t securities	/oting #	ights held in i	each class of	No. of Shares Underlyin 8	Shareholdi ng , ax a % assuming full	ahares	cked in	Number of 5h pledged or of encumbered		Number of equity shares held in dematerialise
						y Receipts		SCRR, 1957 As a % of (A+8+C2)	No of Vesing	Sights		Total sca % of (A+B+C)	Outstandi rig convertibi	conversion of convertible securities (As a % of total Shares	No. (a)	As a % of total Shares	d form
									Class eg: X	Class eg y	Total		securities (including Warrants)	ax a		held b)		held(b)	
	0)	00	(m)	(IV)	80	(vi)	[vii] =	(VIII) As a % of (A+B+C2)			(04)		00	(XII)= (XII)+(X) As a % of (A+B+C2)	(%)	0	(OII	u u	(KIV)
	1 Indian															T			
	Individuals / Hindu Undivided				100		Lycon					New York		0.510		Contract		2.000	
[0]	Family			88135	-		88115			- 0	88115	0.0130		0.0180		0.0000		0.0000	8811
	Anil Biyeni			2121			2121	0.0000000000000000000000000000000000000			2121	0.0004	. 0	0.0004		0.0000		0,0000	212
	Ashni Kishore Biyani			71147	- 0		71147	0.0145	-				. 0			0.0000	1	0.0000	7114
-	Copikishen Byani		-	7121			2121				2121	0.0004			3	0.0000			
	Kishare Siyani			2121		-	2121				2121	The second second second			1	0.0000			
	Laterrinarayan Biyani Dahash Shasal		-	2121		-	2121	0.0004		- (0			0.0000			
_	Rakesh Biyani Sunil Riyani		-	2121	- 0	0	2121			- (0,0004	- 0	-	1	0,0000			
_	Vipry Diyani		-	7121	- 0	- 0	2121	The second second second second	The second second second second	_ (P. P	0.0004			9	0.0000			
	Vivok Bisani			2121		- 0	2121	0.0004		- 1		0.0004		0.00000	1	0.0000			
	Central Government / State			8342	- 0	- 0	2121	0.0004	2121	- 6	2171	0.0004	0	0.0004	- 1	0.0000	4 4	0.0000	212
b)	Government(s)							0.0000				0.0000		0,0000					
r)	Financial Institutions / Banks			0	0			0.0000		-	0			0.0000		0.0000		21222	
d)	Any Other (Specify)		12	210027321	0	0	230027321		230027321	_	230027321	47.0225			9426771	to the same of a December 2 and	The second second second		
	Bodies Corporate		12	The second secon	0	0		47.0225	BOTH THE PROPERTY OF THE PARTY	tion to the last	25002/521	47.0225			94267712			AND DESCRIPTION OF THE PERSON	The second second second
	Puture Corporate Resources Umited			177545090	0		177545000	36,2940	177545090		177545090	36.2940		36,2940	80602685	251100120		1000	-
	PIL industries (imited			441.16090	0	0	44136090	9.0234	44138000		64136030	9.0224	0		8730360	THE RESIDENCE OF THE PARTY OF T	The second little of the second second		-
	Gargi Business Ventures Private Limited		,	4550000	0		4550000	0.9301			4550000		0	0.9901	\$250000				
	Ryka Commercial Ventures					1		-								72			
	Private Umited		1	1634663	. 0	-	1684653	0.3444	1634663	. 0	1684663	0.3444	0	0.3444	1684663	500,0000		0.0000	368466
_	Marx Ratail Private limited	110	- 2	1579108	0	0	1579108	0.3228	1579108	0	1570103	0.7228	- 0	0.3228	1	0.0000	200000	12,6654	157910
	Future Capital Investment Private Unite		1	531375	0	a	531375	0.1086	531375	0	\$31375	0.1086	U	0.1086	3	0.0000		0.0000	53137
	Akar Estate And Finance Private Lamited		3	1000	1 84	1 19	320	12.225		39	1 1	12387	1 88		1 6	17.33		50.0	1.8
_	Sub Total (A)(1)	-	26	230115436	0		230115436	0.0002 47.0405	1000 230115436	- 0	280315436	47.0405	0	0.0002 47.0405	94267712	40.9654		76.1459	



	2 Foreign						7									
(a)	individuals (Non-Resident Individuals / Foreign Individuals)		0	0	0 0	0.0000	0	0 0	0.0000	0	0.0000	o	0.0000	0	0.0000	
(6)	Government	0	0	0	0 0	0.0000		0 0	0.0000	0	0.0000	0	0.0000	0	0.0000	
(c)	Institutions	. 0	0	0	0 0	0.0000	0	0 0	0.0000	a	0.0000	al.	0.0000	0	0.0000	
(d)	Foreign Portfolio Investor	0	0	0	0 0	0.0000	0	0 0	0.0000	0	0.0000	g g	0.0000	0	0.0000	
(e)	Any Other (Specify)	0	0	0	0 0	0.0000		0 0	0.0000	0	0.6000	0	0.0000	0	0.0000	
	Sub Total (A)(2)	0	0	0	0 0	0.0000		0 0	0.0000	0	0.0000		0.0000		0.0000	
	Total Shareholding Of Promoter And Promoter Group A = A (1)+(A (2)	26	230115436		0 230115436	47,0405	230115438	D 230115436	47.0405	0	47,0405	94267712	40.9654	175223483	76.1459	230115486

Dutails of Shares which remain unclaimed may be given hear along with details such as number of shareholders, outstanding shares held in demat/unclaimed suspense account, voting rights which are freeze etc.

Note:

- (1) PAN would not be displayed on website of Stock Exchange(s)
- (2) The term 'Encumbrance' has the same meaning as assigned under regulation 28(3) of SEBI (Substantial Acquisition of Shares and Takeovers) Regulations, 2011.



	Category & Name of the shareholders	PAN	Nos. of sharehold ers	No. of fully paid up equity shares held	Partly paid-up equity	No. of shares underlyin	Total nos. shares held	Shareholding % calculated as per SCRR, 1957 As a % of		ing Night	s held in each	class of securities	No. of Shares Underlying Outstandin	, as a % assuming full	Number of Loc shares	ked in	Number o pledged o ancumber	rotherwise	Number of oquity shares hold in dematerialise
					shares held	Depositor y Receipts		(A=B+C2)	No of Voting R	ights		Total as a % of (A+R+C)	6 convertible securities (including	af	No. (a)	As a % of total Shares held(b)	Mp. (n)	As a % or total Shares held(b)	d form
									Class og: X	Chos eg: y	Total		Warments)	of diluted share cogital)				Population	
	01	101	(10)	(M)	(v)	(91)	(M)+(A)+ (A)	(VIII) As a % of (A=B+C2)		-	(08)	1	DO	(H)- (H)+(X) As a % of (A+R+C)	ÇII	•		(MIII)	(Nov)
	1 Institutions													0				73	
(a)	Mutual Fund		1			0 0				_	0 11054			0 2.2590				NA NA	1105413
(b)	Venture Capital Funds		15 - 2	0 (1	0 0				_	01	0.000		0.0000		-		NA.	
[0]	Alberrate Investment Funds	-	1	0 0	-	0 0		0.0000		0	0	0 0.000	-	0.0000	1	0.000	100		1
180	Foreign Venture Capital		100					0.0000		0	0	0.000	0	0,0000		0.000	MA	MA	1 4
100	Investore		9	2 86165110	-	a .	86165110			0	0 86165	1.00		0 17.6144		0.000		NA	8816913
(e)	Foreign Portfolio Investor Aricale Partners (Asta) Pse Ltd.		-	Z selbolis		1	- Alleria	1,,,,,											***************************************
	A/C Arising India Fund Limited			31687116		0 1	31637116	8,467	3168711	а	0 31637	138 6.46	73	6.467	-	0.000	NA.	NA	\$169711
	Covernment Persion Fund Global			587234		0 1	9572940	1,950	987294	12	U 9572	942 1.956	79	0 1.9563		0,000	NA	NA	957294
	Variance: Sa	17.	-	1073071		0 0	10730713	2.193	1073071	2	0 10790	712 2.19	96	0 2.1930		0.000	NA	NA.	1073071
	Wigi Drowiging Markets Smeller		-	1 6000			- 33.53			1 3		30	179	3.373		10000		1860	333.11
	Companies Fund, Lic.			10799879	4	0 0	1,0755673				0 30755	The state of the s		D 2198		0.000	4.7	NA.	1075587
60.	Financial Institutions / Banks	E	4 300 19	506841		0 (5068411	1.016	506841	3	0 5068	413 1.030	53	0 1.036	9	0.0000	NA.	NA.	506841
	Life Insurance Corporation Of							1000						0.11			1	-	477828
	freis	2		477926	-	0 0	477526				0 4771	Colonia Coloni		0.976		0.000		NA NA	263958
40	Incurance Companies			4 263358		0 0	261958	0.535	6 263951	13	0 2635	583 0.53	14	0.539	1	0.000	il ron.	POPL.	203000
										J	-	0 0.000		0.000		D.000	n Pua	na.	3
(b)	Provident funds/ Pension funds		-	0	0	0 0		0.000		0	u u	0.00		0 0.000				NA.	
10	Any Other (Specify)		-	2 10092724		0 0				0	a 104923			0 21.449		0.000		MA.	10492724
-	Sub Total (0)(1) Central Government/ State		11	2 10452224		*	13452.724	22.40	100000		10.00							-	
	Government(s)/ President of		1 3			0 /		0.000	0	1		0.00	10	0.0000	0				3 - 2
	Z India Sub Total (8)(2)	1	1 5	0	0	0 0		0.000	-	0	0	0 0,000		0 0.0000	0 0	0.000	I NA	MA	0
	3 Non-institutions		1	-				1											
felt	Individuals											0.09	10	0.000	0	0	MA	NA.	100
	Individual shareholders holding cominal share capital up to Rs. 2		4170	1104542			1104542	2.257	9 3354640	9	D 1104	429 2.25	79	g 2,257		0.000	0 46	944	990574
	initis. ii. Individual shareholders	1	4270	1354045	1	1	7,110,112	2.20	222.000	1	7.00						1		
	holding numinal share capital in success of Rs. 2 labbs.		,	721308	,		721306	7 5.474	5 72130	17	D 721			0 1,474				NA	721306
(6)	NBFCs registered with RBI			a	0	0 1	D	0.000	0	a	0	0 0.00		0.000	0 1	0.000		MA	
(c)	Ereployee Trusts			0	-	0	D	0.000	0	0	0	0 0.00	00	0.000	9	0.000	D NA	NA	
(d)	Overseas Depositionles/holding Ofts) (balancing figure)			a	6	0	0	0.000		0	0	0 0.00		0.000	Commence of the Control of the Contr	0.000		NA.	
60	Any Other (Specify)		218			0	13588478				0 13568			0 27.377				0	0 13587884 9556
900	Trusts			3 9556		0	0 5556					SET DOD		0 0.019		0.000		NA NA	236145
	Hindu Undhelded Family		31	236145	1	0	0 236145	0.482	7 23614	51	0 236	1451 0.48	"	0 0.482	1	0.000		-235	230145
	Non-Regident Indians (Non- Report)		1	11190	1		0 13390	0.027	1 1319			908 0.02	10	0 0.027		0.000	olms /	NA.	19990

Table III - Statement showing shareholding pattern of the Public shareholder

Ortagory K Name of the shareholders	PAN	Nos. of sharehold ere	paid up equity skares held	Fartly gold-up equity shares	No. of shares undertyin g	Total nos. shares held	Shareholding 'A calculated as per SCRR, 1957 As a % or	Number of Ve	ting Right	s held In each cla	ns of securities	No. of Shares Underlying Outstandin	, as a Si	Number of Loc shares	ked in	Number of St pledged or or encumbered	250 P. P. CO.	Number of equity shares held in deceateristics
				held	Depositor y Receipts		(A+B+C2)	No of Voting R	lghts		Total as a % of (A+R+C)	g convenible securities (including	50 22 12 12 12 12 1 1 1 1 1 1 1 1 1 1 1 1		As a 15 of soral Shares	No. (a)	As a % of total Shares held(b)	The state of the s
								Class eg: X	Class ag: y	Yotal		Warrenta	of diluted share capital)		heldft)			
m m	øij	(44)	(IV)	(M)	(VI)	Sol-ton toll (AH =	(V110) As a % of (A+8+C2)			lod		(XI)	(MI)= (MI)=(X) As a % of (A+R+C2)	(XI	•	00	115	(MV)
Non Resident Indians (Report)	-	297		- (0	2.200	2,42.44	25808	_	258081			0.0528	0	0.0000	NA.	No.	255883
Gearing Monher	-	266		-	0	485683		48568		485683	0.0555		0.0911	- 0	0.0000	NA	NA.	485583
Bodies Corporate	_	621	132550101		0	132550101	27.09GL	13255010	£ (132550101	27.0961		27.0961	45478262	32,6014	NA.	NA.	132546360
Codar Support Services Limited			43478261		0	49478261	8.8879	4347826	1 0	43478261	8.8875		1.0329	43478361	100.0000	NA .	NA	43978255
Brand Equity Treation Umited	_		24950909		- 0	24999996	5.1105	2499999	9 (24999999	5,1105		5 1105	0	8,0000	NA.	NA.	24999999
Sennett, Coleman And Company United			20391700			20091700	4.1685	2039170	0 0	20391700	4.1685		4.1635		0,0000	NA	No.	20391700
Heritage Foods Limited (*)			17847410			17847420	5,6484	1794742	0 0	1/84/420	3,6484		3 6484	i i	0,0000	No.	NA.	17847420
Counciled Mercantile Private Limited		4	9591574			9560574	1,5544	956057		9560574	1.9544			- 3	Service of the	Jan.	Visco in	
Sub Total (9)(3)		43921	154143301		0	354143301	31,5102			154143301	81.5102	- 3	1.9544	4747676	0,0000	1000	No.	9560574
Total Public Stareholding (8)- (0][1]+[8](2)+(0][3]		44031		0	0	259070541	52.9595	25907054		259070541	53.9545		52,9695	43478261	28.2064		NA NA	257924516

Details of the shareholders acting as persons in Concert Including their Shareholding (No. and N):

Details of Shares which remain unclaimed may be given bear along with details such as number of shareholders, outstanding shares held in demat/unclaimed suggests account, voting rights which are frozen etc.

No. of shareholders		No. of Shares
	185	101290

Note:

(1) PAN would not be displayed on website of Stock Exchange(s).

(2) The above former accels to be duclosed along with the name of following persons institutions/Non inetitutions holding more than 1% of total number of shows

(3) W.r.t. the information pertaining to Depository Receipts, the same may be disclosed in the respective columns to the extent information available and the habance to be disclosed as held by contention.

(*) To be issued by FRL pursuant to the Scheme of Arrangement



Table IV - Statement showing shareholding pattern of the Non Promoter- Non Public shareholder

	Category & Name of the PAN shareholders	PAN	Nos. of sharehold ers	No. of fully paid up equity shares held	Partly paid-up equity shares	No. of shares underlying Depository Repripts	Total nos. shares held	ng % calculated		r of Vot securiti		held in each	No. of Shares Underlyin	ng , as a % assuming	Number of L shares	ocked in	Number o pledged o encumber	otherwise	Number of equity share: held in
					held	Jacoba		as per SCRR, 1957 As a % of (A+B+C2)	No of 1	oting R	givta	Total as a % of (A+B+C)	ng	of convertibil	No. (a)	As a % of total Shares held(b)	No. (s)	As a % of total Shares held(b)	dematerialis d form
									Class eg: X	Clarac eg: y	Total		100000000000000000000000000000000000000	e securities (as a percentage of diluted share capital)				held(b)	
	m	011	Quq	(IV)	(v)	(M)	(M)+(n)+ (n) (M) =	(VIII) As a % of (A+B+C2)			(m)		(N)	(XI)= (VII)+(X) As a % of (A+B+CZ)	,	×8)		Çunü	(NIV)
1	t. Custodian/Dit Holder			0		0 6		0.0000			0	0 0.000	0 0	0.0000		0 0.000	NA.	NA.	
,	Employee Renefit Trust (under SEBI (Share based Employee Benefit) Regulations, 2014)			0		0 0		0.0000				0.000	0 0	0.0000		0 0.000) NA	NA.	
	Total Non-Promoter- Non Public Shareholding (C)= (C)(1)+(C)(2)					0		0.0000				0 0.000		0.0000		0 0.000	No.	NA.	

Note:

(1) PAN would not be displayed on website of Stock Eschange(s).

(2) The above format needs to disclose name of all holders holding more than 1% of total number of shares
(3) W.r.t. the Information pertaining to Depository Receipts, the same may be disclosed in the respective columns to the extent information available





The financial details and capital evolution of the transferee/resulting and transferor/demerged companies for the previous 3 years as per the audited statement of Accounts:

Name of the Company: HERITAGE FOODS LIMITED

(Rs. in Crores except EPS & Book Value)

	(Unaudited)	As per last Audited Financial Year	1 year prior to the last Audited Financial Year	2 years prior to the last Audited Financial Year
	As on 30th September 2016	2015-16	2014-15	2013-14
Equity Paid up Capital	23.20	23.20	23.20	23.20
Reserves and surplus	248.98	216.79	169.81	155.79
Carry forward losses	0	0	0	0
Net Worth	272.18	239.99	193.01	178.99
Miscellaneous Expenditure	0	0	0	0
Secured Loans	112.22	124.56	157.40	141.54
Unsecured Loans	1.31	1.47	0.00	0.00
Fixed Assets (Net)	351.52	321.61	289.65	282.60
Income from Operations	1275.54	2380.58	2072,97	1722.04
Total Income	1277,68	2387.13	2079.99	1726.99
Total Expenditure	1230.44	2301.15	2040.85	1665.78
Profit before Tax	47.25	85,98	39.14	61.21
Profit after Tax	32.22	55.43	28.21	45.31
Cash profit	50.69	89.94	62.20	70.31
EPS	13.89	23,89	12.16	19.53
Book value	117.32	103.45	83.20	77.16

For HERITAGE FOODS LIMITED

UMAKANTA BARIK Company Secretary M. No. FCS-6317







(Formerly known as M/s. Heritage Foods (India) Limited)
CIN: L15209TG1992PLC014332
AN ISO: 22000 CERTIFIED COMPANY





Heritage Foods Retail Limited

Name of the Company: HERITAGE FOODS RETAIL LIMITED

(Rs. in Crores except EPS & Book Value)

	Latest Financials (Audited)	As per last Audited Financial Year	l year prior to the last Audited Financial Year	2 years prior to the last Audited Financial Year
	As on 30 th September 2016	2015-16	2014-15	2013-14
Equity Paid up Capital	0.1656	0.1656	0,1656	0.1656
Reserves and surplus	(0.1409)	(0.1409)	(0.1385)	(0.1309)
Carry forward losses	0	0	0	.0
Net Worth	0.0247	0.0247	0.0271	0.0347
Miscellaneous Expenditure	0.	0	0	.(
Secured Loans	0	0	0	.0
Unsecured Loans	0	0	0	.0
Fixed Assets	0	0	0	
Income from Operations	0	0	.0	0
Total Income	0	0	0	
Total Expenditure	0	0.0024	0.0076	0.0028
Profit before Tax	0	(0.0024)	(0.0076)	(0.0028
Profit after Tax	0	(0.0024)	(0.0076)	(0.0028)
Cash profit	0	(0.0024)	(0.0076)	(0.0028
EPS	0	(1.45)	(4.59)	(1.69
Book value	1.49	1.49	1.64	2.10

For HERITAGE FOODS RETAIL LIMITED

Authorized Signatory



Regd. Office: #6-3-541/C, Panjagutta, Hyderabad - 500082

The financial details of the resulting company for the previous 3 years as per the audited statement of Accounts:

Name of the Company: Future Retail Limited

(Rs. in Crores except EPS and Book Value)

Particulars	As per Un- audited Financial For the Quarter ended	As per last Audited Financial Year	1 year prior to the last Audited Financial Year	2 years prior to the last Audited Financial Year
	30th June, 2016	2015-16	2014-15	2013-14
Equity Paid up Capital	94.27	94.27*	1,398.66	1,398.66
Reserves and surplus	2,270.58	2,270.58	396.00	396.00
Carry forward losses	(420.35)	(490.66)	(2,216.26)	(1,837.05)
Net Worth	1,548.50	1,478.19	(817.60)	(438.39)
Miscellaneous Expenditure	-	-	-	
Secured Loans	1,427.46	918.20	294.00	
Unsecured Loans	68.00	50.00	96.40	183.00
Fixed Assets	277.20	260.93	177.94	190.20
Income from Operations	3,979.68	6,844.96	1,779.42	2,343.12
Total Income	3,985.22	6,860.47	1,849.45	2,748.45
Total Expenditure	3,914.67	6,845.92	2,228.66	3,063.06
Profit before Tax	70.55	14.55	(379.21)	(314.61)
Profit after Tax	70.55	14.55	(379.21)	(314.61)
Cash profit / (Loss)	77.42	51.31	(335.11)	(253.90)
EPS	1.50	0.66	(2.71)	(5.42)
Book value	32.85	31.36	(5.85)	(3.13)

^{*}Includes Share Capital Suspense aggregating to INR 85.57 Crores

For Future Retail Limited

Virendra Samani

Dy. Company Secretary ACS # 12971





Partners inners as HERTAGE FOCKS (INDEX), LIMITED)
Rept. Office: 5 - 3 - 541/4, Auf. to Wildl., Punjagotta, Hydroriani - 588 662.
CIN: 125.00016196191656012 - www.hertners.h.m.

	-		EDAREN	ALDRE					10910	MEANTE		
Perfection		Quarter Ended		Hell Yes	or Enoled	Year Ended		Queter Ended			ar Carbod	The Uniet
- Constant	30.09.2014	30:05.2016	30.00,3815	30.09,2016	26.09.2015	33.23.2016	30.09.2018	90.06.2918	30.09.3015	30.09.3018	80.08.(015	81.03.201A
	(Livearited)	[Unsudited]	(insuffed)	[Draudbed]	(Unsettied)	(kerting)	[ireofine]	Drawfield	(Monthell	(Unsentited)	(Lessonited)	(furthed)
1. Secrete from specifican	200	2000						40985.08	12500.00	139522.06	124793.65	111015.0
110 Net Sales / Number Book Operations Clint of number duty!	\$2876.86 1854.08	01545.18 1877.74				232899-32	62570.6E	3877.76	1000.29	2001.00	2129.38	1163.0
80 Other Operating Switze		the second secon	1980.29	2991.416	2129:18	The second secon		63422.90	18679.54	177553.33		
Total income from operations (ark)	64130.96	63422.90	58878.94	127553.88	19821.13	20000-00	64130.96	1945531	189.00.39	711,300,00	338523,25	2,98058,33
I. Expresditure							33847.61	15296.11	\$1869.27	67143.72	86501.17	141176.70
(g) Cost of purcerials research	33847,61	310000000000000000000000000000000000000	7,700,000			341376,79					23064.69	49296.64
All Providence of March on Greater	\$4180.36	20071100	1000000	25344.10		49296.66		14165.74	1,1998,49			
to Changes in revolution of facilitied goods work to progress and stock in trade	3064.34	1893,50	2562.36	4075-20	2160.66	(2948.45)	3384,24	3991,94	2061.16	407£38	2160.60	(2008.AS
All Emphases benefits represe	4152.75	4028.43	3571.50	8193.68	8678.28	13,999,11	4152.25	4026.43	3501.50	8180.68	6678.26	13999.11
tit Depostation and uncertaining represer	943.42	902,00	865.71	1985.41	1007.00	3451.22	943.42	902.99	861.75	3345.41	1677.88	3451.21
ID Other expresses	4506.26	1 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	5793.33	700 10 00	11506.43	23114.88		6394.35	5795.10		11466.43	29115.80
Total espenditure (arknordnerf)	£1776.14		56024.58			229432.20	81776.14	60777.57	36028.36		111918-12	228452 6
t. Potfit / (Lond Nem operations before other income, finance.						9829.13		1645.25	2850.36	\$507000mmd1	885 (2)	9625.32
code & exceptional Huma (1-8)	229033	2041.00	4400.30									
6. Other Swime	109.32	100000000000000000000000000000000000000		234,58	300.40		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	124.23	130.39		300:48	654.94
i. Profit / (Loss) from criticary artivities before Teasure musta and recognitional Scene (5+4)	246A.15	2769.36	2790.74	5231.75	4905.60	10091-00	2464-15	1709.56	2780.74	5239.71	4905.45	10390.68
5. Tananier coma	245.36	268.60	100.79	309.16	887.00	154130	345.58	363.68	400.79	\$29.10	897.30	1545.36
 Profit Glosel from ordinary unitrities ofter finance mate but before exceptional itsus 6-41 	2218.55	2505.96	2019.65	4724.55	4868.19	6735.30	3336.00	2505.96	2379.70	4724.35	4008.18	8794.75
K. Europtional Demo	1 1	14	(4			136.90	E 34					130.90
b. Profit / (Loss) from neckstary activities before tag (7-8)	2218.59	2505.96	2379.95	4724,55	9966.03	8544.25	2238.50	2505.96	3379.99	H724.55	4558.18	639T-8
16. Tan expense	100000	2000	1000000	3865	10000		0.000	15000	e comm			
Correct Tox 6417	536.00	943.00	905.00	3476.00	1539.00	3099.00	339,00	940.00	905.20	1436.00	1555:00	1986,9
Prior posted time	(107.54)	0.33	[19,140	(337.43)	(38.06)	116.06	007.540	0.11	(19.14)	(107.43)	(38.26)	138.06
Delicord Tax Charge (Cords)	220.30	(86.49)	199-240	18540	(75.86)	[153.34]	220.30	(86.48)	(35.24)	133.02	(79.86)	123.94
\$5.Not Profit (Loss) from ordinary activities after too (\$ 150)	1569.81	1652.33	1579.35	\$222.17	29/01.01	5543.35	1909083	1402.00	1539.33	3222.47	260 L 11	5562.7
U. Entracedinary Rema but of tax exponent	-		-									
III. Not Prodictional for the period before above of profit Abasel of association and foliastic interest (1):430	3549.83	1612.13	1529.00	3323.17	2601.51	SMLIS	1169.01	1652.33	1529.10	3372.17	3901.11	5542.7
16.6harn of profit / thest of associator	1.0						10,00	10.037	19:32	(0.00)	(0.02)	177-06
15. Misserby teternet		Market Li				0.500	1000	1,000	1/1/00/2	1000	12.00	13.04
38. Not Frodi's (Loss) after teams, minurity interest and discretely profit (flood) of essections (13+14+15)	1569.83	1652.11	1575.31	3822.17	2901.11	5343.15	2569.81	1657.32	1529.33	3332.34	2601.29	5542.6
	2315.90	7719.90	23108.00	1313.30	2313.90	2319.90	2319.90	2319.90	2515.90	2519.90	2109.90	2319.9
17. Paid up equity share repital	10.00											10.0
Face value per share (Ra) 1. Reservas excluding reservas as per Balance Blood if previous Assessible year		-		20.00	-	21679.31		1000	18.00	-	-	21670.3
(ii. Earnings per share (iia.) a) UPs believ Extensiblery stems for the period, for the year- te date and limite province one (and associated)					_							
1. Deate	6.27	7.12	6.39	13-89	11.21	23.81	6.77	7.12			13.25	
is. Differed	6.77	7.12	6.31	13.69	11.31	25.03	6.27	7.18	6.70	13.65	11.71	113
to UPS after Extraordinary storms for the power, for the years to date and by the previous year and extraordinals						C (C)()			. 500			
i, Enset	6.77	7.10	6.00	11189	11.71	23.83	6.77	7.52	6.55	13.85	11.21	24.8
s. Dibard	6.77		8.50							13.60		13.8







(Formerly known as M/s, Heritage Foods (India) Limited)





	Standalo	Standalone		
Particulars	September 30, 2016	As at March 31, 2016	September 30, 2016	As a March 81, 201
All the water of the state of t	(Unaudited)	(Audited)	(Unaudited)	(Audited)
A. EQUITY AND LIABILITIES	4.000	THE STREET	- Indiana de la company	
1. SHAREHOLDERS' FUNOS				
a, Share Capital	2319.90	7319.90	2319.90	2319.9
b. Reserves and Surplus	24897.87	21679.38	24888.82	21670.3
Seb-Total Shareholders' funds	27217.77	23999.28	27208.72	23990-21
2. MINORITY INTEREST			1.57	1.53
3. NON-CURRENT LIABILITIES			8938	
a. Long-term bortowings	7229.33	6584.71	7229.93	6584.71
b. Deferred tax Rabilities (Net)	2001.55	1867.72	2001.55	1867.77
c. Other Long term labilities	1152.59	1045.32	1152.59	1545.33
d. Long term provisions	551.53	551.54	551.53	551.54
Sub-Total Non-Current Liabilities	10935.60	10049.29	10935.60	10049.2
4. CURRENT LIABILITIES	2100000	30003.23	10933.00	19049.2
a. Short-term borrowings	2437.28	3925.16	2437.28	3925.18
b. Trade payables	7704.85	8456.93	7704.85	8466.93
c. Other current liabilities #	9218.87	9469.71	9719.36	9470.55
d. Short-term provisions	1171.59	1582.15	1171.59	1982.15
Sub-Total Current Liabilities	20532.59	23443.95	20533.08	23446.81
TOTAL EQUITY AND LIABILITIES	58685.96	57492.52	58678.97	57485.95
I. ASSCTS	- 14		2-1111	1000000
1. NON-CURRENT ASSETS	7-860000	0.988886	500000000	
a. Fixed assets	35152.03	32161.35	35152.03	32161.35
b. Non-current investments	95.66	96.19	79.29	79.85
c. Long term loans and advances	2932.78	2371.52	2932.78	2371.57
d. Other non-current assets	70.10	68.37	70.10	68.37
Sub-Total Non-Current Assets 2. CURRENT ASSETS	38250.57	34697.43	38234.20	34681.09
- TATA W 100 TO 100 AND 4	1000	99950	9999	
a. Current investments	2.15	1.65	2.15	1.60
b. Inventories	11170.30	14491.11	11170.30	14491,11
c. Trade receivables	2908.98	2878.60	2908.98	2978.60
d. Cash and Bank balances	5279.93	4522.71	5287.81	4531.4
n. Short-term loans and advances	867.15	680.88	868.65	580.88
f. Other current assets	206.88	720.14	206.88	220.14
Sub-Total Corrent Assets TOTAL ASSETS	20435.39 58665.96	22795.09	20444.77	22804.86
TUTAL AGOCTO	58685.96	57492.52	58678.97	57485.93

F Standalone: Other current Nabilities includes current maturities of Long-torm borrowings. As at 30.09.2016; Rt. 1686-56 lattin. [As at 31.03.2016; Rt. 2051 98 lability.]







(Formerly known as M/s, Heritage Foods (India) Limited) CIN: L15209TG1992PLC014332







SIZMENT REPORTING FOR THE QUARTER AND HALF YEAR EN	T	-	21590	SHOW					1000	CETAGO		(the in taking
Pertindant		Dearter (Indea)			rtem I	Year Student		Granter Inded			a finalised	Tear Ancied
	30.09.2018	10:06:3010	\$9.69.2013	30.09.3018	90.09-3115	53.65.2010	30.09.2018	\$0.08.2018	30.09.2015	30.09.2010	10.06.2015	81.01.3010
	(Linaudited)	(Unaudited)	Dreaked	(Amenitori)	Mandad	(Audited)	Stranding	[Onestred]	(Charakter)	[Unantited]	(Dated let)	(Auditori)
L Segment Decemen	5 mage				0.000	3.345.55				- 0.00		
a Davy	46200.34	48050.36		92250.70		176880.13	46330.34	46650.36	44187.28			176800.13
b. Retail	17463.00	16081-70		34546.27	27256.16	58288.49	17463.00	16881.28	53962.69		27258.38	58795.47
s. Agri	1279.12	1305.48		2554,60	1791.76	3877.15	1279.33	1305.48	962.11		1795.38	3672.33
d. Sakery	360.06	254.19	\$93.30	614.25	399.04	812.04	360.06	254.15			199.04	812.54
a. Renewable Energy	333-13	111.45	63.32	234.58	121.59	304.30	123.13	131.45	69.13	234.58	329.00	304.30
1. Wet Ca	1365.66	5735.58	1338.88	2701-24	2747.90	5665.57	1305.00	1335.38	1303.38	7105.34	2342.33	3695.57
g. Prentage Foods Setal Lindted	1	1 1	10.	1			7.6		-	100	-	
in, Heritage Corper Ltd						- 2			- 1		4.	
Total Segment Revenue	66791.00	65930.36	60506.07	132729-66	120104-49	269690.68	66790.30	\$5938.88	60509.07	132729.64	100184 49	21000048
	4. PTTS:10000		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	777000000000	100000000000000000000000000000000000000	0.000	1,000,000	77.004.30	11.000000000000000000000000000000000000			
Lane: Heter Segreent Revenue	2660.34	2515.42	1829.13	5575.76	2601.30	7792.35	1960,54	2515.42	1829.13		3663.36	1752.2
Het Sales / Income from Operations	64130.96	EM21.92	38073,34	122355,88		J39018.33	64130,96	19422.53	58678.94	127553.88	7160270	238058.31
		155		- 4	1.0				. *:	7.1		- 1
2. Regiment Results (Frain (+) / (Loss) (-) before finance mats and test)												
a. Daire	3047.62	3391.79	501.32	6436.81	8023.09	51530.19	1047.62	3391.19	3410.52	6438.81	6213.09	\$1530.19
b. Retail	(623.20)	[617.18]		(1340.38)		(3,980,990)	(61).20)	(617.10)			(1993.80)	11380.99
e. Agri	(81.15)	(16.84)		(312.59)	D31.96	(357.45)		179,840			D31.941	(312.41
d. Satiety	4.36	(9:33)		(5.30)	[25.300]	191.323	4.30	(9.5%)			\$25,360	963.50
s. Somewatin Deergy		1,000,000	1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1				1,000,000				7.000.00	126.17
The state of the s	81.37	12.64	20.46	133-41	38.44	526.17	81,17	52.64		133.41	36,44	
f, wet Ca	33,38	23,08	47.46	57.26	106.25	175.61	33.56	21.88	67.46	57.36	996,25	179,65
g. Hertiags Foods Rated Limited				-	- 20	-			- 65	5.0		(0.24
h, Heritage Conpre Ltd	-	1.0	-	-	William to	4	7.0	1000		1		(0.17)
Total Segment Results	2452.18	3784.17	2778.72	5226.34	4856-68	20080.06	2453.18	2764.17	3273,72		8306.68	10079.60
Less 1. Finance south	245.56	269.60	400.79	529.36	837.83	2545.89	245.56	265,60	400.79	309.16	817.50	1545.81
II. Other sevalocable expenses not off	2000	S 1988	0.300	5,50	7.6	100000	1981	1 1 1 1 1 1 1 1	1-30	7.050	28.4	1 251
Add is Manual Income	7.97	5.41	3.97	13.36	35.74	61,01	7.97	5.40	3,97	13.36	55.74	63.00
is. Other smallocable income	4.00	100	3.00	4.00	1.08	3.04	A-00		3.56	4.00	3.09	3.00
Total Profit before Tae	2218.55	2505.96	2,579.00	A724.55	A068.18	X599.25	2218.59	2525.96	2329.95	4724.15	4599.18	81/97 84
S. Segrecott Assatts:												
s. Dairy	28942.12	30729,55	27529.43	25942.12	27529.43	20794.64	38542.12	30729.55	27529.13	39942.12	25529.40	30764-84
t. Setal	14440.91	13970.57	13878.30	14440.93	11479.30	13665.42	34443.91	13970.52	11874.39	14440,91	11878.39	13063.4
e, Apri	2723.62	303.8.50	3502.75	2773.63	3182.31	2958.55	2223.62	2013.30	31102.31	2729.63	3002.01	2958.5
d. Bakery	5039.96	1140.98	1108.44	1009.50	1308.49	3130.32	1233.18	1342 98	1108.40	1. 2000000	13396-64	1130 8
a. Farawable Energy	4521,79			4523,79	1542.77	2915.89	4521.29	2899.29				2966.85
I. Vet Ce	1105.40					103.43	3305.40	933.84				963.4
g. Heritage Foods Retail Limited				0.00			0.00	0.00				8.00
N. Harttage Corpro Utd	41	-		0.00			1.50	2.00			0.10	11.00
i. Straitscated	1445.78	1640.12	1294.81	1645.79	1394.81	1921.33	146.26	1445.12			1294.83	1971.0
Total	\$4219.60	54131.31	47242.77	54219.60		53746.82	54221.50	54131.31			AT242.77	53745.8
4. Segreent Lightitries:	1 4 3 4 3 5	1	0.752.00	1000000	1000	775500	1000	100000	177115	-	E-127.12	17.70
a. Dairy	13858.17	11768.55	12776.95	11658,17	10275.95	13170.99	13858.37	11768.55	10278.95	131551.17	10078.95	11170.9
b Setal	4720.84	440,000,000			4161.32	5081.15	4720.84	4802.96			4351.52	5031.1
E. 549.	284.68				194.09	241.80	284.68	268.22			134.09	241-80
d. Beltyry	135.87	21.30				66.01	135.87	73.38			The	66.0
e. Ronewable Energy	719.61	24.00		219.63	12.04	185.26	219.65	38.00			12.04	195.7
f Ver Ce	499.21	201.30		659.21	465.79	40.45	459.21	HLB			845.38	400.0
	490,61	301.30	100.00	0.00		444.85						
g. Heritage Foods Retail Limited	100	1 55		4			5.25	2,47	0.98	0.35	0.38	0.47
h. Heritage Corporated	4746.44		2000	0.00		22.25	0.34	2.40	0.32	0.24	0.32	0.40
i. Uralfocated	1349.86	1833.29	3449.56	1349,86	the same of the sa	1829.39	1345.86	1835.29	1443.66	1345.86	1449.66	1873.22
Total	18819.34	19120.34	18613.13	18878-14	26615.13	18657.92	18828.34	19121.20	18613.83	18829.74	16623.85	18958,7

The above results for the quarter ended September 38,2036 have been reviewed by the Auto Committee and soproved by the Board of Directors in their making had an Occober 26, 2016.

The Saturary Auditors have conducted a limited review of accounts for the allowed period.

As per the Accounting Standard (AS- 17), the Conguery has elemented trainy. Result, Agri, fallowy, Reviewalde Energy and Vet Ca segments as reportable segments. The segment was results are given above.

4 The Company has commissioned a 2.1 MW Wind Power Unit under Reviewable Energy regiment for coptive consumption in the month of Experiment, 2016 at Veginnary in Another Product.

5. Figures of previous period(s)/ year(s) have been regrouped/minninged inherever necessary

huanimasi

Vice Charperson & Managing Director DIN -00002743

Date: October 26, 3016 Place: Hydersbad

HERITAGE FOODS LIMITED

(Formerly known as Mrs. Heritage Foods (India) Limited)

CIN: L15209TG1992PLC014332

AN ISO: 22000 CERTIFIED COMPANY

Regil, DH: # 6-3-541/C, Paragoritia, Hyderahad - 500 082. Telaropesa, INDIA. Tel.: -91-40-23391221. 23291227 For: 30695458 armid: http://entagefoods.in. evibolitr: www.horizopelands.in

RAJU & PRASAD

CHARTERED ACCOUNTANTS

Head Office :

401, "Diamond House",

Adj. Amrutha Hills, Punjagutta, Hyderabad - 500 082.

Ph 040 - 23410404/05/06 Fax 040 - 23410403

E-mail: hyderabad@rajuandprasad.com rajuandprasad@gmail.com

LIMITED REVIEW REPORT

To The Board of Directors, M/s Heritage Foods Limited

We have reviewed the accompanying statement of unaudited standalone financial results of M/s Heritage Foods Limited for the period ended 30th September 2016. This statement is the responsibility of the Company's Management and has been approved by the Board of Directors / Committee of Board of Directors. Our responsibility is to issue a report on these financial statements based on our review.

We conducted our review in accordance with the Standard on Review Engagement (SRE) 2410, "Review of Interim Financial information performed by the Independent auditor of the Entity" issued by the Institute of Chartered Accountants of India. This standard requires that we plan and perform the review to obtain moderate assurance as to whether the financial statements are free of material misstatement. A review is limited primarily to inquiries of company personnel and analytical procedures applied to financial data and thus provide less assurance than an audit. We have not performed an audit and accordingly, we do not express an audit opinion.

Based on our review conducted as above, nothing has come to our attention that causes us to believe that the accompanying statement of unaudited financial results prepared in accordance with applicable accounting standards and other recognized accounting practices and policies, has not disclosed the information required to be disclosed in terms of Regulations 33 of the SEBI (Listing obligations and Disclosure Requirements) Regulations, 2015, as amended from time to time, including the manner in which it is to be disclosed, or that it contains any material misstelement.

Place: Hyderabad Date: 28-10-2016 For Raju & Prasad Chartered accountants

Y Bala Kish Ready

Y. Bala Krishna Reddy Partner

Membership No: 223701 Firm Reg. No: 003475S

RAJU & PRASAD CHARTERED ACCOUNTANTS

Head Office

401, "Diamond House", Adj. Amrutha Hills,

Punjagutta, Hyderabad - 500 082 Ph 040 - 23410404/05/06

Fax 040 - 23410403

E-mail: hyderabad@rajuandprasad.com rajuandprasad@gmail.com

LIMITED REVIEW REPORT

To The Board of Directors, M/s Heritage Foods Limited.

We have reviewed the accompanying statement of consolidated unaudited financial results of M/s Heritage Foods Limited, its subsidiaries M/s. Heritage Foods Retail Limited and M/s. Heritage Conpro Limited and its associate M/s. SKIL Raigam Power (India) Limited for the quarter ended 30th September 2016. This statement is the responsibility of the Company's Management and has been approved by the Board of Directors / Committee of Board of Directors. Our responsibility is to issue a report on these financial statements based on our review.

We conducted our review in accordance with the Standard on Review Engagement (SRE) 2410, "Review of Interim Financial information performed by the Independent auditor of the Entity" issued by the Institute of Chartered Accountants of India. This standard requires that we plan and perform the review to obtain moderate assurance as to whether the financial statements are free of material misstatement. A review is limited primarily to inquiries of company personnel and an analytical procedure applied to financial data and thus provides less assurance than an audit. We have not performed an audit and accordingly, we do not express an audit opinion.

Based on our review conducted as above, nothing has come to our attention that causes us to believe that the accompanying statement of unaudited financial results prepared in accordance with applicable accounting standards and other recognized accounting practices and policies, has not disclosed the information required to be disclosed in terms of Regulations 33 of the SEBI (Listing obligations and Disclosure Requirements) Regulations, 2015 as amended from time to time, including the manner in which it is to be disclosed, or that it contains any material misstatement.

Place: Hyderabad Date: 28-10-2016 For Raju & Prasad Chartered accountants

Y. Bala Kishna Reddy

Partner

Membership No: 223701 Firm Reg. No: 003475S

RAJU & PRASAD CHARTERED ACCOUNTANTS

Head Office :

401, "Diamond House".

Adj. Amrutha Hills,

Punjagutta, Hyderabad - 500 082 040 - 23410404/05/06

040 - 23410403

Fax E-mail: hyderabad@rajuandprasad.com

rajuandprasad@gmail.com

Independent Auditor's Report

To The Board of Directors Heritage Foods Retail Limited

Report on the Standalone Interim Financial Statements

We have audited the accompanying standalone interim financial statements of Heritage Foods Retail Limited ("the Company"), which comprise the balance sheet as at 30 September 2016, the statement of profit and loss for the six months then ended and the statement of cash flows for the six months then ended and a summary of the significant accounting policies and other explanatory information.

Management's Responsibility for the Standalone Interim Financial Statements

The Company's Board of Directors is responsible for the preparation of these standalone interim financial statements that give a true and fair view of the financial position, financial performance and cash flows of the Company in accordance with the accounting principles generally accepted in India, including the Accounting Standards 25, Interim Financial Reporting as specified under section 133 of the Companies Act, 2013 ('the Act') read with relevant rules issued thereunder.

This responsibility also includes maintenance of adequate accounting records in accordance with the provisions of the Act for safeguarding of the assets of the Company and for preventing and detecting frauds and other irregularities; selection and application of appropriate accounting policies; making judgments and estimates that are reasonable and prudent; and design, implementation and maintenance of adequate internal financial control, that were operating effectively for ensuring the accuracy and completeness of the accounting records, relevant to the preparation and presentation of the interim financial statements that give a true and fair view and are free from material misstatement, whether due to fraud or error.

Auditor's Responsibility

Our responsibility is to express an opinion on these standalone interim financial statements

We have taken into account the provisions of the Act, the accounting and auditing standards and matters which are required to be included in the audit report under the provisions of the Act and the Rules made thereunder.

We conducted our audit of the standalone interim financial statements in accordance with the Standards on Auditing issued by the Institute of Chartered Accountants of India, Those Standards require that we comply with ethical requirements and plan and perform the audited a

New Delhi Branch: A 20/20, DLF City. Phase - I, Gurgaon - 122 002

Ph: 0124-4056132

Mumbai Branch: 511, The Corporate Centre, Nirmal Lifestyle Mail, L.B.S. Road, Mulund West Mumbai - 400 080 Ph: 022-25671155 / 1199

Bangalore Branch: # 202, Akash Avenue, #34, 1st Main, SBM Colony, Anand Nagar, Bangalore - 560 024 Ph: +91-9241570047

Jalgaon Branch: Flat No. 7. Knunal Apartments. Ganapathi Nagar Jalgson - 475-001 Cell: +91-9966738900 Navi Mumbai Brancho 8-712, Groma House Near APMC Market, Vashi.

Navi Mumbai - 400703 Cell: +91-9967220322

URL: www.rajuandprasad.com

RAJU & PRASAD CHARTERED ACCOUNTANTS

Head Office :

401, "Diamond House",

Adj. Amrutha Hills,

Punjagutta, Hyderabad - 500 082 Ph : 040 - 23410404/05/06

Fax : 040 - 23410403

E-mail: hyderabad@rajuandprasad.con rajuandprasad@gmail.com

Standards require that we comply with ethical requirements and plan and perform the audit to obtain reasonable assurance about whether the standalone interim financial statements are free from material misstatement.

An audit involves performing procedures to obtain audit evidence about the amounts and disclosures in the standalone interim financial statements. The procedures selected depend on the auditor's judgment, including the assessment of the risks of material misstatement of the standalone interim financial statements, whether due to fraud or error. In making those risk assessments, the auditor considers internal financial control relevant to the Company's preparation of the standalone interim financial statements that give a true and fair view in order to design audit procedures that are appropriate in the circumstances. An audit also includes evaluating the appropriateness of accounting policies used and the reasonableness of the accounting estimates made by the Company's Directors, as well as evaluating the overall presentation of the standalone interim financial statements.

We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion on the standalone interim financial statements.

Opinion

In our opinion and to the best of our information and according to the explanations given to us, the aforesaid standalone interim financial statements give the information required by the Act in the manner so required and give a true and fair view in conformity with the accounting principles generally accepted in India, of the state of affairs of the Company as at 30 September 2016 and its profit and loss account and its cash flows for the year ended on that date.

For Raju and Prasad Chartered Accountants

FRN: 003475S

Y. Bala Krishna Reddy

Partner

M. No: 223701

Date: 07/11/2016 Place: Hyderabad

New Delhi Branch: A 20/20, DLF City, Phase - I, Gurgaon - 122 002 Ph: 0124-4056132

Mumbai Branch; 511, The Corporate Centre, Nirmal Lifestyle Mall, L.B.S. Road, Muland West, Mumbai - 400 080 Ph; 022-25871155 / 1199

PRA

HYDERUBAD

EREDP

Bangalore Branch: # 202, Akash Avenue, # 34, 1st Main, SBM Colony, Anand Nagar, Bangalore - 560 024 Ph : +91-9241570047

Jalgson Branch: Flat No. 7, Krunal Apartments, Ganapathi Nagar, Jalgson - 475 001 Cell : +91-9956738900

Navi Mumbai Branch: B-712, Groma House, Near APMC Market, Vashi, Navi Mumbai - 400703 Cell: +81-9967220322

URL: www.rajuandprasad.com

Heritage Foods Retail Limited Balance Sheet as at 30th September, 2016

(Amount in Rs.)

			(Amount in Rs.)
Particulars	Note No	As at 30.09.2016	As at 31,03,2016
EQUITY AND LIABILITIES			
1) Shareholders' Funds			
a) Share Capital	- 3	1656000	1656000
b) Reserves and Surplus	1 2	(1409142)	(1409027
Sub Total		246858	246973
2) Share application money pending allotment		-	
3) Non-Current Liabilities			
4) Current Liabilities			
Other current liabilities	3	25054	47313
Total		271912	294286
ASSETS			
1) Non-current assets		•	20
2) Current assets			
Cash and cash equivalents	4	271912	294286
Total		271912	294286
Summary of significant Accounting Policies & Notes to Financial statements	1 to 14		

As per our report attached

For Raju & Prasad

Chartered Accountants

(Firm No.003475S)

Y. Bala Ken Lead Y. BALAKRISHNA REDDY

Partner

Membership No.223701

Date: November 07, 2010

HYDERABAD

Place: Hyderabad

For and on behalf of the Board

N. BRAHMANI

Managing Director

DIN: 02338940

M. SAMBASIVA RAO

Director DIN: 01887410

(Amount in Rs.)

Heritage Foods Retail Limited

Statement of Profit and Loss for the Period ended 30th September, 2016

Period Ended Period Ended Particulars Note No 30.09.2016 30.09,2015 Revenue from operations II. Other Income III. Total Revenue (I +II) IV. Expenses: Cost of materials consumed Purchase of Stock-in-Trade Changes in inventories of finished goods, work-inprogress and Stock-in-Trade Employee benefits expense Finance costs Depreciation and amortization expense Other expenses 115 Total expenses 115 V. Profit before exceptional and extraordinary items and tax (115)VI. Exceptional Items VII. Profit before extraordinary items and tax (V - VI) (115)VIII. Extraordinary Items IX. Profit before tax (VII - VIII) (115)X. Tax expense:

As per our report attached For Raju & Prasad

XII. Earning per equity share:

XI. Profit(Loss) from the perid from continuing

Summary of significant Accounting Policies & Notes to Financial statements

Chartered Accountants (Firm No.003475S)

1. Bala rishu lead

Y.BALAKRISHNA REDDY

Partner

operations

(1) Basic

(2) Diluted

Membership No.223701

Date: November 07, 2016

Place: Hyderabad

For and on behalf of the Board

(115)

(0)

(0)

N. BRAHMANI

Managing Director

I to 14

DIN: 02338940

M. SAMBASIVA RAO

Director

DIN: 01887410

Heritage Foods Retail Limited CASH FLOW STATEMENT FOR THE PERIOD ENDED SEPTEMBER 30, 2016

PARTICULARS	Best Awar a	(Amount in Rs.
	Period Ended	Period Ended
	30.09.2016	30.09.2015
A. CASH FLOW FROM OPERATING ACTIVITIES		
Net Profit / (Loss) before tax	- 1	
Adjustment for	(115)	
Depreciation		
Interest		
00015075		
Operating Profit before working capital changes		
Adjustment for:	(115)	
Inventories	100	
Trade and other Receivables	-	
Other current Liabilities	-	
switch destincts	(22259)	(14,013)
Cash generated from operations		12 110 200
Direct taxes paid (incl.taxiation of earlier years)	(22374)	(14,013)
para (arcittaxiation of earner years)		(-1010)
Vet Cash (Used) / generated for / from		1.50
operations		
THE CONTRACTOR	(22374)	(14013)
B. CASH FLOW FROM INVESTING ACTIVITIES		(10.0000)
THE MANAGEMENT ACTIVITIES		
CASH FLOW FROM FINANCING ACTVITIES		
roceeds from issue of Equity Shares		
som issue of equity shares		
et Cash (Used) / generated for / from		
inancing Activities		
et increase / (decrease) in cash and cash		
quivalents	(22374)	(14,013)
	10	100000000
ash and Cash equivalents as at the beginning of the year	294286	323314
ash and Cash equivalents as at end of the Period	271912	309301
ummary of significant Accounting Policies & Notes to Financial statements		309301
The second secon	1 to 14	

As per our report attached

For Raju & Prasad

Chartered Accountants

(Firm No.0034758)

Y.BALAKRISHNA REDDY

Partner

Membership No.223701

Date: November 07, 2016

Place: Hyderabad

For and on behalf of the Board

N. BRAHMANI

Managing Director

DIN: 02338940

M. SAMBASIVA RAO

Director

DIN: 01887410

Note 1 : Share Capital (Amount in Rs.) As at 30.09.2016 As at 31.03.2016 Particulars No. of Shares Amount No. of Shares I. Authorised Amount a) Equity Share Capital Equity Shares of Rs. 10/- each 15000000 1500000000 15000000 150000000 Total 15000000 150000000 15000000 1.5E+08 II. Issued,Subscribed and Paidup Share Capital : **Equity Share Capital** Equity Shares of Rs. 10/- each 165600 1656000 165600 1656000

III. Reconciliation of number of shares outstanding	As at 30	.09.2016	As at 31.0	nount in Rs.)
Particulars	The second secon	Equity Shares		A Continue Constitution of Co. Co.
	Number	Amount	Equity SI Number	Amount
Shares Outstanding at the beginning of the year	165600	1655000		describe excess
Shares issued during the year	100000	1656000	165600	1656000
Shares bought back during the year		-		-
Shares outstanding at the end of the	-	-	-	
year	165600	1656000	165600	1656000

IV. Shares held by Shareholders holding more than 5%

400000000000000000000000000000000000000	As at 3	0.09.2016	As at 31.0	3.2016
Name of Shareholder	No.of Shares held	% of Holding	No.of Shares	% of
a. Equity Share Capital :	-		held	Holding
1. HERITAGE FOODS LTD	165000	99.64	165000	00.0
		33.04	165000	99.64

V. Rights, Preferences and restrictions attached to equity shares

The Company has only one class of Issued, subscribed and paid up equity shares having a par value of Rs.10/- each per share. Each holder of equity shares is entitled to one vote per share. The Company declares and pays dividends in Indian rupees. The dividend proposed by the Board of Directors is subject to the approval of the share holders in the Annual general meeting.

In the event of liquidation of the Company, the holders of equity shares will be entitled to receive remaining assets of the Company, after distribution of all preferential amounts. The distribution will be in propotion to the number of equity shares held by the share holders.



Note 2: Reserves and Surplus

	(Amount in Rs.
	As at
016	31.03.2016
-	
-	
	-
-	,
200000	
09027)	(1384932)
15.00)	(24,095)
-	
-	
-	
-	-
9142)	(1409027)
9142)	(1409027)

The of Other Current Diabilities		(Amount in Rs.)
Particulars	As at	As at
Other Pavables	30.09.2016	31.03.2016
TOTAL	25054	47313
TOTAL	25054	47313

Note: 4 Cash and cash equivalente

cash and cash equivalents		(Amount in Rs.)
Particulars	As at	As at
a.Balances with banks:	30.09,2016	31.03.2016
On current accounts b.Cash on hand	265675 6237	284990
Grand Total	271912	9296 294286

Particulars	Period Ended	(Amount In Rs. Period Ended
	30.09.2016	30,09.2015
Printing and Stationery		
Rates and taxes, excluding, taxes on income Legal and Professional Charges	-	
Bank charges	100	
Payment to Auditors :	115	
Audit Fee		
otal	-	
DULL	115	



Note 5: SIGNIFICANT ACCOUNTING POLICIES AND NOTES TO FINANCIAL STATEMENTS

- a) Basis of preparation of financial statements: The financial statements have been prepared under historical cost convention and in conformity with the generally accepted accounting principles, applicable provisions of the Companies Act, 2013, and as per the Accounting standards specified under section Companies Act, 2013 read with rule? of Companies (Accounts) Rules 2014. Financial statements have been presented as per the schedule III of
 - b) Changes in Accounting policies: Accounting policies have been consistently applied except where a newly issued Accounting standard is mittally adopted or a revision to an existing accounting standard requires a chage in the accounting policy hitherio in use. Mangement evaluates all recently or revised Accounting standards on an ongoing basis and accordingly changes the Accounting policies as applicable. The same Accounting policies are followed in the Interim financial statements as those followed in the most recent annual financial statements.
- The Material event to the subsequent of end of the Interim period that have not been refelected in the finacial statements for the Interim Period:

The Board of Directors of the company have inter alia, considered and approved the scheme of Amingement between Heritage Foods Limited(HFL), Heritage Foods Retail Limited (HFRL) and Future Retail Limited on 7th November 2016 subject to approval of their respective Share holders and Creditors under section 391-394 read with section 100-103 of the Companies Adf 1956 and section 52 of the Companies Act 2013 and respective applicable provisions of the Companies Act 2013

d) Use of Estimates: The preparation of financial statements in conformity with the Indian GAAP requires mangement to make judgements, estimates and assumptions that affect the reported balances of assets and liabilities and disclosures relating to contingent liabilities as at the end of financial statements and reported amounts of income and expenses during the period.

Accounting estimates could change from period to period. Actual results could differ from those estimates. Appropriate changes in estimates are made as the management becomes aware of changes in circumstances surrounding the estimates. Changes in estimates are reflected in financial statements in the period in which changes are made and, if material, their effects are disclosed in the notes to the financial statements.

- e) Cash Flow statement: The cash flow statement is propored by the "Indirect Method" set out in Accounting standard (AS-3) "Cash Flow statements" and presents the cash flows by operating, financing and investing activities of the company. Operating cash flows are arrived by adjusting profit or loss before tax for the effects of transactions of a non-cash nature, and deferral of actual of past or future operating cash receipts or payments, and items of income or expense associated with investing or financing cash flows.
- f) Cash and Cash Equivalents: Cash and and Cash Equivalents comprises of cash on hand, cash at bank.
- g) Earnings per share: Basic earnings per share is computed by dividing the net profit after tax by the weighted average number of equity shares outstanding during the period. Diluted earnings per share is computed by dividing the profit after tax by the weighted average number of equity shares that could have been issued upon conversion of all dilutive potential equity shares.
- h) Provisions: Provision is recognised when the company has a present obligation as a result of past event and it is probable that an outflow of resources embodying economic benefits will be required to settle the obligation and a reliable estamate can be made of the amount of the obligation. Provisions are not discounted to their present value and are determined based on the best estimate required to settle the obligation at the year end. These estimates are reviewed at each year end and adjusted to reflect the current best estimates.
- ii) Contingent Liabilities: A contingent liability is a possible obligation that arises from past events and the existance of which will be confirmed only by the occurrence or non-occurrence of one or more uncertain future events not wholly whithin the control of the enterprise or a present obligation that arises from past events but is not recognized because it is not probable that an outflow of resources embodying economic banefits will be required to settle the obligation or a reliable estimate of the amount of the obligation cannot be made.

Note 7: Contingent Liabilitis and Commitments (To the extent not provided for)

Particulars (1) Contingent liabilities shall be classified as:	As at 30.09.2016	As at 31.03.2016
 (σ) Claims against the company not acknowledged as debt; (δ) Guarantees; 	NIL	NIL.
(c) Other money for which the company is contingently liable (ii) Commitments shall be classified as:	MIL	NIL
 (a) Estimated amount of contracts remaining to be executed on capitacount and not provided for; (b) Heaviled by 10. 	ital NIL	NL
(b) Uncalled liability on shares and other investments partly paid;(c) Other commitments (specify nature).	NIL	MIL
to y come communicates (specify nature).	NIL	NIL
	NIL	NIL.

S PRASTO PROBLEMENT PR

Note 7: Contingent Liabilitis and Commitments (To the extent not provided for)

		(Amt in Rupees)
Perticulars	As at 30,09,2016	As at 31.03.2016
 (r) Contingent liabilities shall be classified us: (a) Claims against the company not acknowledged as debt; 	NIL	NIL
(b) Guarantees; (c) Other money for which the company is contingently liable (ii) Commitments shall be classified as:	NIL	NE
(a) Estimated amount of contracts remaining to be executed on capital account and not provided for; (b) I to effect to be a contract.	NIL	NIL
(b) Uncalled liability on shares and other investments partly paid; (c) Other commitments (specify nature)	NIL	NS.
otal	NIL	NIL.
SSR Employees honeits	NH.	NIL.

There are no employees engaged by the company. Hence no provision towards employee benefits has been made

Note 9. Segment Reporting :

The Company has not comenced commercial operations. Hence reporting as per AS17: Segment Reporting is not applicable.

Note 10. Related Party disclosures as per AS-18

Name of the related party	Heritage Foods Limited	Heriage Conpro Limited	N Bhuvaneswari	N.Brahmani	N Lokesh	M Sambasiva Rac
Description of the Relationship between the parties	Holding Company	Fellow subsidiary	Key Management Personnel (Director)	Key Management Personnel (Managing Director)	Personnel	Key Management Personnel (Director)
Description of the nature of transactions	Inter-company transactions	Inter company transactions	a) Receiving of Services	Receiving of Services	Receiving of Services	Receiving of Services
Volume of the transactions either as an amount or as appropriate proportion	NIL.	NIL.	NE.	NIL	NIL	NL
Any other elements of the related party transacting of the financial statements	NIL.	NIL	NIL	NIL	NIL	NE.
The amounts or appropriate proportions of outstanding items pertaining to stated parties at the Balance Sheet tate	NIL	NIL:	NIL.	NL	NIL	NIL
Amounts written off or written back in the period in respect of debts due from or to related parties.	NIL.	NIL.	NIL.	NIL	NIL	NE.



Note 11. Earnings per share (EPS) (Amt in Rupees) Particulars As at As 30 a) Calcutation of weighted average number of equity shares of Rs 10/- each 30.09.2016 31.03.2018 Number of Equity Shares outstanding at the beginning of the year Number of Equity Shares outstanding at the end of the year 165600 Weighted average number of Equity Shares outstanding during the year 165600 165600 155600 b) Net Profit/(Loss) after tax 165600 165600 c) Basic Earnings per share of Re 10/- each (115) (24095) d) Diluted Earnings per share of Rs 10/- each (0) (0.15)(0) $\{0.15\}$

Note 12. There is no amount or interest which are outstanding for more than 45 days payable to Micro, Small and Medium Enterprises Development Act, 2006 as at 31st March 2016.

Note 13. Expenditure, Earnings, Remittance in Foreign Currency

Particulary	For the Year ended	(Amount in Rs.) For the Year ended
2 Expenditure in Foreign Currency:	30.09.2018	31.03.2018
3 Earnings in Foreign Currency	NIL	NIL
4 Remittance in Foreign Currency	NIL	NIL.
	NIL	NIL

Note 14. Paise is rounded off to the nearest Rupee

As per our report attached for Raju & Prasad Chartered Accountants (Firm No.0034755)

Y.BALAKRISHNA REDDY

Partner

Membership No.223701

Date: November 07, 2016 Place: Hyderabad For and on behalf of the Board

N. BRAHMANI

Managing Director

DIN: 02338940

M. SAMBASTVA RAD

Director

DIN: 01887410

FUTURE RETAIL

30th August, 2016

To.

Department of Corporate Services

BSE Limited

Phiroze Jesjesbhoy Towers,

Dalal Street, Mumbai - 400 001

Scrip Code: 540064

To, Listing Department National Stock Exchange of India Limited Exchange Plaza, Bandra Kurla Complex, Bandra (East), Mumbal - 400 051 Scrip Symbol : FRETAIL

Dear Sirs.

Sub.: Board Meeting Decision / Outcome

A meeting of Board of Directors of the Company was held today, the 30th August, 2016, the Board, inter alia, took the following decisions:

- The Board considered and approved the Unaudited Financial Results for the quarter ended 30th June, 2016 along with the Limited Review Report of the Statutory Auditors, which are attached herewith pursuant to the Regulation 33 of SEBI (Listing Obligations and Disclosure Requirements) Regulations, 2015;
- Structuring of a Share Based Employee Benefit Scheme with a ceiting limit of near to 2% (Two) of the paid-up
 equity share capital of the Company, computed as at the end of the quarter ended 30 June 2016 based on
 recommendation of Nomination and Remuneration Committee, which is subject to the approval of shareholders
 of the Company.

The above Financial Results have been reviewed by the Audit Committee and approved by the Board of Directors at its meeting held today.

The meeting of the Board of Directors of the Company commenced at 11:45 am and concluded at 01:50 pm

Kindly take the above information on your records.

Yours truly, For Future Retail Limited

Nitruch Euro

Virendra Samani Dy. Company Secretary

Enci: as above



Fature Retail Limited Furnery Incomes Brain Floral Colonia
Technology Office Amonyology Process Nagar Cell Engineering Technology Floral Englander (Nagar Floral Englander) (

FUTURE RETAIL

	Statement of Standalane Financial Regula for the Guarter Ende		6Z at Crorel
Sr. No.	Particulars	3 months unded 30-06-3016	Corresponding 3 months anded in provious year 30-06-2715
_		Unsudited	Unautited
-	Ipcome From Operations		
-	al Net assestedome from operations	3,891,51	408-A1
_	b) Other Ocerating Income	68,17	1.16
_	Total Incurse from operations	3,979.69	410.07
2	Experises		
-	a) Purchases of Words-In-Disde	3,109,42	216.94
_	(b) Changes in triversones of apok-in-diade - (trore-isie) Decimale	(130,13)	53.4
_	c) Employee benefits exponds	188,77	49.00
_	g Depreciption and amortisation experies	3.87	10.07
-	e) Rent including Lesson nitritie	331.19	24.61
-	fr Coner Expanses	263.61	62.3
-	Y-tol Expenses	5,869.76	408.20
1	Profit (Lose) from Operations before other income, finance costs and Exceptional	109.92	(77.83
-	Items (1-2)	5.54	0.3
0	Profit/[Losa) from ordinary activities before finance costs and Exceptional items	115.46	(27,4)
_	(2+4)	44.01	9.4
	Frontic costs Profit(Loss) from ordinary cotivides after finance costs but before Escapitional	79.65	(06.0)
	items (\$-8)	V.	
- 8	Exceptional literis	70.55	(16.5
. 9	Protti[Loss) from ordinary activities before tax (7+8)		
10	The second secon	70.66	(86.0)
. 11	Not Profit(Lons) after Tax (5-19)	(0.24	(0.1
12	Other Comprehensive Income (OCI) (net of tax)	70.51	
13		94.27	1,715.3
18		1,(4)	(21)

Notes

The Company adequed tracks: Accounting Standards ("Ind AS") from 5 April 2016 and accordingly these Enancial result have been prepared is accordance with recognition and measurement principles laid down in the lod A5-34 Teterits Financial Reporting prescribed under section 133 of the complemes Act., 2912 read with the relevant rules issued there under and the other accounting principles permissy accepted in India. Financial resums for all the periods prescrited save seen prepared in acquisitance with the recognision and measurement grindplea of Ind AS-34.

Recomplished of Net Lake for the porterpording quarter ended 20 June 2015 between previous index GAAP and ind AS is as under

Quarter ended 30-June 2016
(67.05)
(0.31
0.15
(86.96)

- The above results were reviewed by the Audit Commissed and were thereafter approved by the Soord of Directors of this Company at its meeting beld on 30 August, 2016. Statutory Auddons have lacued Limited Review report on the same
- Figures for the provious financial period have been re-arranged and re-groupes whosever necessary and are not comparable for the mason of effect gives to the Scheme of Amingaissen
- The Name of the company has been changed from Shard Retail Limited to Future Retail Limited way 25 May 25 No
- This Federical Results and other financial information for the quarter ended 20 June 2015 have been reviewed and have been presented based on the information complete by the management after making the necessary adjustment to give a true and for new in the results in appointmon with and 45
- During the sputter under review, the Composite Schöme of energyment between enternie Public Retail Landed (now engan self-bours) Emergence Livroised) and Congressy and their respective decisions and shareholders ("the Squares") was given effect on 1 May 2915 based on Scheme approved by the Hombie Bornous High Court on 4 March 2016. The Scheme was made effective with Approved Data of 31
- Published to the Scheme, 42,78.80.296 equity shares of CD- cosh were slocked to the shareholders of Future Enterprises Limited on 18 May 2016 and the Company has made application to BSE Limited and National Stock Exchange of India Limited for exemption under Rule 19 (b), of Securities Contracts (Magnitum) Rules, 1957, and he integ of the shares to Stock Exchanges. Pursuant to appearing received from stack exchanges and Selbi, trading in shares commerced with effect from 29 August 2016.
- The Company has only one business sugment Le. "Rator"

For Future Retall Limited

Xishare Biyant Chairman & Managing Director

Piece - Norths

Date 30 August, 2016 roture recant Limited !

Registered Officer is insurance modes, fitness league, DH Jugostnigas SANTEPERS PLANTED BOTH TWO FACES AT MICE SCATT, WHEN TO A LONG THE PARTY OF

SW-NZ-NZSWYWZGO-FAZOWYW



LIMITED REVIEW REPORT

To,
The Board of Directors
Future Retail Limited
(formerly known as Bharti Retail Limited)

We have reviewed the accompanying Statement of Standalone Unaudited Financial Results of Future Retail Limited (formerly known as Bharti Retail Limited) ("the Company") for the Quarter ended June 30, 2016 ("the Statement") being submitted by the Company pursuant to the requirement of Regulation 33 of the SEBI(Listing Obligations and Disclosure Requirements) Regulations, 2015, as modified by Circular No. CIR/CFD/FAC/62/2016 dated July 5, 2016. This Statement which is the responsibility of the Company's Management and approved by the Board of Directors, has been prepared in accordance with the recognition and measurement principles faild down in the Indian Accounting Standard 34 "Interim Financial Reporting" ("Ind AS 34"), prescribed under Section 133 of the Companies Act, 2013 read with relevant rules issued thereunder and other accounting principles generally accepted in India. Our responsibility is to issue a report on the Statement based on our review.

We conducted our review of the Statement in accordance with the Standard on Review Engagements (SRE)2410 "Review of Interior Financial Information Performed by the Independent Auditor of the Entity", issued by the Institute of Chartered Accountants of India. This Standard requires that we plan and perform the review to obtain moderate assurance as to whether the Statement is free of material misstatement. A review is limited primarily to inquiries of Company personnel and analytical procedures applied to financial data and thus provides less assurance than an audit. We have not performed an audit and, accordingly, we do not express an audit opinion.

Based on our review conducted as stated above, nothing has come to our attention that causes us to believe that the accompanying Statement of unaudited financial results prepared in accordance with the alloresaid Indian Accounting Standard and other accounting principles generally accepted in India, has not disclosed the information required to be disclosed in terms of Regulation 33 of the SEBI(Listing Obligations and Disclosure Requirements) Regulations, 2015, as modified by Circular No. CIR/CFD/FAC/62/2016 dated July 5, 2016, including the manner in which it is to be disclosed, or that it contains any material misstatement.

We have not reviewed the financial results and other financial information for the quarter ended June 30, 2015.

For NGS & Co. LLP Chartered Accountants

Firm Registration No: 119850W

Ashok A: Trived

Partner

Membership No.042472

Mumbal

August 30, 2016

ON THE PARTY OF TH



Compliance Report on Corporate Governance as on 30th September, 2016

1. Name of Listed Entity: Heritage Foods Limited

2. Quarter ending: 30th September, 2016

I.	Composition of B	oard of Dir	ectors				()	
Title (Mr/ Ms)	Name of the Director	DIN	Category (Chairperso n /Executive/\ Non- Executive/in dependent/N ominee)	Date of Appoint ment in the current term /cessation	Ten ure*	No of Directorship in listed entities including this listed entity (Refer Regulation 25(1) of Listing Regulations)	Number of membership s in Audit/ Stakeholder Committee(s) including this listed entity (Refer Regulation 26(1) of Listing Regulations)	in Audit/ Stakeholder
Mr.	D Seetharamaiah	00005016	Non-Executive Independent Chairperson	26.09.2014	5	1	2	1
Mr.	N Srivishnu Raju	00025063	Non-Executive Independent Director	26.09.2014	5	2	3	*
Mr.	V Nagaraju Naidu	00003730	Non- Executive Director	26.09.2014	-	1	1	1
Mr.	N Lokesh	02230945	Non- Executive Director	24.09.2015	2	1	1	ž.
Mrs.	N Bhuvaneswari	00003741	Vice Chairperson & Managing	01.07.2014	×	1	1	8
Mrs.	N Brahmani	02338940	Executive Director	01.06.2014	-	1	1	25
Mr.	Rajesh Thakur Ahuja	00371406	Non-Executive Independent Director	23.03.2016	5	1	1	2

&Category of directors means executive/non-executive/independent/Nominee. if a director fits into more than one category write all categories separating them with hyphen

* to be filled only for Independent Director. Tenure would mean total period from which Independent director is serving on Board of directors of the listed entity in continuity without any cooling off period.



HERITAGE FOODS LIMITED

(Formerly known as Ms. Heritage Foods (India) Limited) CIN: L15209TG1992PLC014332







Name of Committee	Name of Committee	Category(Chairperson	1 - Carl Carl Carl Carl Carl Carl Carl Carl	
	members	Executive/independe	nt/Nominee) \$	
Audit Committee	D. Seetharamaiah Dr. V. Nagaraju Naidu N. Sri Vishnu Raju Rajesh Thakur Ahuja N. Lokesh	Chairperson (Non-Executive Independent) Member (Non-Executive) Member (Non-Executive Independent) Member (Non-Executive Independent) Member (Non-Executive)		
Nomination & Remuneration Committee	N. Srivishnu Raju D. Seetharamaiah N. Lokesh Rajesh Thakur Ahuja	Chairperson (Non-Exe Member (Non-Executive Member (Non-Executive Member (Non-Executive Member (Non-Executive	ve Independent) ve)	
Risk Management Committee(if applicable)	Rajesh Thakur Ahuja D. Seetharamaiah N. Bhuvaneswari N. Sri Vishnu Raju	Chairperson (Non-Executive Independent) Member (Non-Executive Independent) Member (Executive) Member (Non-Executive Independent)		
Stakeholders Relationship Committee'	Dr. V. Nagaraju Naidu D. Seetharamaiah N. Bhuvaneswari N. Sri Vishnu Raju	Chairperson (Non-Executive) Member (Non-Executive Independent) Member (Executive) Member (Non-Executive Independent)		
CSR Committee	D. Seetharamaiah N. Bhuvaneswari N. Sri Vishnu Raju	Chairperson (Non-Executive Independent) Member (Executive) Member (Non-Executive Independent)		
Management Committee	D Seetharamaiah N Srivishnu Raju N Lokesh N Bhuvaneswari	Chairperson (Non-Executive Independent) Member (Non-Executive Independent) Member (Non-Executive) Member (Executive)		
\$Category of directors : than one category write III. Meeting of Boar	means executive/non-executive/i all categories separating them wit d of Directors	ndependent/Nominee. if h hyphen.	a director fits into more	
Date(s) of Meeting (if any) in the previous quarter	Date(s) of Meeting (if any) in the relevant quarter	Maximum gap consecutive (in num	between any two aber of days)	
23rd May, 2016	29th July, 2016 19th August, 2016	67	67 Days	
IV. Meeting of Com	mittees - Audit Committee Mee	ting	1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -	
Date(s) of meeting of the committee in the relevant quarter	Whether requirement of Quorum met (details)	Date(s) of meeting of the committee in the previous quarter	Maximum gap between any two consecutive Meetings in number of days*	
23st May, 2016	Yes (4 out of 5 members)	29th July, 2016	67 Days	
	Stakeholders Relations	ship Committee		
29th July, 2016	NIL	NIL	NIL	



HERITAGE FOODS LIMITED

(Formerly known as M/s. Heritage Foods (India) Limited) CIN: L15209TG1992PLC014332

AN ISO: 22000 CERTIFIED COMPANY





	Risk Management	Committee	
NIL	NIL	NIL	NIL
	CSR Comm	ittee	
NIL	NIL	NIL	NIL
	Management C	ommittee	
NIL	NIL	17th June, 2016	NIL
	Nomination & Remune	ration Committee	
29th July, 2016	NIL	NIL	NIL
V. Related Party Transacti	ons		
Subj		Compliance state	us (Yes/No/NA)
Whether prior approval of a	udit committee obtained		W-3-2-30
Whether shareholder approved RPT	val obtained for material	N.A	
Whether details of RPT ente omnibus approval have bee Committee		N.A	
POSSESSES CONTRACTOR C			

Note

1 In the column "Compliance Status", compliance or non-compliance may be indicated by Yes/No/N.A.. For example, if the Board has been composed in accordance with the requirements of Listing Regulations, "Yes" may be indicated. Similarly, in case the Listed Entity has no related party transactions, the words "N.A." may be indicated.

If status is "No" details of non-compliance may be given here.

VI. Affirmations

- The composition of Board of Directors is in terms of SEBI (Listing obligations and disclosure requirements) Regulations, 2015.
- The composition of the following committees is in terms of SEBI(Listing obligations and disclosure requirements) Regulations, 2015
 - a. Audit Committee
 - b. Nomination & remuneration committee
 - c. Stakeholders relationship committee
 - d. Risk management committee (applicable to the top 100 listed entities)
- The committee members have been made aware of their powers, role and responsibilities as specified in SEBI (Listing obligations and disclosure requirements) Regulations, 2015.
- The meetings of the board of directors and the above committees have been conducted in the manner as specified in SEBI (Listing obligations and disclosure requirements) Regulations, 2015.

This report and/or the report submitted in the previous quarter has been placed before Board of Directors.

Any comments/observations/advice of Board of Directors may be mentioned here:

For HERITAGE FOODS LIMITED

UMAKANTA BARIK Company Secretary M. No. FCS-6317



HERITAGE FOODS LIMITED

(Formerly known as M/s. Heritage Foods (India) Limited) CIN: L15209TG1992PLC014332

AN ISO: 22000 CERTIFIED COMPANY





I Affirmations	ou	
Broad heading	Regulation Number	Compliance Status (Yes/No/NA)refer note below
Copy of the Annual Report including balance sheet, profit and loss account, directors report, corporate governance report, business responsibility report displayed on website	46(2)	YES
Presence of Chairperson of Audit Committee at the Annual General Meeting	18(1)(2)	YES
Presence of Chairperson of the nomination and remuneration committee at the Annual General Meeting	19(3)	YES
Whether "Corporate Governance Report" disclosed in Annual Report	34(3) read with Para C of Schedule V	YES

Note:

- In the column "Compliance Status", compliance or non-compliance may be indicated by Yes/No/NA. For example, if the Board has been composed in accordance with the requirements of Listing Regulations, "Yes" may be indicated. Similarly, in case the Listed Entity has no related party transactions, the words "N.A". may be indicated.
- 2. If status is "No" details of non-compliance may be given here.
- If the Listed Entity would like to provide any other information the same may be indicated here.

For HERITAGE FOODS LIMITED

UMAKANTA BARIK Company Secretary M. No. FCS-6317







FUTURE RETAIL!

			Con	repliance Report on Corp	erate Governanc	e		
1	Name of Listed Entity	Future Retail Limited						
2	Period Ending	September 30, 2016				-0 as 2		
				Ameure	1	The state of the s	TARTE OF THE PARTY	
9				1. Composition of Boars	of Directors			
mu jw	Mame of the Director	DIR	Category (Chairperson /Exceptive/M onExceptive/in stependent/M ornings)	Date of Appoint ment in the current term /ccssation	Tenure*	No of Directorship in Ested entities including this Rucel entity (Refer Regulation 25(1) of Listing Regulations)	Mumber of memberships in Audit/Scokeholder Committee(s) inducing this listed entity (Refer Regulation 26(3) of Listing Regulations)	No of port of Chairperson in Austry Straight adder Committee held in Used entities include this land entity (Refe Regulation 26(1) of Usting Regulations)
Mr	Kishore Lasminarayan Biyani	DIN No-00005740	Champerson- Executive	30-04-2016	N.A.	1		1
Mr.	Rakesh Biyani	DIN No-00005806	Executive	30-04-2026	NA.	b	3	0
Mr	Rajan Bharti Mittal	DIN No-00035016	Non-Esscutive	30-04-2008	N.A.	D	4	2
741	Shaifandra Bhandari	ON No-00317994	Independent	30-04-2016	G0 Months	1	1	1
Mr.	Bayindra Dhanaval	ON No-00009922	Independent	30-04-2016	60 Months	2	10	2
Ms	Gagar Sireh	ONN No-03257134	Indopeniera	30-04-2016	60 Months	2	4	2

Eightsgory of directors means accounting/non-association/independent/hornises. If a director fits into more category write all categories separating there with happen with a separating there with happen of the fitter on the fi

			IL Composition of Committees		
51. No.	Harse of Committee Mama of Committee members		Catagory (Chairperson/Executive/NonExecutive/And pendent/Nominee) \$		
1	Audit Correltse		Ms. Gegan Singh		Chairpasson - Non-Executive-Independent
			Mr. Revisaira Ohariwal		Non-Executive-Independent
	- Charles and a series of the	AND THE PERSON NAMED IN	Mr. Takesh Biyani		Executive
3	Nomination & Retrucers	Ion Committee	Mr. Revirsire Oterieval		Chairpanton - Non-Esecutive-Independent
			Wr. Agan Bhart Mittel		Non-Executive
	Section 1		Mr. Shar energ Shandan		Non-Executive-Independent
1	Stalia holde v Reletionship	Сопутитея	Mr. Shallendre Shandari		Chaircenton - Non-Executive-Independent
			Mr. Caper Sirgh		Non-Executive-Independent
	at the same of	Mr. Telen Obarti Mittel			Non-Leaguille
Conego	y of directors means even	r/ve/con-executive/independent/Nomines. Fa-	firector fits into more than one category w	rite all categories separating ther	m with hyphen.
-			III. Meeting of Sound of Directors		
	Date	els) of meeting Enter dates of Previous quarter	and Current quarter in chronological ards	d	Maximum gap between any
		89th April, 20th (Pres.	four Quarteri		NA.
		#01h April, 2016 (Pres	riss Quarteri		22
		02nd Mag, 2016 (**no	rous Quanter!		1
		25th May, 2016 (*rev	lous Quarter)		22
		30th August, 3016 (Cu	rrent Quarter)		96
-	-		IV. Meeting of Correlities:		
	old of meeting of the se in the selevant quarter	Whether requirement of Quarum met (details)	Requirement of Quarum met (details)	Date(s) of meeting of the committee in the previous quarter	Miximum gap between any two consecutive meetings in number of days.*
-	0.00	A	Audit Committee		
Oth Aug	urt, 3018	Yes	All the Three members (including leve Independent Directors) were present in the Meeting	25th May, 2006	16
	Andrew Control	//	Stakeholders Relationship Committee	Marian Inc.	
			(To be held in future)		
		No.	miration & Remuneration Committee:	Anna de la companya d	
MILES - 100	nation & Remuneration se was held in this	N.A.	HA.	02nd May, 2016	NA.
		only be given for audit committee, for rest of the			



V. Related Party Transactions				
Subject	Compliance status (Yes/No/NA)refer note below			
Whether prior approval of sudit committee obtained	Yes			
Whether shareholder approval obtained for restence 1827	Yes			
Whether details of RPT entered into pursuant to cronibus approve have been reviewed by Audit Copyrillate	Tas			

Notes

- In the column "Contactions Status", compliance or non-compliance may be indicated by Yan/No/M.A. For example, if the Board has been composed in econtinuor with the requirements of Urting Regulations, "Yes" may be indicated. Similarly, in case the Listed Entity has no related party transactions, the words "N.A." may be indicated.
- If statue is "No" details of non-compliance may be given here.

- VI. Affirmations

 1. The composition of Scord of Directors is in larms of SEB (Listing oblique and disclosure requirements) Regulations, 2015.
- 2. The composition of the following committees is in terms of SEBN) ating obligations and disclosure requiremental Regulations, 2005
- a. Audit Committee
- Non-nation & resumeration committee
- c. Stakeholders relationship committee
- d. Sick management committee (applicable to the top 100 listed artities) Not Applicable
- 1. The committee members have been made ewers of their powers, his and responsibilities as specified in SED (Listing old garliers and distinctor require memos) Regulations, 2015.
- 4. The precings of the board of chectors and the above committees have been conducted in the manner as specified in SESI (Listing obligations and disclosure requiremental Regulations, 2005.
- 5. This report and/or the report submitted in the previous quarter has been placed before beard of Directors. As a communication occurrence of Science of Occupant may be mentioned here.

Note: Information at Table I and II above need to be necessarily given in 1st counter of each financial year. However if there is no change of information in subsequent quarterity of that financial year, this information may not be given by listed entity and instead autatement "same as previous quarter" may be given.

Place: Munical Date: 14th October, 2016



For Future Retail United

Hi ingle Stor Virendra Samani Dy. Company Secretary

Com	pliance Report on Corporate Governance					
	ANNEXURE III					
Affirmations						
Broad heading	Regulation Number	Compliance status (Yes/No/NA)				
Copy of the Annual Report including Balance Sheet, Profit and Loss Account, Directors' Report, Corporate Governance Report, Business Responsibility Report displayed on website.	46(2)	Yes				
Presence of Chairperson of Audit Committee at the Annual General Meeting.	18(1)(d)	Yes				
Presence of Chairperson of the Nomination and Remuneration Committee at the Annual General Meeting.	19(3)	Yes				
Whether "Corporate Governance Report" disclosed in Annual Report.	34(3) read with para C of Schedule V	Yes				

Note

1 in the column "Compliance Status", compliance or non-compliance may be indicated by Yes/No/N.A.. For example, if the Board has been composed in accordance with the requirements of Listing Regulations, "Yes" may be indicated. Similarly, in case the Listed Entity has no related party transactions, the words "N.A." may be indicated.

2 If status is "No" details of non-compliance may be given here.

3 If the Listed Entity would like to provide any other information the same may be indicated here.

Place: Mumbai

Date: 14th October, 2016

For Future Retail Limited

Virendra Samani Dy. Company Secretary



Compliance report with the requirements specified in Part-A of Annexure I of SEBI circular CIR/CFD/CMD/16/2015 dated November 30, 2015

Sub: Application under Regulation 37 of the SEBI (Listing Obligations and Disclosure Requirements), Regulations, 2015 for the proposed Composite Scheme of Arrangement Among Heritage Foods Limited ("Transferor Company" or "HFL") and Heritage Foods Retail Limited ("Transferoe Company" or "Demerged Company" or "HFRL") and Future Retail Limited ("Resulting Company" or "FRL") and their respective Shareholders and Creditors

In connection with the above application, we hereby confirm that we satisfy all the conditions as stipulated in the aforesaid SEBI circular, as given hereunder:

Sr. No.	Requirements as per SEBI circular CIR/CFD/CMD/16/2015 dated 30/11/2015	Whether Complied or not & How		
1.	Listed companies shall choose one of the stock exchanges having nation-wide trading terminals as the designated stock exchange for the purpose of coordinating with SEBI.	Yes, the company has chosen BSI Limited as the designated stock exchange.		
	Compliance as per Part A, Annexure I to the Circular			
2.	Documents to be submitted:			
2.a	Draft Scheme of arrangement/ amalgamation/ merger/ reconstruction/ reduction of capital, etc.	The Composite Scheme of Arrangement is enclosed with the application as per Annexure-B		
2.b	Valuation Report from Independent Chartered Accountant	Complied and attached in Annexure-C		
2.c	Report from the Audit Committee recommending the Draft Scheme	Report of Audit Committee is enclosed as per Annexure- D		
2.d	Fairness opinion by merchant banker	Attached in Annexure-E		
2.e	Pre and post amalgamation shareholding pattern of unlisted company	The Share holding pattern is attached in Annexure-F		
2.f	Audited financials of last 3 years (financials not being more than 6 months old) of unlisted company;	Complied and attached the last 3 years financial as per Annexure-G		
2.g	Compliance with Regulation 17 to 27 of Listing Regulations	The same is complied. The compliance report is enclosed as Annexure-H		
2.h	Complaints Report	The same shall be submitted within 7 days from the date of expiry of 21 days from the date of filing of the Scheme with the Stock Exchange.		
3.	The equity shares sought to be listed are proposed to be allotted by the unlisted Issuer (transferce entity) to the holders of securities of a listed entity (transferor entity) pursuant to a scheme of reconstruction or amalgamation (Scheme) sanctioned by a High Court under Section 391-394 of the Companies Act, 1956 or Section 230 to 234 of the Companies Act 2013	Not Applicable		
4.	At least 25% of the post scheme paid up share capital of the transferee entity shall comprise of shares allotted to the public holders in the transferor entity.	Not Applicable		
5.	The transferee entity will not issue/reissue any shares, not covered under the Draft scheme.	Not Applicable		
6,	As on date of application there are no outstanding warrants/ instruments/ agreements which give right to any person to take the equity shares in the transferee entity at any future date. If there are such instruments stipulated in the Draft scheme, the percentage referred to in point (4) above, shall be computed after giving effect to the consequent increase of capital on account of compulsory conversions outstanding as well as on the assumption that the options outstanding, if any, to subscribe for additional capital will be exercised.	Not Applicable		
7.	The shares of the transferce entity issued in lieu of the locked-in shares of the transferor entity are subjected to the lock-in for the remaining period.	This clause is not applicable		

Date: 14/11/2016

HERNAGE FOODS BIMITED
(Formerly known as We. Heritage Foods (India) Limited)

Umakanta Barik Company Secretary



RAJU & PRASAD CHARTERED ACCOUNTANTS

Head Office :

401, "Diamond House", Adj. Amrutha Hills.

Punjagutta, Hyderabad - 500 082 Ph 040 - 23410404/05/06

Fax : 040 - 23410403 E-mail: hyderabad@rajuandprasad.com

rajuandprasad@gmail.com

To, The Board of Directors, Heritage Foods Limited 6-3-541, Punjagutta, Hyderabad – 500 082

CERTIFICATE BY THE STATUTORY AUDITORS OF HERITAGE FOODS LIMITED

We, M/s Raju & Prasad Chartered Accountants (FRN No. 003475S) the statutory auditors of Heritage Foods Limited having its office at401, Diamond House, Adj. to Amrutha Hills, Punjagutta, Hyderabad-500082, have examined the draft Composite Scheme of Arrangement among Heritage Foods Limited ("Transferor Company" or "HFL" or "the Company"), Heritage Foods Retail Limited ("Transferee Company" or "Demerged Company" or "HFRL") and Future Retail Limited ("Resulting Company" or "FRL") and their respective Shareholders and Creditors ("the Scheme") under Sections 391-394 read with Sections 100-103 of the Companies Act, 1956 and Section 52 of the Companies Act, 2013 and other applicable provisions of the Companies Act, 1956 / Companies Act, 2013.

On the basis of information and explanation given to us, we certify the attached undertaking issued by Heritage Foods Limited on the non-applicability, of the conditions prescribed in Para 9 of Annexure 1 of Circular No. CIR/CPD/CMD/16/2015 dated 30th November 2015 issued by Securities Exchange Board of India for the reasons stated below:

- a) This clause is not applicable as additional shares would not be allotted Promoter /Promoter Group, Related Parties of Promoter / Promoter Group, Associates of Promoter / Promoter Group, Subsidiary/(s) of Promoter / Promoter Group of the listed company since the shares are being allotted to the Heritage Foods Limited (Listed Company).
- b) This clause is not applicable as the scheme is envisaged between Company, its wholly owned subsidiary (Heritage Foods Retail Limited) and Future Retail Limited and scheme does not involve any other entity involving Promoter / Promoter Group, Related Parties of Promoter / Promoter Group, Associates of Promoter / Promoter Group, Subsidiary/(s) of Promoter / Promoter Group.

c) This clause is not applicable as the scheme of arrangement does not envisage merger of any subsidiary company with the parent listed company.

New Delhi Branch: A 20/20, DLF City. Phase - I. Gurgaon - 122 002 Ph: 0124-4056132 Mumbal Branch: 511, The Corporate Centre, Nirmal Lifestyle Mail, L.B.S. Road, Muland West, Mumbal - 400 080 Ph: 022-25671155 / 1199 Bangalore Branch: # 202, Akash Avenue. # 34, 1st Main, SBM Colony. Anand Nagar, Bangalore - 580 024 Ph: +91-9241570047

Jalgson Branch: Flat No. 7. Krunal Apertments, Ganapathi Nagar, Jalgson - 475 001 Cell : +91-9966738900 Novi Mumbai Brench: 8-712, Groma House, Near APMC Market. Vashi, Navi Mumbai - 400703 Cell: +91-9967220322

RAJU & PRASAD CHARTERED ACCOUNTANTS

Head Office : 401, "Diamond House", Adj. Amrutha Hills,

Punjagutta, Hyderabad - 500 082 Ph : 040 - 23410404/05/06 Fax : 040 - 23410403

E-mail hyderabad@rajuandprasad.com rajuandprasad@gmail.com

Auditors' Responsibility

We conducted our examination in accordance with the 'Guidance Note on Audit Reports and Certificates for Special Purposes' issued by the Institute of Chartered Accountants of India. Our scope of work did not involve us performing any audit tests in the context of our examination. We have not performed an audit, the objective of which would be the expression of an opinion on the financial statements, specified elements, accounts or items thereof, for the purpose of this certificate. Accordingly, we do not express such opinion. Further our examination did not extend to any aspects of a legal or propriety nature in the aforesaid arrangement /scheme other than the matters referred to in the said certificate.

Restrictions on Use

This Certificate is issued at the request of the Heritage Foods Limited pursuant to the requirements of circulars issued under SEBI (Listing Obligations & Disclosure Requirements) Regulations, 2015 for onward submission to National Stock Exchange and Bombay Stock Exchange. This Certificate should not be used for any other purpose without our prior written consent.

For Raju and Prasad Chartered Accountants FRN: 0034755

UBPA

HYDERALAD

L'Dileep Kumar Partner M No. 223943

Date: 07/11/2016 Place: Hyderabad



Brief particulars of the transferee/resulting and transferor/demerged companies

Transferce & Demerged Company	Transferor Company	Resulting Company
Heritage Foods Retail Limited (HFRL)	Heritage Foods Limited (HFL)	Future Retail Limited (FRL)
01/12/2008	05/06/1992 The name of the Company has changed from Heritage Foods India Limited to Heritage Foods Limited on 16/08/2013	The Company was originally incorporated as Bharti Retail Private Limited on February 7, 2007 at NCT of Delhi, Further, the name of the Company was changed to Bharti Retail Limited vide the necessary resolution dated April 24, 2009 and a fresh certificate of incorporation was obtained on May 21, 2009. Further, name was changed to its present name viz, Future Retail Limited vide special resolution dated November 18, 2015 and fresh certificate of incorporation pursuant to change of name dated May 25, 2016 was obtained from the Registrar of Companies, Maharashtra at Mumbai
# 6-3-541/C, Punjagutta Hyderabad- 500082 Telangana, India	# 6-3-541/C, Punjagutta Hyderabad 500082 Telangana, India	Knowledge House, Shyam Nagar, Off Jogeshwari- Vikhroli Link Road, Jogeshwari (E), Mumbai – 400060
going concern basis as a slu- subsidiary of M/s. Heritage F November 1, 2016 (being the discharge the consideration of issue of 1,40,00,000 (One Co- premium of Rs.86.43/- to HFI The Retail Business of M/s. Heritage F Retail Limited by way of demo 1956 and other applicable pro- be applicable) with effect from Date. In consideration for the demo based on share entitlement Accountants and fairness opi Merchant Banker, FRL will is Seven Thousand Four Hundre to the equity shareholders of the	mp sale to M/s. Heritage Foods Reta Foods Limited (HFL) with effect from the e Appointed Date for slump sale). He f Rs.135 Crores (Rupees One Hundred from Forty Lakhs) equity shares, at formal deritage Foods Retail Limited shall be to erger as per the provisions of Sections ovisions of the Companies Act, 1956 the close of business on March 31, 20 erger of Retail Business of HFRL into report issued by M/s S.R. Batliboi & nion provided by M/s Keynote Corpo usue an aggregate of 1,78,47,420 (One ed and Twenty) equity shares of Rs. 2, HFRL as on the Record Date (as of now	il Limited (HFRL), a wholly owned the commencement of business on critage Foods Retail Limited shall and Thirty Five Crores) by way of face value of Rs.10/- each with a transferred to and vested in Future 391 to 394 of the Companies Act, and Companies Act, 2013 (as may 017 (being the Demerger Appointed FRL in terms of the Scheme and Co. LLP, Independent Chartered rate Services Limited, a Category I of Crore Seventy Eight Lakhs Forty (Rupees Two) each, fully paid up the HFL) on a proportionate basis.
	# 6-3-541/C, Punjagutta Hyderabad- 500082 Telangana, India The Retail, Bakery, Agri and V going concern basis as a slusubsidiary of M/s. Heritage F November 1, 2016 (being the discharge the consideration of issue of 1,40,00,000 (One Copremium of Rs.86.43/- to HFI The Retail Business of M/s. Here Retail Limited by way of demonstrated by the supplicable probability of the supplicable probability of the supplicable probability of the supplicable probability with effect from Date). In consideration for the demonstrated on share entitlement Accountants and fairness oping Merchant Banker, FRL will is Seven Thousand Four Hundreto the equity shareholders of the supplicable of	Heritage Foods Retail Limited (HFL) 01/12/2008 #6-3-541/C, Punjagutta Hyderabad- 500082 Telangana, India The Retail, Bakery, Agri and VetCa verticals as defined in the Schengoing concern basis as a slump sale to M/s. Heritage Foods Retail Subsidiary of M/s. Heritage Foods Limited (HFL) with effect from the Schenge of 1,40,00,000 (One Crore Forty Lakhs) equity shares, at in premium of Rs.86.43/- to HFL. The Retail Business of M/s. Heritage Foods Retail Limited shall be Retail Limited by way of demerger as per the provisions of Sections 1956 and other applicable provisions of the Companies Act, 1956 be applicable) with effect from the close of business on March 31, 20



HERITAGE FOODS LIMITED

(Formerly known as M/s. Heritage Foods (India) Limited)

CIN: L15209TG1992PLC014332 AN ISO: 22000 CERTIFIED COMPANY





Particulars	Transferee & Demerged Company	Transferor Company	Resulting Company
Rationale for the scheme	i. facilitate each business to independently; ii. enhance management focu iii. facilitate investment by str iv. create a platform to enhan v. consolidation of the Retail vi. unlocking of value; and		g growth for each of the verticals
Date of resolution passed by the Board of Director of the company approving the scheme	07/11/2016	07/11/2016	07/11/2016
Date of meeting of the Audit Committee in which the draft scheme has been approved	NA.	07/11/2016	07/11/2016
Appointed Date	Commencement of business on 01/11/2016 (for Part II of the Scheme) & close of business on 31/03/2017 (for Part III of the Scheme)	Commencement of business on 01/11/2016 for Part II of the Scheme	Close of business on 31/03/2017 for Part III of the Scheme
Name of Exchanges where securities of the company are listed	NA	BSE Limited & National Stock Exchange of India	BSE Limited & National Stock Exchange of India
Nature of Business	Retail Trading and Veterinary Care	Manufacturing of Milk & Milk products, Retail trading, Agriculture, Bakery, Veterinary Care and Renewable Energy.	Retail Trading
Capital before the scheme	Authorized Capital: Rs.15,00,00,000 (divided into 1,50,00,000 equity shares of face value Rs.10/each) Paid up Capital: Rs.16,56,000 (divided into 1,65,600 equity shares of face value Rs.10/each) as on 30th September, 2016	Authorized Capital: Rs.50,00,00,000 divided into 4,80,00,000 equity shares of face value Rs.10/- each and 20,00,000 Preference Shares of Rs.10 each. Paid up Capital: Rs.23,19,90,000 (divided into 2,31,99,000 equity shares of face value Rs.10/- each) as on 30th September, 2016	Authorized Capital: Rs.2500,00,00,000 (divided into 1250,00,00,000 equity shares of face value Rs.2/- each) Paid up Capital: 94,26,77,114 (divided into 47,13,38,557 equity shares of face value of Rs 2 each), fully paid up as on 30th September, 2016.



HERITAGE FOODS LIMITED

(Formerly known as M/s. Heritage Foods (India) Limited)

CIN: L15209TG1992PLC014332 AN ISO: 22000 CERTIFIED COMPANY





Particulars	Transferee & Demerged Company	Transferor Company	Resulting Company	
No. of shares to be issued	Rs.135 Crores (Rupees One I- Crore Forty Lakhs) equity shar As a consideration for Demerg Lakhs Forty Seven Thousand	Sale, Heritage Foods Retail Limited shi dundred and Thirty Five Crores) by wees, at face value of Rs.10/- each with a er, FRL will issue an aggregate of 1,78 Four Hundred and Twenty) equity sha shareholders of HFRL as on the Rec	ray of issue of 1,40,00,000 (Once a premium of Rs.86.43/- to HFL 3,47,420 (One Crore Seventy Eight ares of Rs. 2/- (Rupees Two) each,	
Cancellation of shares on account of cross holding, if any		NA		
Capital after the scheme	Paid up Capital: Rs.14,16,56,000 (divided into 1,41,65,600 equity shares of face value Rs.10/- each)	Paid up Capital: Rs.23,19,90,000 (divided into 2,31,99,000 equity shares of face value Rs.10/- each)	Paid up Capital: Rs 97,83,71,954/- (divided into 48,91,85,977 equity shares of face value Rs. 2 each)	
Net Worth Pre Post	as at 30/09/2016 (Rs. in crores) 0.025 6.20	as at 30/09/2016 (Rs. in crores) 278.09 278.60	as at 31/03/2016 (Rs. in crores) 1478.19 1773.19	
Valuation by independent Chartered Accountant - Name of the valuer/valuer firm and Regn no.	Firm No:0034755 and	. Raju and Prasad Chartered Accountant n No:0034755 and S.R. Batliboi & Co. LLP, Chartered Accountants ICAI		
Methods of valuation and value per share arrived under each method with weight given to each method, if any.	For Slump Sale under Part II of the Scheme, Net Assets Value Method has been considered; and For Demerger under Part III of the Scheme, Discounted Cash Flows Method has been considered	For Slump Sale under Part II of the Scheme, Net Assets Value Method has been considered	For Demerger under Part III of the Scheme, Market Price Method (based on average weekly high and low of volume weighted average price during last 26 weeks and 2 weeks) has been considered	
Fair value per shares	Rs.96.43 - for the purpose of Part II of the Scheme (including Retail division and VetCa division of the Company); and Rs.208.25 - for the purpose of Part III of the Scheme [excluding the VetCa division of the company)	NA	Rs. 165.29	





(Formerly known as M/s. Heritage Foods (India) Limited)







Particulars	Com	& Demerged pany	1950/250035050390	Company	Resulting Co	11.0000000
Exchange ratio	As a consideration for Slump Sale, Heritage Foods Retail Limited shall discharge the consideration Rs.135 Crores (Rupees One Hundred and Thirty Five Crores) by way of issue of 1,40,00,00 Crore Forty Lakhs) equity shares, at face value of Rs.10/- each with a premium of Rs.86.43/- FRL will issue an aggregate of 1,78,47,420 (Once Crore Seventy Eight Lakhs Forty Seven T Four Hundred and Twenty) equity shares of Rs. 2/- (Rupees Two) each, fully paid up to the shareholders of HFRL as on the Record Date (HFL).					
Name of Merchant Banker giving fairness opinion	M/s. Keynote	Corporate Service	ces Ltd, Mumbai			
Shareholding	Pre (1	HFRL)	Pre (HFL)	Pre (FR	L)
pattern as on 30/11/2016	No. of Shares	% of holding	No. of Shares	% of holding	No. of Shares	% of holding
Promoter	165600	100.00	9256846	39.90	230115436	48.82
Public	0	0.00	13942154	60.10	241223121	51.18
Custodian	0.	0.00	0	0.00	0	0.00
TOTAL	165600	100.00	23199000	100.00	471338557	100.00
	Post (HFRL)	Post	(HFL)	Post (FR	EL)
	No. of Shares	% of holding	No. of Shares	% of holding	No. of Shares	% of holding
Promoter	14165600*	100.00	9256846	39,90	230115436	47.04
Public	0	0.00	13942154	60.10	259070541	52.96
Custodian	0	0.00	0	0.00	0	0.00
TOTAL	14165600	100.00	23199000	100.00	489185977	100.00
	effective, the capital of HFI reduced upon the paid up However, HFL	heme becoming paid up share RL shall stand cancellation of share capital would continue ire share capital				
No of shareholders	As on 30th Sc	ptember, 2016 7		ptember, 2016 225	As on 30th September, 26 44058 (without considering allotment of shares pursua the Scheme)	
Names of the Promoters	M/s. Heritage	Foods Limited	Promoters along with Persons Acting in Concert Promoters along with Persons Acting in Concert		ni nn Biyani ni Biyani re Biyani te And nited ipital Limited	



HERITAGE FOODS LIMITED*

(Formerly known as M/s. Heritage Foods (India) Limited) CIN: L15209TG1992PLC014332

AN ISO: 22000 CERTIFIED COMPANY







Particulars	Transferee & Demerged Company	Transferor Company	Resulting Company
		13.Mrs. N Brahmani 14.M/s. Megabid Finance & Investment Pvt Ltd	13. M/s. Ryka Commercial Ventures Private Limited 14. M/s. Gargi Business Ventures Private Limited 15. M/s. PfL Industries Limited 16. M/s. Future Corporate Resources Limited
Names of the Board of Directors	1.Mr. D Seetharamaiah 2.Mrs. N Bhuvaneswari 3.Mr. N Lokesh 4.Mrs. N Brahmani 5.Dr. M Sambasiva Rao	1.Mr. D Seetharamaiah 2.Mr. Rajesh Thakur Ahuja 3.Mr. N Sri Vishnu Raju 4.Dr. V Nagaraja Naidu 5.Mr. N Lokesh 6.Mrs. N Bhuvaneswari 7.Mrs. N Brahmani	Mr. Kishore Biyani Mr. Rakesh Biyani Mr. Rajan Bharti Mittal Mr. Rayindra Dhariwal Mr. Shailendra Bhandari Ms. Gagan Singh
Details regarding change in management control if any	No change in control or Mana effective, the Retai	agement of either of the Companies. I business of HFRL shall stand trans	However, upon the Scheme becoming sferred and vested into FRL.





(Formerly known as M/s. Heritage Foods (India) Limited) CIN: L15209TG1992PLC014332



RAJU & PRASAD

CHARTERED ACCOUNTANTS

Head Office :

401, "Diamond House", Adj. Amrutha Hills.

Punjagutta, Hyderabad - 500 082 : 040 - 23410404/05/06

Fax : 040 - 23410403 E-mail: hyderabad@rajuandprasad.com

rajuandprasad@gmail.com

CERTIFICATION OF NETWORTH

Dear Sirs,

- At the request of Heritage Foods Limited ("the Company"), We M/s Raju & Prasad Chartered Accountants (FRN No. 003475S) the statutory auditors of Heritage Foods Limited having its office at 401, Diamond House, Adj. to Amrutha Hills, Punjagutta, Hyderabad-500082, have examined the attached statement of computation of pre and post net worth of the Company as at October 31, 2016 ("the statement"), prepared by the company and initialled by us for identification purposes.
- 2. It has been explained to us that as per the attached copy of the draft Composite Scheme of Arrangement ("the Scheme") between Heritage Foods Limited ("Transferor Company" or "HFL" or "the Company"), Heritage Foods Retail Limited ("Transferee Company" or "Demerged Company" or "HFRL") and Future Retail Limited ("Resulting Company" or "FRL") and their respective Shareholders and Creditors ("the Scheme") under Sections 391-394 read with Sections 100-103 of the Companies Act, 1956 and Section 52 of the Companies Act, 2013 and other applicable provisions of the Companies Act, 1956 / Companies Act, 2013,the proposed appointed date for the purposes of the Part II of the Scheme will be the commencement of business on 1 November, 2016 or such other date as the High Court and/ or the National Company Law Tribunal may direct or approve under the relevant provisions of the Act.
- 3. In the attached statement, the account balances (Equity Share Capital, Reserves and Surplus) used in the computation of pre arrangement net worth of the Company as at October 31, 2016 has been traced from the unaudited books of accounts of the company for the period ended October 31, 2016 made available to us and the post arrangement net worth of the Company has been computed considering the accounting treatment contained in clause16 of the Scheme, which is subject to court approval. As represented to us, the scheme is purposed to be filed with the Hon'ble High Court of judicature at Mumbai and High Court of Judicature at Hyderabad.
- 4. The accompanying statement is responsibility of company's management. Our responsibility is to verify the factual accuracy of the facts stated in the certificate. We conducted our examinations in accordance with the guidance notes on Audit reports and certificates for special purposes issued by the Institute of Chartered Accountants of India. Our Scope of work did not involve us performing any audit tests in the context of our examination. We have not performed an audit, the objective of which would be the expression of an opinion on the financial statements, specified elements, accounts or items thereof, for the purpose of this certificate. Accordingly, we do not express such opinion.
- Based on the procedures mentioned in paragraph 3 above and according to the information and explanation given to us and specific representation received from the apartment, we certify that the pre arrangement net worth and the post arrangement net worth of the Company as at October 31, 2016 computed in accordance with paragraph 3 above is a paragraph.

New Delhi Branch: A 20/20, DLF City, Phase - I, Gurgaon - 122 002 Ph; 0124-4056132

Mumbal Branch: 511, The Corporate Centre. Nirmal Lifestyle Mall. L.B.S. Road, Mulund West. Mumbai - 400 080 Pn: 022-25871155 / 1199

Bangalore Branch: # 202. Akash Ayenue. # 34, 1st Main, S8M Colony. Anand Nagar, Bangalore - 560 024 Ph: +91-9241570047

Jalgaon Branch Flat No. 7, Krunal Apartments. Ganspathi Nagar, Jalgann - 475 001 Cel: +91-9968738900

Nevi Mumbai Branch: 8-712, Groma House Near APMC Market. Vashi. Navi Mumbei - 400703 Cell: +91-9967220322

URL: www.rajuandprasad.com

RAJU & PRASAD

CHARTERED ACCOUNTANTS

Head Office:

401, "Diamond House", Adj. Amrutha Hills,

Punjagutta, Hyderabad - 500 082 Ph : 040 - 23410404/05/06 Fax : 040 - 23410403

E-mail: hyderabad@rajuandprasad.com rajuandprasad@gmail.com

Particulars	Prior to giving effect to the proposed scheme (in Lakhs)	After giving effect to the proposed scheme (in Lakhs)		
Share Capital: Equity issued & Paid up Capital	2319.90	2319.90		
Reserves & Surplus:				
Capital Redemption Reserve	81	81		
Securities Premium Reserve	3,784.14	3,784.14		
Deferred Govt Grants	47.84	47.84		
Warrants money				
Appropriated	318.69	318.69		
General Reserve	7,427,38	7,427.38		
Surplus in statement of P&L	13,830.25	13,830.25		
Capital Reserve on				
composition Scheme	*	51		
Total Reserves & Surplus	25,489.30	25,540.30		
Net Worth	27,809.21	27,860.20		

6. This certificate is intended solely for the use of the management of the Company for the purpose of submission to the Stock Exchange, and is not to be used, referred to or distributed for any other purpose without our prior written consent.

For Raju & Prasad Chartered Accountants FRN No: 0034,255

I Dileep Kumar Partner

M. No: 223943

Date: 07/11/2016 Place: Hyderabad. HYDERABAD

RAJU & PRASAD CHARTERED ACCOUNTANTS

Head Office ; 401, "Diamond House", Adj. Amrutha Hills,

Punjagutta, Hyderabad - 500 082 Ph : 040 - 23410404/05/06 Fax : 040 - 23410403

E-mail: hyderabad@rajuandprasad.com rajuandprasad@gmail.com

CERTIFICATION OF NETWORTH

Dear Sirs.

- At the request of Heritage Foods Retail Limited ("the Company"), we have examined the attached statement of computation of pre and post net worth of the Company as at October 31, 2016 ("the statement"), prepared by the company and initialled by us for identification purposes.
- 2. It has been explained to us that as per the attached copy of the draft Composite Scheme of Arrangement ("the Scheme") between Heritage Foods Limited ("Transferor Company" or "HFL"), Heritage Foods Retail Limited ("Transferee Company" or "Demerged Company" or "HFRL" or "the Company") and Future Retail Limited ("Resulting Company" or "FRL") and their respective Shareholders and Creditors ("the Scheme") under Sections 391-394 read with Sections 100-103 of the Companies Act, 1956 and Section 52 of the Companies Act, 2013 and other applicable provisions of the Companies Act, 1956 / Companies Act, 2013, the proposed appointed date for the purposes of Part II of the Scheme will be the close of business on 31March, 2017 or such other date as the High Court and/ or the National Company Law Tribunal may direct or approve under the relevant provisions of the Act.
- 3. In the attached statement, the account balances (Equity Share Capital, Reserves and Surplus) used in the computation of pre arrangement net worth of the Company as at October 31, 2016 has been traced from the unaudited books of accounts of the Company for the period ended October 31, 2016 made available to us and the post arrangement net worth of the Company has been computed considering the accounting treatment contained in clause 15 and clause 29of the Scheme, which is subject to court approval. As represented to us, the Scheme is purposed to be filed with the Hon'ble High Court of judicature at Mumbai and High Court of Judicature at Hyderabad.
- 4. The accompanying statement is responsibility of company's management. Our responsibility is to verify the factual accuracy of the facts stated in the certificate. We conducted our examinations in accordance with the guidance notes on Audit reports and certificates for special purposes issued by the Institute of Chartered Accountants of India. Our Scope of work did not involve us performing any audit tests in the context of our examination. We have not performed an audit, the objective of which would be the expression of an opinion on the financial statements, specified elements, accounts or items thereof, for the purpose of this certificate. Accordingly, we do not express such opinion.
- 5. Based on the procedures mentioned in paragraph 3 above and according to the information and explanation given to us and specific representation received from the malitagement, we certify that the pre arrangement net worth and the post arrangement over worth of the Company as at October 31, 2016 computed in accordance with paragraph 3 above is as noted below:

New Delhi Branch: A 20/20, DLF City, Phase - I, Gurgaon + 122 002 Ph: 0124-4056132 Mumbal Branch: 511, The Corporate Centre, Nirmal Lifestyle Mail, L.B.S. Road, Mulund West, Mumbai - 400 080 Ph: 022-25671155 / 1198 Bangalore Branch: # 202, Akash Avenue, # 34, 1st Main, SBM Colony, Anand Nagar, Bangalore - 560 024 Ph: +91-9241570047 Jalgaon Branch: Flat No. 7, Krunal Apartments, Ganapathi Nagar, Jalgaon - 475 001 Cell : +91-9966738900 Nevr Mumbai Branch: 8-712, Groma House, Near APMC Market, Vashi, Navi Mumbai - 400703 Cell: +91-9967220322

URL; www.rajuandprasad.com

RAJU & PRASAD

CHARTERED ACCOUNTANTS

Head Office:

401, "Diamond House", Adj. Amrutha Hills,

Punsagutta, Hyderabad - 500 082 Ph 040 - 23410404/05/06 Fax 040 - 23410403

E-mail: hyderabad@rajuandprasad.com rajuandprasad@gmail.com

Particulars		Prior to giving effect to the proposed scheme (in Lakhs)	After giving effect to the proposed scheme (in Lakhs)
Share Capital: Equity Issued & Paid up Capital	(A)	16.56	633.71
Reserves & Surplus: Securities premium Deficit in statement of P&L Total Reserves & Surplus	(B)	(14.09) (14.09)	(14.09) (14.09)
Net Worth		2.47	619.62

6. This certificate is intended solely for the use of the management of the Company for the purpose of submission to the Stock Exchange, and is not to be used, referred to or distributed for any other purpose without our prior written consent.

For Raju & Prasad Chartered Accountants

HYDERABAD

FRN No: 003475S

I. Dileep/Kumar Partner

M. No: 223943

Date: 07/11/2016 Place: Hyderabad.

44, Khatau Building, 1st Floor, Bank Street, Fort, Mumbai - 400 001.

Tel.: 022-22660109 Mob.: +91-92210 39100 gg : rajwan11115@rediffmall.com

Date: November 7, 2016

To,
The Board of Directors,
Future Retail Limited
Knowledge House, Shyam Nagar,
Off Jogeshwari-Vikhroli Link Road,
Jogeshwari (East), Mumbai – 400 060

Dear Sirs.

- At the request of Future Retail Limited ("the Company"), we have examined the attached statement of computation of pre and post net worth of the Company as at March 31, 2016 ("the statement"), prepared by the company and initialed by us for identification purposes.
- 2. It has been explained to us that as per the attached copy of the draft Composite Scheme of Arrangement ("the Scheme") between Heritage Foods Limited ("Transferor Company" or "HFL"), Heritage Foods Retail Limited ("Transferee Company" or "Demerged Company" or "HFRL") and Future Retail Limited ("Resulting Company" or "the Company" or "FRL") and their respective Shareholders and Creditors ("the Scheme") under Sections 391-394 read with Sections 100-103 of the Companies Act, 1956 and Section 52 of the Companies Act, 2013 and other applicable provisions of the Companies Act, 1956 / Companies Act, 2013. The proposed appointed date for the purposes of the Scheme will be the close of business on 31 March, 2017 or such other date as the High Court and/ or the National Company Law Tribunal may direct or approve under the relevant provisions of the Act.
- 3. In the attached statement, the account balances (Equity Share Capital, Reserves and Surplus) used in the computation of pre arrangement net worth of the Company as at March 31, 2016 has been traced from the audited books of accounts of the company for the period ended March 31, 2016 made available to us and the post arrangement net worth of the Company has been computed considering the accounting treatment contained in clause 28 of the Scheme, which is subject to court approval. As represented to us, the scheme is proposed to be filed with the Hon'ble High Court of judicature at Mumbai and High Court of Judicature at Hyderabad.
- 4. The accompanying statement is responsibility of company's management. Our responsibility is to verify the factual accuracy of the facts stated in the certificate. We conducted our examinations in accordance with the guidance notes on Audit reports





44, Khatau Building, 1st Floor, Bank Street, Fort, Mumbel - 400 001. Tel.: 022-22660109

and certificates for special purposes issued by the Institute of Chartered Accountants of India. Our Scope of work did not involve us performing any audit tests in the context of our examination. We have not performed an audit, the objective of which would be the expression of an opinion on the financial statements, specified elements, accounts or items thereof, for the purpose of this certificate. Accordingly, we do not express such opinion.

5. Based on the procedures mentioned in paragraph 3 above and according to the information and explanation given to us and specific representation received from the management, we certify that the pre arrangement net worth and the post arrangement net worth of the Company as at March 31, 2016 computed in accordance with paragraph 3 above is as noted below:

(INR in Crores)

Particulars	Pre-Arrangement	Post- Arrangement
Equity Share Capital	94-27	97.84
Reserves		
Securities Premium		291.43
On Composite Scheme of Arrangement	1874.58	1874.58
Surplus/(Deficit) in P&L Account	(490.66)	(490.66))
Total Net-worth	1478.19	1,773-19

This certificate is intended solely for the use of the management of the Company for the purpose of submission to the Stock Exchange, and is not to be used, referred to or distributed for any other purpose without our prior written consent.

For R S Ajwani & Co.

Chartered Accountants

ICAI Firm Registration No.: 131853W

Ravi Ajwani Proprietor

Membership Number: 035574

Place: Mumbai

Date: 07th November, 2016



Details of Capital evolution of the transferor company (HERITAGE FOODS LIMITED):

Date of Issue	No. of shares issued	Issue Price (Rs.)	Type of Issue (IPO/FPO/ Preferential Issue/ Scheme/ Bonus/ Rights, etc.)	Cumulative capital (No of shares)	Whether listed, if not listed, give reasons thereof
5-05-1992	700	10/	The state of the s	700	These shares
8-09-1993	380000	10/		380700	were not listed
8-09-1993	50000	10/	Pre- IPO allotments	430700	the IPO, all the shares got listed
24-11-1993	375000	10/		805700	
16-11-1994	3494300	10/		4300000	
9-01-1995	6500000	10/	IPO	10800000	
	(810000)	10/	Buy Back (4-1-2002)	9990000	
8-02-2008	1000000	10/	Preferential Issue	10990000	Listed
29-03-2008	539500	10/	Preferential Issue	11529500	Listed
22-01-2013	70000	10/-	Sweat Equity Issue	11599500	
30-07-2013	11599500	10/-	Bonus Issue	23199000	

// Certified True Copy //

FOR HERITAGE FOODS LIMITED

UMAKANTA BARIK Company Secretary M.No: FCS 6317





HERITAGE FOODS LIMITED

(Formerly known as M/s. Heritage Foods (India) Limited)
CIN: L15209TG1992PLC014332
AN ISO: 22000 CERTIFIED COMPANY





Heritage Foods Retail Limited

Details of Capital evolution of the transferee/demerged company (HERITAGE FOODS RETAIL LIMITED):

Date of Issue	No. of shares issued	Issue Price (Rs.)	Type of Issue (IPO/FPO/ Preferential Issue/ Scheme/ Bonus/ Rights, etc.)	Cumulative capital (No of shares)	Whether listed, if not listed, give reasons thereof
01/12/2008	50600	10/-	Initial Paid up Capital	50600	
17/03/2011	65000	10/-	Preferential Issue	115600	Not Listed
12/03/2012	50000	10/-	Preferential Issue	165600	



FUTURE RETAIL

Capital evolution details of the resulting company (Future Retail Limited) as per format enclosed at Annexure VI

Date of Issue	No. of Shares issued	Issue Price (in Rs.)	Type of Issue (IPO/FPO/Preferential Issue/Scheme/Bonus/ Rights, etc.)	Cumulative capital (No of shares)	Whether listed, if not listed, give reasons thereof
February 7, 2007	10,000	10	Initial Paid up Capital	10,000	Not Applicable
March 20, 2008	40,000	10	Conversion of Loan into Equity	50,000	Not Applicable
November 27, 2008	1,99,50,000	10	Right Issue	2,00,00,000	Not Applicable
December, 24, 2008	80,00,000	10	Right Issue	2,80,00,000	Not Applicable
March 20, 2009	38,66,000	10	Right Issue	3,18,66,000	Not Applicable
July 23, 2009	1,69,00,000	10	Right Issue	4,87,66,000	Not Applicable
November 6, 2009	61,80,000	10	Right Issue	5,49,46,000	Not Applicable
January 12, 2010	93,60,000	10	Right Issue	6,43,06,000	Not Applicable
June 1, 2010	1,27,00,000	10	Right Issue	7,70,06,000	Not Applicable
October 18, 2010	9,86,46,000	10	Right Issue	17,56,52,000	Not Applicable
May 23, 2011	33,74,74,000	10	Right issue (Conversion of Loan into Equity)	51,31,26,000	Not Applicable
March 21, 2012	1,96,80,000	10	Right Issue	53,28,06,000	Not Applicable
March 7, 2014	86,58,54,000	10	Right issue (Conversion of Loan into Equity)	1,39,86,60,000	Not Applicable
April 27, 2015	13,30,00,000	10	Right Issue	1,53,16,60,000	Not Applicable
The second secon	18,80,90,000	10	Right Issue	1,71,97,50,000	Not Applicable
and their respective	creditors and si	each and fi Not	between Future Enterpris the Reduction of capital urther Allotment was give Reduction and	and consolidation	Listed with BSE
		200	Reorganization of paid up share capital as provided in the Scheme of Arrangement between FEL and the Company		Limited and National Stock Exchange of India Limited.
May 18, 2016	42,78,60,296	Not Applicable	Pursuant to the Composite Scheme of Arrangement between	47,13,38,557	Listed with BSE Limited and National Stock

Notes:

Our Company was originally incorporated as Bharti Retail Private Limited on February 7, 2007 at NCT of Delta. Further, the name
of our Company was changed to Bharti Retail Limited vide the necessary resolution deted April 24, 2009 and a fresh certificate of
incorporation was obtained on May 21, 2009. Pursuant to the Composite Scheme of Arrangement, the name was further changed
to its present name viz. Future Retail Limited vide special resolution dated November 18, 2015 and fresh certificate of incorporation
pursuant to change of name dated May 25, 2016 was obtained from the Registrar of Companies, Maharashtra et Mumbal

FEL and the Company.

Exchange india Limited.

 As on date 47, 13,35,557 shares of Rs. 24 each of the Company are Listed on BSE Limited and The National Stock Exchange of India Limited effective 29th August, 2016





The General Manager, Department of Corporate Services, BSE Limited, P.J. Towers, Dalal Street. Mumbai - 400 001.

Dear Sir.

Application under Regulation 37 of the SEBI (Listing Obligations and Disclosure Requirements), Sub: Regulations, 2015 for the proposed Composite Scheme of Arrangement Among Heritage Foods Limited ("Transferor Company" or "HFL" or "the Company") and Heritage Foods Retail Limited ("Transferee Company" or "Demerged Company" or "HFRL") and Future Retail Limited ("Resulting Company" or "FRL") and their respective Shareholders and Creditors ("the Scheme") under Sections 391-394 read with Sections 100-103 of the Companies Act, 1956 and Section 52 of the Companies Act, 2013 and other applicable provisions of the Companies Act, 1956 / Companies Act, 2013

In connection with the above application, we hereby confirm that:

- The proposed scheme of arrangement to be presented to any Hon'ble High Court of Hyderabad / High Court of Mumbai does not in any way violate or override or circumscribe the provisions of the SEBI Act, 1992, the Securities Contracts (Regulation) Act, 1956, the Depositories Act, 1996, the Companies Act, 1956 / Companies Act, 2013, the rules, regulations and guidelines made under these Acts, and the provisions as explained in Regulation 11 of the SEBI (Listing obligations and Disclosure Requirements) Regulations, 2015 or the requirements of BSE Limited.
- b) In the explanatory statement to be forwarded by the company to the shareholders u/s 393 or accompanying a proposed resolution to be passed u/s 100 of the Companies Act, it shall disclose:
 - the pre and post-arrangement (expected) capital structure and shareholding pattern;
 - ii) the "fairness opinion" obtained from an Independent merchant banker on valuation of assets / shares done by the valuer for the company and unlisted company;
 - iii) The Complaint report; and
 - iv) The observation letter issued by the stock exchange
- c) The draft scheme of arrangement together with all documents mentioned in Para 1(A)(7)(a) of SEBI Circular no. CIR/CFD/CMD/16/2015 dated November 30, 2015, has been disseminated on company's website as per Website link given hereunder:

http://heritagefoods.in/draft.html

- d) The company shall disclose the observation letter of the stock exchange on its website within 24 hours of receiving the same.
- The Company is not required to obtain shareholders' approval by way of special resolution passed through postal ballot/ e-voting. The requirement of the Company to proceed with the draft Composite Scheme only if the vote cast by the public shareholders in favor of the proposal is more than the number of votes cast by public shareholders against it, is not applicable.
- The documents filed by the Company with the Exchange are same/ similar/ identical in all respect, which have been filled by the Company with Registrar of Companies/SEBI/Reserve Bank of India, wherever applicable.
- The paid up capital of HFRL may undergo a change from the one given in the Scheme on account of further infusion for business purposes. Further, the Scheme also provides for reduction of paid up capital of HFRL. However, the number of shares of shares to be issued by FRL pursuant to the Composite Scheme shall remain unchanged (i.e. 1,78,47,420 equity shares of Rs. 2/- each fully paid) irrespective of the change in the share capital of HFRL

Date: 14/11/2016

HERITAGE FOODS LIMITED

(Formerly known as M/s. Heritage Foods (India) Limited)

CIN: L15209TG1992PLC014332



Umakanta Barik

Company Secretary

RAJU & PRASAD

CHARTERED ACCOUNTANTS

Head Office:

401, "Diamond House", Adj. Amrutha Hills,

Punjagutta, Hyderabad - 500 082 Ph = 040 - 23410404/05/06

Fax : 040 - 23410403

E-mail: hyderabad@rajuandprasad.com rajuandprasad@gmail.com

Auditor's Certificate

To, The Board of Directors, Heritage Foods Limited, Hyderabad.

Certificate on accounting treatment in the Draft Scheme of Arrangement in accordance with Para I (A) 5 of Annexure 1 of SEBI Circular No. CIR/CFD/CMD/16/2015 dated November 30, 2015.

We, M/s Raju & Prasad Chartered Accountants (FRN No. 003475S), the statutory auditors of Heritage Foods Limited having its office at 401, Diamond House, Adj. to Amrutha Hills, Punjagutta, Hyderabad-500082, have examined the proposed accounting treatment specified in clause 15 of the draft Composite Scheme of Arrangement between Heritage Foods Limited ("Transferor Company" or "HFL" or "the Company"), Heritage Foods Retail Limited ("Transferee Company" or "Demerged Company" or "HFRL") and Future Retail Limited ("Resulting Company" or "FRL") and their respective Shareholders and Creditors ("the Scheme") under Sections 391-394 read with Sections 100-103 of the Companies Act, 1956 and Section 52 of the Companies Act, 2013 and other applicable provisions of the Companies Act, 1956 / Companies Act, 2013and Other Generally Accepted Accounting Principles.

The responsibility for the preparation of the Draft Composite Scheme and its compliance with the relevant laws and regulations, including the applicable Accounting Standards as aforesaid, is that of the Board of Directors of the Companies involved. Our responsibility is only to examine and report whether the Draft Scheme complies with the applicable Accounting Standards and Other Generally Accepted Accounting Principles. Nothing contained in this Certificate, nor anything said or done in the course of, or in connection with the services that are subject to this Certificate, will extend any duty of care that we may have in our capacity of the statutory auditors of any financial statements of the Company. We carried out our examination in accordance with the Guidance Note on Audit Reports and Certificates for Special Purposes, issued by the Institute of Chartered Accountants of India.

Based on our examination and according to the information and explanations given to us, we confirm that the accounting treatment contained in the aforesaid Draft Scheme is in compliance with Securities and Exchange Board of India (Listing Obligations and Disclosure Requirements) Regulations, 2015 and circulars issued there under and all the applicable Accounting Standards notified by the Central Government read with relevant rules issued thereunder and Government all the Ministry of Corporate Argos.

New Delhi Branch: A 20/20, DLF City, Phase - I. Gurgaon - 122 002 Ph: 0124-4056132 Mumbai Branch: 511, The Corporate Centre, Nimus Lifestyle Mat, L.B.S. Road, Mulund West, Mumbai - 400 080 Ph; 022-25671155 / 1199

HYDERABAD

Bangalore Branch: # 202, Akash Avenue, # 34, 1st Main, SBM Colony, Anand Nagar, Bangalore - 560 024 Ph : +91-9241570047

Jalgaon Branch: Flat No. 7, Krunal Apartments, Ganapathi Negar, Jalgaon - 475 001 Cell : +91-9866738900 Navi Mumbai Branch: B-712, Grome House, Near APMC Market, Voshi, Navi Mumbai - 400703 Cell: +91-9967220322

RAJU & PRASAD CHARTERED ACCOUNTANTS

Head Office :

401, "Diamond House", Adj. Amrutha Hills,

Punjagutta, Hyderabad - 500 082 Ph : 040 - 23410404/05/06 Fax : 040 - 23410403

E-mail: hyderabad@rajuandprasad.com rajuandprasad@gmail.com

This Certificate is issued at the request of the Heritage Foods Limited pursuant to the requirements of circulars issued under SEBI (Listing Obligations and Disclosure Requirements) Regulations, 2015 for onward submission to the BSE Limited and National Stock Exchanges of India Limited. This Certificate should not be used for any other purpose without our prior written consent.

For Raju & Prasad Chartered Accountants

MYDERABAD

FRN: 003475S

I. Dileep Kumar M No: 223943

Date: 07/11/2016 Place: Hyderabad



Heritage Foods Limited

List of Directors

SI. No	Name	Designation	Residence Address	PAN Number/ DIN No	Date of Birth
1	Sri Devineni Seetharamaiah S/o D Venkatramayya	Chairman	Plot No.1303A, Road No.65, Jubilee Hills, Hyderabad – 500 033 Phone: 040-23548424	ACYPD0157H/ 00005016	22-04-1926
2	Smt N Bhuvaneswari W/o N Chandra Babu Naidu	Vice Chairperson & Managing Director	Plot No.1310, Road No.65, Jubilee Hills, Hyderabad – 500 033 Phone: 040-23608121	ABEPN6265D/ 00003741	20-06-1962
3	Dr Vadlamudi Nagaraja Naidu S/o V Rama Naidu	Director	Plot No. 238 Road No. 18, Jubilee Hills, Hyderabad – 500033 Phone: 040-23548345	AAZPV1250G/ 00003730	01-07-1947
4	Srivishnu Raju Nandyala S/o, N.P.K Raju	Director	Plot No.616, Road Noo.33 Jubilee Hills, Hyderabad-500033 phone:040-23555445	AAUPN8358A/ 00025063	28-12-1973
5	Sri N Lokesh S/o N Chandra Babu Naidu	Director	Plot No.1310, Road No.65, Jubilee Hills, Hyderabad – 500 033 Phone: 040-23608121	ACBPN2492G/ 02230945	23-01-1983
6	Smt. N Brahmani W/o N Lokesh	Executive Director	Plot No. 1310, Road No.65, Jubilee Hills, Hyderabad – 500 033 Phone: 040-23608121	AERPN8112P/ 02338940	21-12-1987
7	Sri. Rajesh Thakur Ahuja S/o. Thakur T Ahuja	Director	Mehrise Flat no.501, Gujar lane, Near Poddar school, Santacruz (W) Mumbai-400054	AAAPA9558P/ 00371406	02-05-1971

//CERTIFIED TRUE COPY//

UMAKANTA BARIK

Company Secretary M No: FCS-6317







(Formerly known as M/s. Heritage Foods (India) Limited) CIN: L15209TG1992PLC014332







Heritage Foods Retail Limited

Heritage Foods Retail Limited

List of Directors

Name	Designation	Residence Address	PAN No.
Sri D Seetharamaiah	Director	Plot No.1303A, Road No.65, Jubilee Hills, Hyderabad – 500 033	ACYPD0157H
Smt N Bhuvaneswari	Director	Plot No.1310, Road No.65, Jubilee Hills, Hyderabad – 500 033	ABEPN6265D
Sri Nara Lokesh	Director	Plot No.1310, Road No.65, Jubilee Hills, Hyderabad – 500 033	ACBPN2492G
Dr. M Sambasiya Rao	Director	Flat No. 111,Teja Block, My Home, Navdweep Apartments, Madhapur, Hyderabad-500081	AEDPM0491G
Smt N Brahmani	Managing Director	Plot No.1310, Road No.65, Jubilee Hills, Hyderabad – 500 033	SRTPN8112P

For HERITAGE FOODS RETAIL LIMITED

Authorized Signatory



Regd. Office: #6-3-541/C, Panjagutta, Hyderabad - 500082



List of Directors of Future Retail Limited

Sr. No.	Name	Designation	DIN	PAN No.
1.	Mr. Kishore Biyani 406, Jeevan Vihar, Manav Mandir Road, Mumbai-400 006	Chairman and Managing Director	00005740	AACPB0199B
2.	Mr. Rakesh Biyani Flat No. 1903,19th Floor, B Wing, Vivarea Building, Sane Guruji Marg, Jacob Circle, Mumbal 400 011	Joint Managing Director	00005806	AAEPB3651L
3.	Mr. Rajan Sharti Mittel E-9/ 17, Vasant Vihar, Vasant Marg, New Delhi-110057	Non-executive Director	00028016	ABBPM8873D
4.	Mr. Shallendra Bhandari E 27 Dhanraj Mahal, Chhatrapati Shivaji Maharaj Marg, Apollo Bundar, Muntbai 400001	Independent Director	00317334	AADPB2390K
5.	Mr. Ravindra Dhariwal Asshray Farm, Opp. N.V. Farm, Sub P.O., S P School, Bhatti Mines, Asola Village, New Delhi, Delhi - 110030	Independent Director	00003922	ADPPD1049Q
6.	Ms. Gagan Singh L = 4/4 Outab Enclave, DLF Phase = II, Gurgaon = 122 002, Haryana	Independent Director	01097014	AATPS7285C

For Future Retail Limited

Virendra Samani Dy. Company Secretary ACS # 12971